As filed with the Securities and Exchange Commission on March 18, 2011 Reg

Registration No. 333-

UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, DC 20549

FORM S-8 REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

INTEL CORPORATION

(Exact Name of Registrant as Specified in its Charter)

Delaware (State or Other Jurisdiction of Incorporation or Organization)

2200 Mission College Blvd. Santa Clara, CA (Address of Principal Executive Offices) 94-1672743 (I.R.S. Employer Identification Number)

> 95054-1549 (Zip Code)

McAfee, Inc. 1997 Stock Incentive Plan McAfee, Inc. 2010 Equity Incentive Plan Foundstone, Inc. 2000 Stock Plan MX Logic, Inc. 2002 Equity Incentive Plan Secure Computing Corporation (formerly CipherTrust, Inc.) 2000 Stock Option Plan Secure Computing Corporation 2002 Stock Incentive Plan Cyberguard Corporation Employee Stock Option Plan SafeBoot Option Plan 2006 Network Associates, Inc. 2000 Nonstatutory Stock Option Plan (Full Title of the Plans)

> CARY I. KLAFTER, ESQ. Vice President and Secretary Intel Corporation 2200 Mission College Blvd. Santa Clara, CA 95054-1549 (Name and Address of Agent for Service)

(408) 765-8080 (Telephone Number, Including Area Code, of Agent for Service)

> Copies to: Robert Townsend, Esq. Jackie Liu, Esq. Morrison & Foerster LLP 425 Market Street San Francisco, California 94105 (415) 268-7000

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer," "accelerated filer," and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer \square Non-accelerated filer \square (Do not check if a smaller reporting company) Accelerated filer \square Smaller reporting company \square

	CALCULATION OF I	REGISTRATION FEE		
Title of securities to be registered	Amount to be registered (1) Pro		Proposed maximum aggregate offering price (2)	Amount of registration fee (3) (4)
Common Stock, par value \$0.01 per share, issuable under the McAfee, Inc. 1997 Stock Incentive Plan	15,352,437	\$17.65	\$271,035,574	\$31,467.23
Common Stock, par value \$0.01 per share, issuable under the McAfee, Inc. 2010 Equity Incentive Plan	1,741,397	\$19.40	\$33,784,583	\$3,922.39
Common Stock, par value \$0.01 per share, issuable under the Foundstone, Inc. 2000 Stock Plan	45,970	\$1.33	\$61,140	\$7.10
Common Stock, par value \$0.01 per share, issuable under the MX Logic, Inc. 2002 Equity Incentive Plan	56,408	\$6.52	\$367,780	\$42.70

Common Stock, par value \$0.01 per share, issuable under the Secure Computing Corporation (formerly CipherTrust, Inc.) 2000 Stock Option Plan	3,947	\$20.75	\$81,900	\$9.51
Common Stock, par value \$0.01 per share, issuable under the Secure Computing Corporation 2002 Stock Incentive Plan	84,830	\$20.75	\$1,760,223	\$204.36
Common Stock, par value \$0.01 per share, issuable under the Cyberguard Corporation Employee Stock Option Plan	8,370	\$20.75	\$173,678	\$20.16
Common Stock, par value \$0.01 per share, issuable under the SafeBoot Option Plan 2006	119,560	\$7.09	\$847,680	\$98.42
Common Stock, par value \$0.01 per share, issuable under the Network Associates, Inc. 2000 Nonstatutory Stock Option Plan	86,903	\$9.32	\$809,936	\$94.03

 Pursuant to Rule 416(a) under the Securities Act of 1933, as amended (the "Securities Act"), this Registration Statement shall also cover any additional shares of the Registrant's common stock in respect of the securities identified in the above table as a result of any stock dividend, stock split, recapitalization or other similar transaction.

(2) Estimated solely for the purpose of calculating the registration fee, based on the average of the high and low prices of the Registrant's common stock on March 14, 2011, as quoted on the Nasdaq Global Select Market as well as the weighted average exercise prices of the options outstanding under the applicable equity plan.

(3) Calculated solely for purposes of this offering pursuant to Rule 457(h) of the Securities Act on the basis of the maximum offering price per share underlying the options issued pursuant to the Plans.

(4) Pursuant to Rule 457(p), Intel Corporation, the holder of all of the outstanding voting securities of McAfee, Inc., offsets the \$35,865.90 registration fee by fees totaling \$72,129.67 paid by McAfee, Inc. with respect to shares of McAfee, Inc. common stock remaining unsold after the termination of McAfee, Inc.'s Registration Statement on Form S-8, Registration Number 333-168619, filed on August 6, 2010. Accordingly, a filing fee of \$0 is being paid herewith.

EXPLANATORY NOTE

On August 18, 2010, Intel Corporation, a Delaware corporation (the "Corporation" or the "Registrant") and its wholly-owned subsidiary, Jefferson Acquisition Corporation, entered into an Agreement and Plan of Merger (the "Merger Agreement") with McAfee, Inc. ("McAfee") pursuant to which McAfee became a wholly-owned subsidiary of the Corporation (the "Merger"). The Merger was completed on February 28, 2011. This Registration Statement on Form S-8 is filed by the Corporation, relating to 17,499,822 shares of its common stock, par value \$0.001 per share (the "Common Stock"), issuable to eligible employees of the Corporation pursuant to awards assumed by the Corporation granted under the McAfee, Inc. 1997 Stock Incentive Plan and the McAfee, Inc. 2010 Equity Incentive Plan, Foundstone, Inc. 2000 Stock Plan, MX Logic, Inc. 2002 Equity Incentive Plan, Secure Computing Corporation (formerly CipherTrust, Inc.) 2000 Stock Option Plan, Secure Computing Corporation Plan, Cyberguard Corporation Employee Stock Option Plan, SafeBoot Option Plan 2006 and Network Associates, Inc. 2000 Nonstatutory Stock Option Plan (the "Plans").

PART I

INFORMATION REQUIRED IN THE SECTION 10(a) PROSPECTUS

Item 1. Plan Information.

Not filed as part of this Registration Statement pursuant to Note to Part 1 of Form S-8.

Item 2. Registrant Information and Employee Plan Annual Information.

Not filed as part of this Registration Statement pursuant to Note to Part 1 of Form S-8.

PART II

INFORMATION REQUIRED IN THE REGISTRATION STATEMENT

Item 3. Incorporation of Documents by Reference.

The following documents, which have previously been filed by the Corporation with the Securities and Exchange Commission (the "Commission"), are incorporated by reference herein and shall be deemed to be a part hereof:

(1) The Corporation's latest Annual Report pursuant to Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or latest prospectus filed pursuant to Rule 424(b) under the Securities Act of 1933, as amended, that contains audited financial statements for the Registrant's latest fiscal year for which such statements have been filed;

(2) All other reports filed pursuant to Section 13(a) or 15(d) of the Exchange Act since the end of the fiscal year covered by the Registrant's latest annual report or prospectus referred to in (1) above; and

(3) The description of the Common Stock set forth under the caption "Description of Capital Stock" in the Registrant's registration statement on Form S-4, filed with the Commission on March 26, 2009, File No. 333-158222, together with any amendment or report filed with the Commission for the purpose of updating such description.

In addition, all documents filed by the Corporation pursuant to Sections 13(a), 13(c), 14 or 15(d) of the Exchange Act prior to the filing of a post-effective amendment to this Registration Statement which indicates that all securities offered hereby have been sold or which deregisters all such securities then remaining unsold, shall be deemed to be incorporated by reference in this Registration Statement and made part hereof from their respective dates of filing (such documents, and the documents listed above, being hereinafter referred to as "Incorporated Documents"); provided, however, that the documents listed above or subsequently filed by the Corporation pursuant to Sections 13(a), 13(c), 14 and 15(d) of the Exchange Act in each year during which the offering made by this Registration Statement is in effect prior to the filing with the Commission of the Corporation's Annual Report on Form 10-K covering such year shall cease to be Incorporated Documents or be incorporated by reference in this Registration Statement from and after the filing of such Annual Report. The Corporation's Exchange Act file number with the Commission is 000-06217.

Any statement contained in an Incorporated Document shall be deemed to be modified or superseded for purposes of this Registration Statement to the extent that a statement contained herein or in any other subsequently filed Incorporated Document modifies or supersedes such statement. Any statement contained herein shall be deemed to be modified or superseded for purposes of this Registration Statement to the extent that a statement contained in any subsequently filed Incorporated Document modifies or supersedes such statement. Any such statement so modified or superseded shall not be deemed, except as so modified or superseded, to constitute a part of this Registration Statement.

Item 4.	Description	of Securities.
Item 4.	Description	of securities.

Not applicable.

Item 5. Interests of Named Experts and Counsel.

Not applicable.

Item 6. Indemnification of Directors and Officers.

Section 145 of the Delaware General Corporation Law (the "DGCL") makes provision for the indemnification of officers and directors of corporations in terms sufficiently broad to indemnify the officers and directors of the Corporation under certain circumstances from liabilities (including reimbursement of expenses incurred) arising under the Securities Act. Section 102(b)(7) of the DGCL permits a corporation to provide in its certificate of incorporation that a director of the corporation shall not be personally liable to the corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except for liability (i) for any breach of the director's duty of loyalty to the corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) in respect of certain unlawful dividend payments or stock redemptions or repurchases, or (iv) for any transaction from which the director derived an improper personal benefit.

As permitted by the DGCL, the Corporation's Certificate of Incorporation (the "Charter") provides that, to the fullest extent permitted by the DGCL or decisional law, no director shall be personally liable to the Corporation or to its stockholders for monetary damages for breach of his fiduciary duty as a director. The effect of this provision in the Charter is to eliminate the rights of the Corporation and its stockholders (through stockholders' derivative suits on behalf of the Corporation) to recover monetary damages against a director for breach of fiduciary duty as a director thereof (including breaches resulting from negligent or grossly negligent behavior) except in the situations described in clauses (i)-(iv), inclusive, above. These provisions will not alter the liability of directors under federal securities laws.

The Corporation's Bylaws (the "Bylaws") provide that the Corporation shall indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative by reason of the fact that he is or was a director, officer, employee or agent of the Corporation or is or was serving at the request of the Corporation as a director, officer, employee or agent of any other corporation or enterprise (including an employee benefit plan), against all expenses, liability and loss (including attorneys' fees, judgments, fines, ERISA excise taxes and penalties, and amounts paid or to be paid in settlement, and any interest, assessments, or other charges imposed thereof, and any taxes imposed on such person as a result of such payments) reasonably incurred or suffered by such person in connection with investigating, defending, being a witness in, or participating in (including on appeal), or preparing for any of the foregoing in such action, suit or proceeding, to the fullest extent authorized by the DGCL, provided that the Corporation or brought to enforce certain indemnification rights.

The Bylaws also provide that expenses incurred by an officer or director of the Corporation (acting in his capacity as such) in defending any such action, suit or proceeding shall be paid by the Corporation, provided that if required by the DGCL such expenses shall be advanced only upon delivery to the Corporation of an undertaking by or on behalf of such director or officer to repay such amount if it shall ultimately be determined that he is not entitled to be indemnified by the Corporation. Expenses incurred by other agents of the Corporation may be advanced upon such terms and conditions as the Board of Directors of the Corporation deems appropriate. Any obligation to reimburse the Corporation for expenses advanced under such provisions shall be unsecured and no interest shall be charged thereon.

The Bylaws also provide that indemnification provided for in the Bylaws shall not be deemed exclusive of any other rights to which the indemnified party may be entitled; that any right of indemnification or protection provided under the Bylaws shall not be adversely affected by any amendment, repeal, or modification of the Bylaws; and that the Corporation may purchase and maintain insurance to protect itself and any such person against any such expenses, liability and loss, whether or not the Corporation would have the power to indemnify such person against such expenses, liability or loss under the DGCL or the Bylaws.

In addition to the above, the Corporation has entered into indemnification agreements with each of its directors and certain of its officers. The indemnification agreements provide directors and officers with the same indemnification by the Corporation as described above and assure directors and officers that indemnification will continue to be provided despite future changes in the Bylaws of the Corporation. The Corporation also provides indemnity insurance pursuant to which officers and directors are indemnified or insured against liability or loss under certain circumstances, which may include liability or related loss under the Securities Act and the Exchange Act.

Item /.	Exemption from Registration Claimed.	
	Not applicable.	
Item 8.	Exhibits.	
filed herewith:	Unless otherwise indicated below as being incorporated by reference to another filing of the Corporation with the Commission, each of the following exhibits i	
Exhibit No.	Exhibit Description	
4.1*	Intel Corporation Third Restated Certificate of Incorporation dated May 17, 2006 (incorporated by reference to Exhibit 3.1 of the Registrant's Current Report on Form 8-K as filed on May 22, 2006, File No. 000-06217).	
4.2*	Intel Corporation Bylaws as amended on May 19, 2009 (incorporated by reference to Exhibit 3.1 of the Registrant's Current Report on Form 8-K as filed on May 22, 2009, File No. 000-06217).	
5.1	Opinion of Morrison & Foerster LLP.	
23.1	Consent of Morrison & Foerster LLP (included in Exhibit 5.1).	
23.2	Consent of Ernst & Young LLP, Independent Registered Public Accounting Firm.	
24.1	Power of Attorney (contained on signature page hereto).	
99.1	McAfee, Inc. 1997 Stock Incentive Plan.	
99.2	McAfee, Inc. 2010 Equity Incentive Plan.	
99.3	Foundstone, Inc. 2000 Stock Plan.	
99.4	MX Logic, Inc. 2002 Equity Incentive Plan.	
99.5	Secure Computing Corporation (formerly CipherTrust, Inc.) 2000 Stock Option Plan.	
99.6	Secure Computing Corporation 2002 Stock Incentive Plan.	
99.7	Cyberguard Corporation Employee Stock Option Plan.	
99.8	SafeBoot Option Plan 2006.	
99.9	Network Associates, Inc. 2000 Nonstatutory Stock Option Plan.	
*Incorporated b	v reference	

*Incorporated by reference

Item 7.

Exemption from Registration Claimed.

Item 9. Undertakings.

- (1) The undersigned Registrant hereby undertakes:
 - (a) To file, during any period in which offers or sales are being made, a post-effective amendment to this registration statement:
 - (i) To include any prospectus required by Section 10(a)(3) of the Securities Act;

(ii) To reflect in the prospectus any facts or events arising after the effective date of the registration statement (or the most recent posteffective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the registration statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high and of the estimated maximum offering range may be reflected in the form of prospectus filed with the Commission pursuant to Rule 424(b) if, in the aggregate, the changes in volume and price represent no more than a 20 percent change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in the effective registration statement;

(iii) To include any material information with respect to the plan of distribution not previously disclosed in the registration statement or any material change to such information in the registration statement;

provided, however; that paragraphs (1)(a)(i) and (1)(a)(i) do not apply if the information required to be included in a post-effective amendment by those paragraphs is contained in periodic reports filed with or furnished to the Commission by the Registrant pursuant to Section 13 or 15(d) of the Exchange Act that are incorporated by reference in the registration statement.

- (b) That, for the purpose of determining any liability under the Securities Act, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.
- (c) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.

(2) The undersigned Registrant hereby undertakes that, for purposes of determining any liability under the Securities Act, each filing of the Registrant's annual report pursuant to Section 13(a) or 15(d) of the Exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the exchange Act (and, where applicable, each filing of an employee benefit

(3) Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of the Registrant pursuant to the foregoing provisions, or otherwise, the Registrant has been advised that in the opinion of the Commission such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the Registrant of expenses incurred or paid by a director, officer or controlling person of the Registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the Registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

[SIGNATURES ON THE NEXT PAGE]

SIGNATURES

Pursuant to the requirements of the Securities Act, Registrant certifies that it has reasonable grounds to believe that it meets all the requirements for filing on Form S-8 and has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Santa Clara, State of California, on this 18th day of March, 2011.

INTEL CORPORATION

By: /s/ Stacy J. Smith

Stacy J. Smith Senior Vice President, Chief Financial Officer, and Principal Accounting Officer

Each person whose signature appears below constitutes and appoints A. Douglas Melamed, Stacy J. Smith, and Cary I. Klafter and each of them, his true and lawful attorneys-in-fact and agents, each with full power of substitution and resubstitution, severally, for him and in his name, place and stead, in any and all capacities, to sign any and all amendments (including post-effective amendments) to this Registration Statement, and to file the same, with all exhibits thereto and other documents in connection therewith, with the Securities and Exchange Commission, granting unto said attorneys-in-fact and agents, and each of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes as he might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents, or any of them or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

[REMAINDER OF THIS PAGE LEFT INTENTIONALLY BLANK]

Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following persons in the capacities and on the date indicated.

/s/ PAUL S. OTELLINI Paul S. Otellini

President, Chief Executive Officer, Director and Principal Executive Officer March 18, 2011

/s/ CHARLENE BARSHEFSKY Charlene Barshefsky Director March 18, 2011

/s/ Susan L. Decker

Susan L. Decker Director March 18, 2011

/s/ JOHN J. DONAHOE John J. Donahoe Director March 18, 2011

/s/ REED E. HUNDT Reed E. Hundt Director March 18, 2011

/s/ James D. Plummer

James D. Plummer Director March 18, 2011 /s/ STACY J. SMITH Stacy J. Smith

Senior Vice President, Chief Financial Officer, and Principal Accounting Officer March 18, 2011

/s/ DAVID S. POTTRUCK

David S. Pottruck Director March 18, 2011

/s/ Jane E. Shaw

Jane E. Shaw Chairman of the Board of Directors March 18, 2011

/s/ Frank D. Yeary

Frank D. Yeary Director

March 18, 2011

/s/ DAVID B. YOFFIE David B. Yoffie Director March 18, 2011

Exhibit No.	Exhibit Description
4.1*	Intel Corporation Third Restated Certificate of Incorporation dated May 17, 2006 (incorporated by reference to Exhibit 3.1 of the Registrant's Current Report on Form 8-K as filed on May 22, 2006, File No. 000-06217).
4.2*	Intel Corporation Bylaws as amended on May 19, 2009 (incorporated by reference to Exhibit 3.1 of the Registrant's Current Report on Form 8-K as filed on May 22, 2009, File No. 000-06217).
5.1	Opinion of Morrison & Foerster LLP.
23.1	Consent of Morrison & Foerster LLP (included in Exhibit 5.1).
23.2	Consent of Ernst & Young LLP, Independent Registered Public Accounting Firm.
24.1	Power of Attorney (contained on signature page hereto).
99.1	McAfee, Inc. 1997 Stock Incentive Plan.
99.2	McAfee, Inc. 2010 Equity Incentive Plan.
99.3	Foundstone, Inc. 2000 Stock Plan.
99.4	MX Logic, Inc. 2002 Equity Incentive Plan.
99.5	Secure Computing Corporation (formerly CipherTrust, Inc.) 2000 Stock Option Plan.
99.6	Secure Computing Corporation 2002 Stock Incentive Plan.
99.7	Cyberguard Corporation Employee Stock Option Plan.
99.8	SafeBoot Option Plan 2006.
99.9	Network Associates, Inc. 2000 Nonstatutory Stock Option Plan.

*Incorporated by reference

MORRISON FOE

FOERSTER

425 MARKET STREET SAN FRANCISCO CALIFORNIA 94105-2482 TELEPHONE: 415.268.7000 FACSIMILE: 415.268.7522 WWW.MOFO.COM MORRISON & FOERSTER LLP NEW YORK, SAN FRANCISCO, LOS ANGELES, PALO ALTO, SAN DIEGO, WASHINGTON, D.C. NORTHERN VIRGINIA, DENVER, SACRAMENTO, WALNUT CREEK TOKYO, LONDON, BEIJING, SHANGHAI, HONG KONG, SINGAPORE, BRUSSELS

EXHIBIT 5.1

OPINION OF MORRISON & FOERSTER LLP

March 18, 2011

Intel Corporation 2200 Mission College Blvd. Santa Clara, CA 95054-1549

Re: Registration of Securities of Intel Corporation

Ladies and Gentlemen:

At your request, we have examined the Registration Statement on Form S-8 to be filed with the Securities and Exchange Commission (the "SEC") in connection with the registration under the Securities Act of 1933, as amended, of an aggregate of Seventeen Million Four Hundred Ninety Nine Thousand Eight Hundred Twenty Two (17,499,822) shares (the "Plan Shares") of common stock, par value \$0.001 ("Common Shares"), of Intel Corporation (the "Company"), issuable pursuant to stock options and restricted stock units (the "Equity Awards") granted under the McAfee, Inc. 1997 Stock Incentive Plan and the McAfee, Inc. 2010 Equity Incentive Plan, Foundstone, Inc. 2000 Stock Plan, MX Logic, Inc. 2002 Equity Incentive Plan, Secure Computing Corporation (formerly CipherTrust, Inc.) 2000 Stock Option Plan, Secure Computing Corporation 2002 Stock Incentive Plan, Cyberguard Corporation Employee Stock Option Plan, SafeBoot Option Plan 2006 and Network Associates, Inc. 2000 Nonstatutory Stock Option Plan (collectively, the "Plans"), as assumed in connection with the Agreement and Plan of Merger dated August 18, 2010 (the "Merger Agreement") by and among the Company, Jefferson Acquisition Corporation, and McAfee, Inc.

This opinion is being furnished in accordance with the requirements of Item 601(b)(5) of Regulation S-K under the Securities Act of 1933, as amended.

As your counsel in connection with the Registration Statement, we have examined the proceedings taken by the Company in connection with the assumption of the Equity Awards and the authorization of the issuance of the Plan Shares, and such documents as we have deemed necessary to render this opinion. For the purpose of the opinion rendered below, we have assumed that in connection with the issuance of the Plan Shares, the Company will receive consideration in an amount not less than the aggregate par value of the Plan Shares covered by each such issuance. We have examined the originals, or photostatic or certified copies, of such records of the Company and certificates of officers of the Company and such other documents as we have deemed relevant and necessary as the basis for the opinions set forth below. In our examination, we have assumed the genuineness of all signatures, the legal capacity and competency of all natural persons, the authenticity of all documents submitted to us as originals and the conformity to original documents of all documents submitted to us as copies. We have also assumed that there are no agreements or understandings between or among the Company and any participants in the Plans that would expand, modify or otherwise affect the terms of the Plans or the respective rights or obligations of the participants thereunder, and that each award agreement setting forth the terms of each grant of options or other awards under the Plans is consistent with the Plans and has been duly authorized and validly executed and delivered by the parties thereto.

Based upon and subject to the foregoing, it is our opinion that the Plan Shares, when issued and outstanding pursuant to the terms of the applicable Plans and assumed Equity Awards, will be validly issued, fully paid and nonassessable Common Shares.

We consent to the use of this opinion as an exhibit to the Registration Statement.

Very truly yours,

/s/ Morrison & Foerster LLP

Consent of Ernst & Young LLP, Independent Registered Public Accounting Firm

We consent to the incorporation by reference in the Registration Statement (Form S-8) pertaining to the McAfee, Inc. 1997 Stock Incentive Plan, McAfee, Inc. 2010 Equity Incentive Plan, Foundstone, Inc. 2000 Stock Plan, MX Logic, Inc. 2002 Equity Incentive Plan, Secure Computing Corporation (formerly CipherTrust, Inc.) 2000 Stock Option Plan, Secure Computing Corporation 2002 Stock Incentive Plan, Cyberguard Corporation Employee Stock Option Plan, SafeBoot Option Plan 2006, and Network Associates, Inc. 2000 Nonstatutory Stock Option Plan of our reports dated February 18, 2011, with respect to the consolidated financial statements and schedule of Intel Corporation, and the effectiveness of internal control over financial reporting of Intel Corporation included in its Annual Report (Form 10-K) for the year ended December 25, 2010, filed with the Securities and Exchange Commission.

/ s / Ernst & Young LLP

San Jose, California March 17, 2011

1997 STOCK INCENTIVE PLAN, AS AMENDED

ARTICLE 1.

INTRODUCTION

The Plan was adopted by the Board effective December 1, 1997. The purpose of the Plan is to promote the long-term success of the Company and the creation of stockholder value by (a) encouraging Employees, Outside Directors and Consultants to focus on critical long-range objectives, (b) encouraging the attraction and retention of Employees, Outside Directors and Consultants with exceptional qualifications and (c) linking Employees, Outside Directors and Consultants directly to stockholder interests through increased stock ownership. The Plan seeks to achieve this purpose by providing for Awards in the form of Restricted Shares, Stock Units, Options (which may constitute incentive stock options) or stock appreciation rights.

The Plan shall be governed by, and construed in accordance with, the laws of the State of Delaware (except their choice-of-law provisions).

ARTICLE 2.

ADMINISTRATION

2.1 *Committee Composition*. The Plan shall be administered by the Committee. The Committee shall consist exclusively of two or more directors of the Company, who shall be appointed by the Board. In addition, the composition of the Committee shall satisfy:

(a) Such requirements as the Securities and Exchange Commission may establish for administrators acting under plans intended to qualify for exemption under Rule 16b-3 (or its successor) under the Exchange Act; and

(b) Such requirements as the Internal Revenue Service may establish for outside directors acting under plans intended to qualify for exemption under section 162(m)(4)(C) of the Code.

The Board may also appoint one or more separate committees of the Board, each composed of one or more directors of the Company who need not satisfy the foregoing requirements, who may administer the Plan with respect to Employees and Consultants who are not considered officers or directors of the Company under section 16 of the Exchange Act, may grant Awards under the Plan to such Employees and Consultants and may determine all terms of such Awards.

2.2 *Committee Responsibilities.* The Committee shall (a) select the Employees, Outside Directors and Consultants who are to receive Awards under the Plan, (b) determine the type, number, vesting requirements and other features and conditions of such Awards, (c) interpret the Plan and (d) make all other decisions relating to the operation of the Plan, including but not limited to delegation of authority to an administrative committee in keeping with the Plan and applicable laws. The Committee may adopt such rules or guidelines as it deems appropriate to implement the Plan, including rules and regulations relating to sub-plans established for the purpose of qualifying for preferred tax treatment under foreign tax laws and to otherwise accommodate the requirements of local laws and procedures outside the United States. The Committee's determinations under the Plan shall be final and binding on all persons.

2.3 Transfer of Awards Pursuant to Domestic Relations Order. The Committee may structure one or more Awards other than ISOs so that such Award may be assigned in whole or in part during the Participant's lifetime pursuant to a domestic relations order to one or more family members (within the meaning of Rule 701 of the Securities Act of 1933, as amended) of the Participant. With respect to NSOs and SARs, the assigned portion may only be exercised by the person or persons who acquire a proprietary interest in such Award pursuant to the assignment. The terms applicable to the assigned portion of an Award shall be the same as those in effect for the

Award immediately prior to such assignment and shall be set forth in such documents issued to the assignee as the Committee may deem appropriate.

ARTICLE 3.

SHARES AVAILABLE FOR GRANTS

3.1 *Basic Limitation.* Common Shares issued pursuant to the Plan may be authorized but unissued shares or treasury shares. The aggregate number of Options, SARs, Stock Units and Restricted Shares awarded under the Plan shall not exceed (a) 51,475,000 plus (b) the additional Common Shares described in Section 3.3 and (c) any Common Shares subject to stock options or similar awards granted under the Secure Computing Corporation 2002 Stock Incentive Plan, the Secure Computing Corporation (formerly CipherTrust, Inc.) 2000 Stock Option Plan, the Cyberguard Corporation Third Amended and Restated Employee Stock Option Plan, the Safeboot Option Plan 2006 and the Foundstone, Inc. 2000 Stock Plan (collectively, the " **Acquisition Plans** ") that expire or otherwise terminate without having been exercised in full and Common Shares issued pursuant to awards granted under the Acquisition Plans that are forfeited to or repurchased by the Company, with the maximum number of Common Shares to be added to the Plan pursuant to this clause (c) equal to 956,351 Common Shares. The limitation of this Section 3.1 shall be subject to adjustment pursuant to Article 10.

3.2 *Full Value Awards*. Any Common Shares subject to Awards other than Options or SARs granted under this Plan or the Predecessor Plan shall be counted against the numerical limits in Section 3.1 as 1.62 Common Shares for every Common Share subject thereto. Further, if Common Shares acquired pursuant to Awards other than Options or SARs granted under this plan or the Predecessor Plan are forfeited to the Company and would otherwise return to the Plan pursuant to Section 3.3, 1.62 times the number of Common Shares so forfeited shall become available for issuance.

3.3 Additional Shares. If Restricted Shares, Stock Units, Options or SARs granted under this Plan or the Predecessor Plan are forfeited or if Options or SARs granted under this Plan or the Predecessor Plan terminate for any other reason before being exercised, then the corresponding Common Shares shall become available for Awards under the Plan. If Stock Units are settled, all Common Shares underlying such Stock Units shall reduce the number available under Section 3.1. If SARs are exercised, all Common Shares underlying such Stock Units shall reduce the number available under Section 3.1. If SARs are exercised, all Common Shares underlying such Stock Units shall reduce the number available under Section 3.1. If SARs are exercised, all Common Shares that may be issued under the Plan upon the exercise of ISOs shall not be increased when Restricted Shares are forfeited. In addition, to the extent that (i) Common Shares are surrendered to the Company in payment of the Exercise Price of an Option, (ii) Common Shares are purchased by the Company in the open market with the proceeds from the sale of Common Shares pursuant to the exercise of Options, or (iii) Restricted Shares are repurchased by the Company at their original purchase price, in each case, such Common Shares shall not be available for issuance under the Plan.

ARTICLE 4.

ELIGIBILITY

4.1 *Incentive Stock Options*. Only Employees who are common-law employees of the Company, a Parent or a Subsidiary on the date of grant shall be eligible for the grant of ISOs. In addition, an Employee who owns more than 10% of the total combined voting power of all classes of outstanding stock of the Company or any of its Parents or Subsidiaries shall not be eligible for the grant of an ISO unless the requirements set forth in section 422(c)(6) of the Code are satisfied.

4.2 Other Grants. Only Employees, Outside Directors and Consultants shall be eligible for the grant of Restricted Shares, Stock Units, NSOs or SARs.

4.3 Prospective Employees and Consultants. For purposes of this Article 4, (a) "Employees" shall include prospective Employees to whom Awards are granted in connection with written offers of employment from the Company, a Parent or a Subsidiary and (b) "Consultants" shall include prospective Consultants to whom Awards are granted in connection with written offers of engagement from the Company, a Parent or a Subsidiary. If an ISO is granted to a prospective Employee, the date when his or her service as an Employee commences shall be deemed to be the date of grant of such ISO for all purposes under the Plan (including, without limitation, Section 5.3). No

Award granted to a prospective Employee or prospective Consultant shall become exercisable or vested unless and until his or her service as an Employee or Consultant commences.

ARTICLE 5.

OPTIONS

5.1 Stock Option Agreement. Each grant of an Option under the Plan shall be evidenced by a Stock Option Agreement between the Optionee and the Company. Such Option shall be subject to all applicable terms of the Plan and may be subject to any other terms that are not inconsistent with the Plan. The Stock Option Agreement shall specify whether the Option is an ISO or an NSO. The provisions of the various Stock Option Agreement may provide that a new Option will be granted automatically to the Optionee when he or she exercises a prior Option and pays the Exercise Price in the form described in Section 6.2.

The date of grant of an Option shall, for all purposes, be the date on which the Board or Committee makes the determination granting such Option or, in the event that the Committee meeting takes place during a period in which the trading window is closed, on such future date as the Board specifies at that time (e.g., three days after the Company's next public earnings announcement). Notice of the determination shall be given to each individual to whom an option is so granted promptly but in no event more than three weeks after the date of such grant. Determination shall be defined as including at a minimum, the number of Options granted to each individual and the terms of such Options.

5.2 *Number of Shares.* Each Stock Option Agreement shall specify the number of Common Shares subject to the Option and shall provide for the adjustment of such number in accordance with Article 10. Options granted to any Optionee in a single fiscal year of the Company shall not cover more than 1,000,000 Common Shares, except that Options granted to a new Employee in the fiscal year of the Company in which his or her service as an Employee first commences shall not cover more than 1,500,000 Common Shares. The limitations set forth in the preceding sentence shall be subject to adjustment in accordance with Article 10.

5.3 *Exercise Price*. Each Stock Option Agreement shall specify the Exercise Price; provided that the Exercise Price of any Option shall in no event be less than 100% of the Fair Market Value of a Common Share on the date of grant. In the case of an NSO, a Stock Option Agreement may specify an Exercise Price that varies in accordance with a predetermined formula while the NSO is outstanding.

5.4 *Exercisability and Term.* Each Stock Option Agreement shall specify the date or event when all or any installment of the Option is to become exercisable. The Stock Option Agreement shall also specify the term of the Option; provided that the term of an Option shall in no event exceed 10 years from the date of grant. A Stock Option Agreement may provide for accelerated exercisability in the event of the Optionee's death, disability or retirement or other events and may provide for expiration prior to the end of its term in the event of the Optionee's service. Options may be awarded in combination with SARs, and such an Award may provide that the Options will not be exercisable unless the related SARs are forfeited.

5.5 *Effect of Transfer of Control.* The Committee may determine, at the time of granting an Option or thereafter, that such Option shall become exercisable as to all or part of the Common Shares subject to such Option in the event that a Transfer of Control occurs with respect to the Company, except that the exercisability of an ISO shall not be accelerated without the Optionee's written consent. In addition, a separate agreement between the Optionee and the Company may provide that such Optionee's Options shall become exercisable as to all or part of the Common Shares subject to such Options in the event that a Transfer of Control occurs with respect to the Company.

5.6 Substitution of Options. The Committee may accept the cancellation of outstanding options granted by another issuer in return for the grant of new Options under the Plan for the same or a different number of Common Shares and at the same or a different Exercise Price.

5.7 Buyout Provisions. The Committee may at any time (a) offer to buy out for a payment in cash, cash equivalents, Shares or other form of equity compensation available under the Company's stock plans, an Option

previously granted or (b) authorize an Optionee to elect to cash out an Option previously granted, in either case at such time and based upon such terms and conditions as the Committee shall establish and in compliance with applicable laws. Notwithstanding anything contained in this Section 5.7 to the contrary, the Committee shall not be allowed to authorize a buyout of underwater options or stock appreciation rights without the prior consent of the Company's stockholders.

ARTICLE 6.

PAYMENT FOR OPTION SHARES

6.1 *General Rule.* The entire Exercise Price of Common Shares issued upon exercise of Options shall be payable in cash or cash equivalents at the time when such Common Shares are purchased, except as follows:

(a) In the case of an ISO granted under the Plan, payment shall be made only pursuant to the express provisions of the applicable Stock Option Agreement. The Stock Option Agreement may specify that payment may be made in any form(s) described in this Article 6.

(b) In the case of an NSO, the Committee may at any time accept payment in any form(s) described in this Article 6 to the extent permissible under applicable laws.

6.2 *Surrender of Stock.* To the extent that this Section 6.2 is applicable, all or any part of the Exercise Price may be paid by surrendering, or attesting to the ownership of, Common Shares that are already owned by the Optionee. Such Common Shares shall be valued at their Fair Market Value on the date when the new Common Shares are purchased under the Plan. The Optionee shall not surrender, or attest to the ownership of, Common Shares in payment of the Exercise Price if such action would cause the Company to recognize compensation expense (or additional compensation expense) with respect to the Option for financial reporting purposes.

6.3 *Exercise/Sale.* To the extent that this Section 6.3 is applicable, all or any part of the Exercise Price and any withholding taxes may be paid by delivering (on a form prescribed by the Company) an irrevocable direction to a securities broker approved by the Company to sell all or part of the Common Shares being purchased under the Plan and to deliver all or part of the sales proceeds to the Company.

6.4 Exercise/Pledge. To the extent that this Section 6.4 is applicable, all or any part of the Exercise Price and any withholding taxes may be paid by delivering (on a form prescribed by the Company) an irrevocable direction to pledge all or part of the Common Shares being purchased under the Plan to a securities broker or lender approved by the Company, as security for a loan, and to deliver all or part of the loan proceeds to the Company.

6.5 *Promissory Note.* To the extent that this Section 6.5 is applicable, all or any part of the Exercise Price and any withholding taxes may be paid by delivering (on a form prescribed by the Company) a full-recourse promissory note; provided that the par value of the Common Shares being purchased under the Plan shall be paid in cash or cash equivalents.

6.6 Other Forms of Payment. To the extent that this Section 6.6 is applicable, all or any part of the Exercise Price and any withholding taxes may be paid in any other form that is consistent with applicable laws, regulations and rules.

ARTICLE 7.

STOCK APPRECIATION RIGHTS

7.1 *SAR Agreement.* Each grant of an SAR under the Plan shall be evidenced by an SAR Agreement between the Optionee and the Company. Such SAR shall be subject to all applicable terms of the Plan and may be subject to any other terms that are not inconsistent with the Plan. The provisions of the various SAR Agreements entered into under the Plan need not be identical. SARs may be granted in consideration of a reduction in the Optionee's other compensation to the extent permissible under applicable laws.

7.2 Number of Shares. Each SAR Agreement shall specify the number of Common Shares to which the SAR pertains and shall provide for the adjustment of such number in accordance with Article 10. SARs granted to any

Optionee in a single calendar year shall in no event pertain to more than 1,000,000 Common Shares, except that SARs granted to a new Employee in the fiscal year of the Company in which his or her service as an Employee first commences shall not pertain to more than 1,500,000 Common Shares. The limitations set forth in the preceding sentence shall be subject to adjustment in accordance with Article 10.

7.3 *Exercise Price*. Each SAR Agreement shall specify the Exercise Price; provided that the Exercise Price of any SAR shall in no event be less than 100% of the Fair Market Value of a Common Share on the date of grant. An SAR Agreement may specify an Exercise Price that varies in accordance with a predetermined formula while the SAR is outstanding.

7.4 *Exercisability and Term.* Each SAR Agreement shall specify the date when all or any installment of the SAR is to become exercisable. The SAR Agreement shall also specify the term of the SAR; provided that the term of a SAR shall in no event exceed 10 years from the date of grant. An SAR Agreement may provide for accelerated exercisability in the event of the Optionee's death, disability or retirement or other events and may provide for expiration prior to the end of its term in the event of the termination of the Optionee's service. SARs may be awarded in combination with Options, and such an Award may provide that the SARs will not be exercisable unless the related Options are forfeited. An SAR may be included in an ISO only at the time of grant but may be included in an NSO at the time of grant or thereafter. An SAR granted under the Plan may provide that it will be exercisable only in the event of a Transfer of Control.

7.5 *Effect of Transfer of Control.* The Committee may determine, at the time of granting an SAR or thereafter, that such SAR shall become exercisable as to all or part of the Common Shares subject to such SAR in the event that a Transfer of Control occurs with respect to the Company. In addition, a separate agreement between the Optionee and the Company may provide that such Optionee's SARs shall become exercisable as to all or part of the Common Shares subject to such SARs in the event that a Transfer of Control occurs with respect to the Common Shares subject to such SARs in the event that a Transfer of Control occurs with respect to the Common Shares subject to such SARs in the event that a Transfer of Control occurs with respect to the Company.

7.6 *Exercise of SARs.* Upon exercise of an SAR, the Optionee (or any person having the right to exercise the SAR after his or her death) shall receive from the Company (a) Common Shares, (b) cash or (c) a combination of Common Shares and cash, as the Committee shall determine. The amount of cash and/or the Fair Market Value of Common Shares received upon exercise of SARs shall, in the aggregate, be equal to the amount by which the Fair Market Value (on the date of surrender) of the Common Shares subject to the SARs exceeds the Exercise Price. If, on the date when an SAR expires, the Exercise Price under such SAR is less than the Fair Market Value on such date but any portion of such SAR has not been exercised or surrendered, then such SAR shall automatically be deemed to be exercised as of such date with respect to such portion.

7.7 Modification or Assumption of SARs. Within the limitations of the Plan, the Committee may modify, extend or assume outstanding SARs or may accept the cancellation of outstanding SARs (whether granted by the Company or by another issuer) in return for the grant of new SARs for the same or a different number of shares and at the same or a different exercise price. The foregoing notwithstanding, no modification of an SAR shall, without the consent of the Optionee, alter or impair his or her rights or obligations under such SAR.

ARTICLE 8.

RESTRICTED SHARES

8.1 *Restricted Stock Agreement.* Each grant of Restricted Shares under the Plan shall be evidenced by a Restricted Stock Agreement between the recipient and the Company. Such Restricted Shares shall be subject to all applicable terms of the Plan and may be subject to any other terms that are not inconsistent with the Plan. The provisions of the various Restricted Stock Agreements entered into under the Plan need not be identical. Restricted Shares may be granted in consideration of a reduction in the recipient's other compensation.

8.2 Payment for Awards. To the extent that an Award is granted in the form of newly issued Restricted Shares, the Award recipient, as a condition to the grant of such Award, shall be required to pay the Company in cash or cash equivalents an amount equal to the par value of such Restricted Shares. To the extent that an Award is granted in the form of Restricted Shares from the Company's treasury, no cash consideration shall be required of the

Award recipients. Any amount not paid in cash may be paid with a full-recourse promissory note, subject to applicable laws allowing such form of payment.

8.3 Vesting Conditions. Each Award of Restricted Shares may or may not be subject to vesting. Vesting shall occur, in full or in installments, upon satisfaction of the conditions specified in the Restricted Stock Agreement. The Committee may include among such conditions the requirement that the performance of the Company or a business unit of the Company for a specified period of one or more years equal or exceed a target determined in advance by the Committee. Such performance shall be determined by the Committee. Such target shall be based on one or more of the criteria set forth in Appendix A. The Committee shall determine such target not later than the 90th day of such period. In no event shall the number of Restricted Shares which are subject to performance-based vesting conditions and which are granted to any Participant in a single calendar year exceed 300,000, subject to adjustment in accordance with Article 10. A Restricted Stock Agreement may provide for accelerated vesting in the event of the Participant's death, disability or retirement or other events.

8.4 *Effect of Transfer of Control.* The Committee may determine, at the time of granting Restricted Shares or thereafter, that all or part of such Restricted Shares shall become vested in the event that a Transfer of Control occurs with respect to the Company. In addition, a separate agreement between the Participant and the Company may provide that all or part of such Participant's Restricted Shares shall become vested in the event that a Transfer of Control occurs with respect to the company. In addition, a separate agreement between the Participant and the Company may provide that all or part of such Participant's Restricted Shares shall become vested in the event that a Transfer of Control occurs with respect to the Company.

8.5 Voting and Dividend Rights. The holders of Restricted Shares awarded under the Plan shall have the same voting, dividend and other rights as the Company's other stockholders. A Restricted Stock Agreement, however, may require that the holders of Restricted Shares invest any cash dividends received in additional Restricted Shares. Such additional Restricted Shares shall be subject to the same conditions and restrictions as the Award with respect to which the dividends were paid.

ARTICLE 9.

STOCK UNITS

9.1 Stock Unit Agreement. Each grant of Stock Units under the Plan shall be evidenced by a Stock Unit Agreement between the recipient and the Company. Such Stock Units shall be subject to all applicable terms of the Plan and may be subject to any other terms that are not inconsistent with the Plan. The provisions of the various Stock Unit Agreements entered into under the Plan need not be identical. Stock Units may be granted in consideration of a reduction in the recipient's other compensation.

9.2 Payment for Awards. To the extent that an Award is granted in the form of Stock Units, no cash consideration shall be required of the Award recipients.

9.3 Vesting Conditions . Each Award of Stock Units may or may not be subject to vesting. Vesting shall occur, in full or in installments, upon satisfaction of the conditions specified in the Stock Unit Agreement. The Committee may include among such conditions the requirement that the performance of the Company or a business unit of the Company for a specified period of one or more years equal or exceed a target determined in advance by the Committee. Such performance shall be determined by the Committee. Such target shall be based on one or more of the criteria set forth in Appendix A. The Committee shall determine such target not later than the 90th day of such period. In no event shall the number of Stock Units which are subject to performance-based vesting conditions and which are granted to any Participant in a single calendar year exceed 300,000, subject to adjustment in accordance with Article 10. A Stock Unit Agreement may provide for accelerated vesting in the event of the Participant's death, disability or retirement or other events.

9.4 *Effect of Transfer of Control.* The Committee may determine, at the time of granting Stock Units or thereafter, that all or part of such Stock Units shall become vested in the event that a Transfer of Control occurs with respect to the Company. In addition, a separate agreement between the Participant and the Company may provide that all or part of such Participant's Stock Units shall become vested in the event that a Transfer of Control occurs with respect to the Company.

9.5 Voting and Dividend Rights. The holders of Stock Units shall have no voting rights. Prior to settlement or forfeiture, any Stock Unit awarded under the Plan may, at the Committee's discretion, carry with it a right to dividend equivalents. Such right entitles the holder to be credited with an amount equal to all cash dividends paid on one Common Share while the Stock Unit is outstanding. Dividend equivalents may be converted into additional Stock Units. Settlement of dividend equivalents may be made in the form of cash, in the form of Common Shares, or in a combination of both. Prior to distribution, any dividend equivalents which are not paid shall be subject to the same conditions and restrictions as the Stock Units to which they attach.

9.6 Form and Time of Settlement of Stock Units. Settlement of vested Stock Units may be made in the form of (a) cash, (b) Common Shares or (c) any combination of both, as determined by the Committee. The actual number of Stock Units eligible for settlement may be larger or smaller than the number included in the original Award, based on predetermined performance factors. Methods of converting Stock Units into cash may include (without limitation) a method based on the average Fair Market Value of Common Shares over a series of trading days. Vested Stock Units may be settled in a lump sum or in installments. The distribution may occur or commence when all vesting conditions applicable to the Stock Units have been satisfied or have lapsed, or it may be deferred to any later date. The amount of a deferred distribution may be increased by an interest factor or by dividend equivalents. Until an Award of Stock Units is settled, the number of such Stock Units shall be subject to adjustment pursuant to Article 10.

9.7 Death of Recipient. Any Stock Units Award that becomes payable after the recipient's death shall be distributed to the recipient's beneficiary or beneficiaries. Each recipient of a Stock Units Award under the Plan may designate one or more beneficiaries for this purpose by filing the prescribed form with the Company. A beneficiary designation may be changed by filing the prescribed form with the Company at any time before the Award recipient's death. If no beneficiary was designated or if no designated beneficiary survives the Award recipient, then any Stock Units that become payable after the recipient's death shall be distributed to the recipient's estate.

9.8 Creditors' Rights. A holder of Stock Units shall have no rights other than those of a general creditor of the Company. Stock Units represent an unfunded and unsecured obligation of the Company, subject to the terms and conditions of the applicable Stock Unit Agreement.

ARTICLE 10.

PROTECTION AGAINST DILUTION

10.1 *Adjustments.* In the event of a subdivision of the outstanding Common Shares, a declaration of a dividend payable in Common Shares, a declaration of a dividend payable in a form other than Common Shares in an amount that has a material effect on the price of Common Shares, a combination or consolidation of the outstanding Common Shares (by reclassification or otherwise) into a lesser number of Common Shares, a recapitalization, a spin-off or a similar occurrence, the Committee shall make such adjustments as it, in its sole discretion, deems appropriate in one or more of:

- (b) The limitations set forth in Sections 5.2, 7.2, 8.3 and 9.3;
- (c) The number of Common Shares covered by each outstanding Option and SAR;
- (d) The Exercise Price under each outstanding Option and SAR; or
- (e) The number of Stock Units included in any prior Award which has not yet been settled.

Except as provided in this Article 10, a Participant shall have no rights by reason of any issue by the Company of stock of any class or securities convertible into stock of any class, any subdivision or consolidation of shares of stock of any class, the payment of any stock dividend or any other increase or decrease in the number of shares of stock of any class.

⁽a) The number of Options, SARs, Restricted Shares and Stock Units available for future Awards under Article 3;

10.2 Dissolution or Liquidation. To the extent not previously exercised or settled, Options, SARs and Stock Units shall terminate immediately prior to the dissolution or liquidation of the Company.

10.3 *Reorganizations*. In the event that the Company is a party to a merger or other reorganization, outstanding Awards shall be subject to the agreement of merger or reorganization. Such agreement shall provide for (a) the continuation of the outstanding Awards by the Company, if the Company is a surviving corporation, (b) the assumption of the outstanding Awards by the surviving corporation or its parent or subsidiary, (c) the substitution by the surviving corporation or its parent or subsidiary of its own awards for the outstanding Awards, (d) full exercisability or vesting and accelerated expiration of the outstanding Awards or (e) settlement of the full value of the outstanding Awards in cash or cash equivalents followed by cancellation of such Awards. If such agreement does not provide for any of the above, all outstanding Awards shall fully vest, become immediately exercisable or payable or have all restrictions lifted as may apply to the type of Award upon or after a Change in Control unless otherwise determined by the Committee.

ARTICLE 11.

DEFERRAL OF AWARDS

The Committee (in its sole discretion) may permit or require a Participant to:

(a) Have cash that otherwise would be paid to such Participant as a result of the exercise of an SAR or the settlement of Stock Units credited to a deferred compensation account established for such Participant by the Committee as an entry on the Company's books;

(b) Have Common Shares that otherwise would be delivered to such Participant as a result of the exercise of an Option or SAR converted into an equal number of Stock Units; or

(c) Have Common Shares that otherwise would be delivered to such Participant as a result of the exercise of an Option or SAR, or the settlement of Stock Units, converted into amounts credited to a deferred compensation account established for such Participant by the Committee as an entry on the Company's books. Such amounts shall be determined by reference to the Fair Market Value of such Common Shares as of the date when they otherwise would have been delivered to such Participant.

A deferred compensation account established under this Article 11 may be credited with interest or other forms of investment return, as determined by the Committee. A Participant for whom such an account is established shall have no rights other than those of a general creditor of the Company. Such an account shall represent an unfunded and unsecured obligation of the Company and shall be subject to the terms and conditions of the applicable agreement between such Participant and the Company. If the deferral or conversion of Awards is permitted or required, the Committee (in its sole discretion) may establish rules, procedures and forms pertaining to such Awards, including (without limitation) the settlement of deferred compensation accounts established under this Article 11.

ARTICLE 12.

AWARDS UNDER OTHER PLANS

The Company may grant awards under other plans or programs. Such awards may be settled in the form of Common Shares issued under this Plan. Such Common Shares shall be treated for all purposes under the Plan like Common Shares issued in settlement of Stock Units and shall, when issued, reduce the number of Common Shares available under Article 3.

ARTICLE 13.

LIMITATION ON RIGHTS

13.1 Retention Rights. Neither the Plan nor any Award granted under the Plan shall be deemed to give any individual a right to remain an Employee, Outside Director or Consultant. The Company and its Parents,



Subsidiaries and Affiliates reserve the right to terminate the service of any Employee or Consultant at any time, with or without cause, subject to applicable laws and a written employment agreement (if any).

13.2 Stockholders' Rights. A Participant shall have no dividend rights, voting rights or other rights as a stockholder with respect to any Common Shares covered by his or her Award prior to the time when a stock certificate for such Common Shares is issued or, if applicable, the time when he or she becomes entitled to receive such Common Shares by filing any required notice of exercise and paying any required Exercise Price. No adjustment shall be made for cash dividends or other rights for which the record date is prior to such time, except as expressly provided in the Plan.

13.3 *Regulatory Requirements.* Any other provision of the Plan notwithstanding, the obligation of the Company to issue Common Shares under the Plan shall be subject to all applicable laws, rules and regulations and such approval by any regulatory body as may be required. The Company reserves the right to restrict, in whole or in part, the delivery of Common Shares pursuant to any Award prior to the satisfaction of all legal requirements relating to the issuance of such Common Shares, to their registration, qualification or listing or to an exemption from registration, qualification or listing.

ARTICLE 14.

WITHHOLDING TAXES

14.1 *General.* To the extent required by applicable federal, state, local or foreign law, a Participant or his or her successor shall make arrangements satisfactory to the Company for the satisfaction of any withholding tax obligations that arise in connection with the Plan. The Company shall not be required to issue any Common Shares or make any cash payment under the Plan until such obligations are satisfied.

14.2 Share Withholding. The Committee may permit a Participant to satisfy all or part of his or her withholding or income tax obligations by having the Company withhold all or a portion of any Common Shares that otherwise would be issued to him or her or by surrendering all or a portion of any Common Shares that he or she previously acquired. Such Common Shares shall be valued at their Fair Market Value on the date when taxes otherwise would be withheld in cash.

ARTICLE 15.

FUTURE OF THE PLAN

15.1 *Term of the Plan.* The Plan, as set forth herein, shall become effective on June 5, 1997. The Plan shall remain in effect until it is terminated under Section 15.2, except that no ISOs shall be granted on or after the 10th anniversary of the later of (a) the date when the Board adopted the Plan or (b) the date when the Board adopted the most recent increase in the number of Common Shares available under Article 3 which was approved by the Company's stockholders.

15.2 Amendment or Termination. The Board or the Committee may, at any time and for any reason, amend, suspend or terminate the Plan. An amendment of the Plan shall be subject to the approval of the Company's stockholders only to the extent required by applicable laws, regulations or rules. No Awards shall be granted under the Plan after the termination thereof. The termination of the Plan, or any amendment thereof, shall not affect any Award previously granted under the Plan.

15.3 No Repricing. The exercise price for the Common Shares to be issued pursuant to an already granted Stock Option or Stock Appreciation Right may not be lowered without the prior consent of the Company's stockholders. This shall include, without limitation, a repricing of the Stock Option or Stock Appreciation Right as well as a Stock Option or Stock Appreciation Right exchange program whereby the Participant agrees to cancel an existing Stock Option or Stock Appreciation Right in exchange for an Option, Stock Appreciation Right or other Award and such transactions described in Section 5.7.

ARTICLE 16.

DEFINITIONS

16.1 "Affiliate" means any entity other than a Subsidiary, if the Company and/or one or more Subsidiaries own not less than 50% of such entity.

16.2 "Award" means any award of an Option, an SAR, a Restricted Share or a Stock Unit under the Plan.

16.3 "Board" means the Company's Board of Directors, as constituted from time to time.

16.4 "Code" means the Internal Revenue Code of 1986, as amended.

16.5 "Committee" means a committee of the Board, as described in Article 2.

16.6 "Common Share" means one share of the Common Stock of the Company.

16.7 "Company" means McAfee, Inc., a Delaware corporation.

16.8 "Consultant" means a consultant or adviser who provides bona fide services to the Company, a Parent, a Subsidiary or an Affiliate as an independent contractor. Service as a Consultant shall be considered employment for all purposes of the Plan, except as provided in Section 4.1.

16.9 "Employee" means a common-law employee of the Company, a Parent, a Subsidiary or an Affiliate.

16.10 "Exchange Act" means the Securities Exchange Act of 1934, as amended.

16.11 "Exercise Price," in the case of an Option, means the amount for which one Common Share may be purchased upon exercise of such Option, as specified in the applicable Stock Option Agreement. "Exercise Price," in the case of an SAR, means an amount, as specified in the applicable SAR Agreement, which is subtracted from the Fair Market Value of one Common Share in determining the amount payable upon exercise of such SAR.

16.12 "Fair Market Value" means the market price of Common Shares, determined by the Committee in good faith on such basis as it deems appropriate. Whenever possible, the determination of Fair Market Value by the Committee shall be based on the prices reported in <u>The Wall Street Journal</u>. Such determination shall be conclusive and binding on all persons.

16.13 "ISO" means an incentive stock option described in section 422(b) of the Code.

16.14 "NSO" means a stock option not described in sections 422 or 423 of the Code.

16.15 "Option" means an ISO or NSO granted under the Plan and entitling the holder to purchase Common Shares.

16.16 "Optionee" means an individual or estate who holds an Option or SAR.

16.17 "Outside Director" shall mean a member of the Board who is not an Employee. Service as an Outside Director shall be considered employment for all purposes of the Plan, except as provided in Section 4.1.

16.18 "Parent" means any corporation (other than the Company) in an unbroken chain of corporations ending with the Company, if each of the corporations other than the Company owns stock possessing 50% or more of the total combined voting power of all classes of stock in one of the other corporations in such chain. A corporation that attains the status of a Parent on a date after the adoption of the Plan shall be considered a Parent commencing as of such date.

16.19 "Participant" means an individual or estate who holds an Award.

16.20 "Plan" means this McAfee, Inc. 1997 Stock Incentive Plan, as amended from time to time.

16.21 "Predecessor Plan" means the McAfee, Inc. 1995 Stock Incentive Plan, as amended.

16.22 "Restricted Share" means a Common Share awarded under the Plan.

16.23 "Restricted Stock Agreement" means the agreement between the Company and the recipient of a Restricted Share which contains the terms, conditions and restrictions pertaining to such Restricted Share.

16.24 "SAR" means a stock appreciation right granted under the Plan.

16.25 "SAR Agreement" means the agreement between the Company and an Optionee which contains the terms, conditions and restrictions pertaining to his or her SAR.

16.26 "Stock Option Agreement" means the agreement between the Company and an Optionee that contains the terms, conditions and restrictions pertaining to his or her Option.

16.27 "Stock Unit" means a bookkeeping entry representing the equivalent of one Common Share, as awarded under the Plan.

16.28 "Stock Unit Agreement" means the agreement between the Company and the recipient of a Stock Unit which contains the terms, conditions and restrictions pertaining to such Stock Unit.

16.29 "Subsidiary" means any corporation (other than the Company) in an unbroken chain of corporations beginning with the Company, if each of the corporations other than the last corporation in the unbroken chain owns stock possessing 50% or more of the total combined voting power of all classes of stock in one of the other corporations in such chain. A corporation that attains the status of a Subsidiary on a date after the adoption of the Plan shall be considered a Subsidiary commencing as of such date.

16.30 "Transfer of Control" shall mean:

(a) The direct or indirect sale or exchange by the stockholders of the Company of all or substantially all of the voting stock of the Company wherein the stockholders of the Company immediately before such sale or exchange do not retain in substantially the same proportions as their ownership of shares of the Company's voting stock immediately before such event, directly or indirectly (including, without limitation, through their ownership of shares of the voting stock of a corporation which, as a result of such sale or exchange, owns the Company either directly or through one or more subsidiaries), at least a majority of the beneficial interest in the voting stock of the Company immediately after such sale or exchange;

(b) A merger or consolidation wherein the stockholders of the Company immediately before such merger or consolidation do not retain in substantially the same proportions as their ownership of shares of the Company's voting stock immediately before such event, directly or indirectly (including, without limitation, through their ownership of shares of the voting stock of a corporation which, as a result of such merger or consolidation, owns the Company either directly or through one or more subsidiaries), at least a majority of the beneficial interest in the voting stock of the Company immediately after such merger or consolidation;

(c) The sale, exchange, or transfer of all or substantially all of the assets of the Company (other than a sale, exchange, or transfer to one or more corporations (the "Transferee Corporation(s)") wherein the stockholders of the Company immediately before such sale, exchange, or transfer retain in substantially the same proportions as their ownership of shares of the Company's voting stock immediately before such event, directly or indirectly (including, without limitation, through their ownership of shares of the voting stock of a corporation which owns the Transferee Corporation(s) either directly or through one or more subsidiaries), at least a majority of the beneficial interest in the voting stock of the Transferee Corporation(s) immediately after such event; or

(d) A liquidation or dissolution of the Company

ARTICLE 17.

EXECUTION

To record the adoption of the Plan by the Board, the Company has caused its duly authorized officer to execute this document in the name of the Company.

McAfee, Inc.

By:

MCAFEE, INC.

2010 EQUITY INCENTIVE PLAN

1. <u>Purposes of the Plan</u>. The purposes of this Plan are to attract and retain the best available personnel for positions of substantial responsibility, to provide incentives to individuals who perform services to the Company, and to promote the success of the Company's business.

The Plan permits the grant of Incentive Stock Options, Nonstatutory Stock Options, Stock Appreciation Rights, Restricted Stock, Stock Units, Performance Units and Performance Shares.

2. Definitions . As used herein, the following definitions will apply:

(a) "Administrator " means the Board or any of its Committees as will be administering the Plan, in accordance with Section 4 of the Plan.

(b) "<u>Affiliate</u>" means any corporation or any other entity (including, but not limited to, partnerships and joint ventures) controlling, controlled by, or under common control with the Company.

(c) "<u>Applicable Laws</u>" means the requirements relating to the administration of equity-based awards under U.S. state corporate laws, U.S. federal and state securities laws, the Code, any stock exchange or quotation system on which the Common Stock is listed or quoted and the applicable laws of any foreign country or jurisdiction where Awards are, or will be, granted under the Plan.

(d) "Award " means, individually or collectively, a grant under the Plan of Options, Stock Appreciation Rights, Restricted Stock, Stock Units, Performance Units or Performance Shares.

(e) "<u>Award Agreement</u>" means the written or electronic agreement setting forth the terms and provisions applicable to each Award granted under the Plan. The Award Agreement is subject to the terms and conditions of the Plan.

(f) "<u>Award Transfer Program</u>" means any program instituted by the Administrator that would permit Participants the opportunity to transfer for value any outstanding Awards to a financial institution or other person or entity approved by the Administrator.

(g) "Board " means the Board of Directors of the Company.

(h) " Change in Control " means the occurrence of any of the following events:

(i) <u>Change in Ownership of the Company</u>. A change in the ownership of the Company which occurs on the date that any one person, or more than one person acting as a group ("<u>Person</u>"), acquires ownership of the stock of the Company that, together with the stock held by such Person, constitutes more than fifty percent (50%) of the total voting power of the stock of the Company; provided, however, that for purposes of this subsection (i), the acquisition of additional stock by any one Person, who is considered to own more than fifty percent (50%) of the total voting power of the stock of the Company will not be considered a Change in Control; or

(ii) <u>Change in Effective Control of the Company</u>. If the Company has a class of securities registered pursuant to Section 12 of the Exchange Act, a change in the effective control of the Company which occurs on the date that a majority of members of the Board is replaced during any twelve (12) month period by Directors whose appointment or election is not endorsed by a majority of the members of the Board prior to the date of the appointment or election. For purposes of this subsection (ii), if any Person is considered to be in effective control of the Company, the acquisition of additional control of the Company by the same Person will not be considered a Change in Control; or

(iii) <u>Change in Ownership of a Substantial Portion of the Company's Assets</u>. A change in the ownership of a substantial portion of the Company's assets which occurs on the date that any Person acquires (or has acquired during the twelve (12) month period ending on the date of the most recent

acquisition by such person or persons) assets from the Company that have a total gross fair market value equal to or more than fifty percent (50%) of the total gross fair market value of all of the assets of the Company immediately prior to such acquisition or acquisitions; provided, however, that for purposes of this subsection (iii), the following will not constitute a change in the ownership of a substantial portion of the Company's assets: (A) a transfer to an entity that is controlled by the Company's stockholders immediately after the transfer, or (B) a transfer of assets by the Company to: (1) a stockholder of the Company (immediately before the asset transfer) in exchange for or with respect to the Company's stock, (2) an entity, fifty percent (50%) or more of the total value or voting power of which is owned, directly or indirectly, by the Company, (3) a Person, that owns, directly or indirectly, fifty percent (50%) or more of the total value or voting power of all the outstanding stock of the Company, or (4) an entity, at least fifty percent (50%) of the total value or voting power of all the subsection (iii)(B)(3). For purposes of this subsection (iii), gross fair market value means the value of the assets of the Company, or the value of the assets being disposed of, determined without regard to any liabilities associated with such assets.

For purposes of this Section 2(h), persons will be considered to be acting as a group if they are owners of a corporation that enters into a merger, consolidation, purchase or acquisition of stock, or similar business transaction with the Company.

Notwithstanding the foregoing, a transaction shall not be deemed a Change in Control unless the transaction qualifies as a change in control event within the meaning of Section 409A of the Code, as it has been and may be amended from time to time, and any proposed or final Treasury Regulations and Internal Revenue Service guidance that has been promulgated or may be promulgated thereunder from time to time.

Further and for the avoidance of doubt, a transaction shall not constitute a Change in Control if: (i) its sole purpose is to change the state of the Company's incorporation, or (ii) its sole purpose is to create a holding company that shall be owned in substantially the same proportions by the persons who held the Company's securities immediately before such transaction.

(i) "<u>Code</u>" means the Internal Revenue Code of 1986, as amended. Reference to a specific section of the Code or Treasury Regulation thereunder will include such section or regulation, any valid regulation or other official applicable guidance promulgated under such section, and any comparable provision of any future legislation or regulation amending, supplementing or superseding such section or regulation.

(j) "Committee "means a committee of Directors or of other individuals satisfying Applicable Laws appointed by the Board in accordance with Section 4 hereof.

- (k) "Common Stock " means the common stock of the Company.
- (1) "Company "means McAfee, Inc., a Delaware corporation, or any successor thereto.

(m) "Consultant" means any person, including an advisor, engaged by the Company or its Affiliates to render services to such entity other than as an Employee.

(n) "Determination Date" means the latest possible date that will not jeopardize the qualification of an Award granted under the Plan as "performance-based compensation" under Section 162(m) of the Code.

(o) "Director" means a member of the Board.

(p) "Disability" means total and permanent disability as defined in Section 22(e)(3) of the Code, provided that in the case of Awards other than Incentive Stock Options, the Administrator in its discretion may determine whether a permanent and total disability exists in accordance with uniform and non-discriminatory standards adopted by the Administrator from time to time.

(q) "Employee " means any person, including Officers and Directors, employed by the Company or its Affiliates. Neither service as a Director nor payment of a Director's fee by the Company will be sufficient to constitute "employment" by the Company.

(r) "Exchange Act " means the Securities Exchange Act of 1934, as amended.

(s) "Fair Market Value" means, as of any date, the value of Common Stock determined as follows:

(i) If the Common Stock is listed on any established stock exchange or a national market system, including without limitation the Nasdaq Global Market, the Nasdaq Global Select Market or the Nasdaq Capital Market, its Fair Market Value shall be the closing sales price for such stock (or, if no closing sales price was reported on that date, as applicable, on the last trading date such closing sales price is reported) as quoted on such exchange or system on the day of determination, as reported in *The Wall Street Journal* or such other source as the Administrator deems reliable;

(ii) If the Common Stock is regularly quoted by a recognized securities dealer but selling prices are not reported, its Fair Market Value shall be the mean between the high bid and low asked prices for the Common Stock on the day of determination (or, if no bids and asks were reported on that date, as applicable, on the last trading date such bids and asks are reported); or

(iii) In the absence of an established market for the Common Stock, the Fair Market Value will be determined in good faith by the Administrator.

(t) "Fiscal Year" means the fiscal year of the Company.

(u) "Incentive Stock Option " means an Option that by its terms qualifies and is otherwise intended to qualify as an incentive stock option within the meaning of Section 422 of the Code and the regulations promulgated thereunder.

(v) "Nonstatutory Stock Option " means an Option that by its terms does not qualify or is not intended to qualify as an Incentive Stock Option.

(w) "Officer " means a person who is an officer of the Company within the meaning of Section 16 of the Exchange Act and the rules and regulations promulgated thereunder.

(x) " Option " means a stock option granted pursuant to the Plan.

(y) "Parent "means a "parent corporation," whether now or hereafter existing, as defined in Section 424(e) of the Code.

(z) "Participant" means the holder of an outstanding Award.

(aa) "Performance Goals" will have the meaning set forth in Section 11 of the Plan.

(bb) "Performance Period" means any Fiscal Year of the Company or such longer or shorter period as determined by the Administrator in its sole discretion.

(cc) "*Performance Share*" means an Award denominated in Shares which may be earned in whole or in part upon attainment of performance goals or other vesting criteria as the Administrator may determine pursuant to Section 10.

(dd) "*Performance Unit*" means an Award which may be earned in whole or in part upon attainment of performance goals or other vesting criteria as the Administrator may determine and which may be settled for cash, Shares or other securities or a combination of the foregoing pursuant to Section 10.

(ee) "*Period of Restriction*" means the period during which the transfer of Shares of Restricted Stock are subject to restrictions and therefore, the Shares are subject to a substantial risk of forfeiture. Such restrictions may be based on the passage of time, the achievement of target levels of performance, or the occurrence of other events as determined by the Administrator.

(ff) "Plan " means this 2010 Equity Incentive Plan.

(gg) "Restricted Stock " means Shares issued pursuant to a Restricted Stock Award under Section 8 of the Plan, or issued pursuant to the early exercise of an Option.

(hh) "Rule 16b-3 " means Rule 16b-3 of the Exchange Act or any successor to Rule 16b-3, as in effect when discretion is being exercised with respect to the Plan.
(ii) "Section 16(b) "means Section 16(b) of the Exchange Act.

(jj) "Service Provider" means an Employee, Director or Consultant.

(kk) "Share" means a share of the Common Stock, as adjusted in accordance with Section 15 of the Plan.

(II) "Stock Appreciation Right" means an Award, granted alone or in connection with an Option, that pursuant to Section 7 is designated as a Stock Appreciation Right.

(mm) "Stock Unit " means a bookkeeping entry representing an amount equal to the Fair Market Value of one Share, granted pursuant to Section 9. Each Stock Unit represents an unfunded and unsecured obligation of the Company.

(nn) "Subsidiary means a "subsidiary corporation", whether now or hereafter existing, as defined in Section 424(f) of the Code.

3. Stock Subject to the Plan .

(a) Subject to the provisions of Section 15 of the Plan, the maximum aggregate number of Shares that may be issued under the Plan is twelve million six hundred thousand (12,600,000) Shares plus: (i) the additional Shares described in Section 3(c), (ii) any Shares available for issuance under the Company 1997 Stock Incentive Plan (the "1997 Plan") and (iii) any Shares subject to stock options or similar awards granted under the MX Logic, Inc. 2002 Equity Incentive Plan, the Secure Computing Corporation 2002 Stock Incentive Plan, the Secure Computing Corporation (formerly CipherTrust, Inc.) 2000 Stock Option Plan, the Cyberguard Corporation Third Amended and Restated Employee Stock Option Plan, the Safeboot Option Plan 2006 and the Foundstone, Inc. 2000 Stock Plan (collectively, the "Acquisition Plans") and/or the 1997 Plan that expire or otherwise terminate without having been exercised in full and Shares issued pursuant to awards granted under the Acquisition Plans and/or the 1997 Plan that are forfeited to or repurchased by the Company, with the maximum number of Shares to be added to the Plan pursuant to this clause (iii) equal to 14,474,986 Shares). The Shares may be authorized, but unissued, or reacquired Common Stock.

(b) *Full Value Awards*. Any Shares subject to Awards of Restricted Stock, Stock Units, Performance Units, and Performance Shares will be counted against the numerical limits of this Section 3 as two and forty-three hundredths (2.43) Shares for every one Share subject thereto. Further, if Shares acquired pursuant to any such Award are forfeited or repurchased by the Company and would otherwise return to the Plan pursuant to Section 3(c), two and forty-three hundredths (2.43) times the number of Shares so forfeited or repurchased will return to the Plan and will again become available for issuance.

(c) *Lapsed Awards*. If an Award expires or becomes unexercisable without having been exercised in full or, with respect to Restricted Stock, Stock Units, Performance Units or Performance Shares, is forfeited to or repurchased by the Company due to failure to vest, the unpurchased Shares (or for Awards other than Options or Stock Appreciation Rights the forfeited, repurchased, or unissued Shares) which were subject thereto will become available for future grant or sale under the Plan (unless the Plan has terminated). Shares purchased by the Company in the open market with the proceeds from the sale of Shares pursuant to the exercise of Options will not be available for issuance under the Plan. Upon the exercise of a Stock Appreciation Right settled in Shares, the gross number of Shares covered by the portion of the Award so exercised will cease to be available for future grant or sale under the Plan. To the extent an Award under the Plan is paid out in cash rather than Shares, such cash payment will not become available for future distribution under the Plan. Shares that have actually been issued under the Plan under any Award will not be returned to the Plan and will not become available for future distribution under the Plan, shares that have actually been issued under the Plan under any Award will not be come available for future grant under the Plan, so are forfeited to the Company due to failure to vest, such Shares will become available for future grant under the Plan. Notwithstanding the foregoing and, subject to adjustment as provided in Section 15, the maximum number of Shares that may be issued upon the exercise of Incentive Stock Options will equal the aggregate Share number stated in Section 3(a), plus, to the extent allowable under Section 422 of the Code and the Treasury Regulations promulgated thereunder, any Shares that become available for issuance under the Plan under this Section 3(c).

(d) <u>Share Reserve</u>. The Company, during the term of this Plan, will at all times reserve and keep available such number of Shares as will be sufficient to satisfy the requirements of the Plan.

4. Administration of the Plan.

(a) Procedure.

(i) <u>Multiple Administrative Bodies</u>. Different Committees with respect to different groups of Service Providers may administer the Plan.

(ii) <u>Section 162(m)</u>. To the extent that the Administrator determines it to be desirable to qualify Awards granted hereunder as "performance-based compensation" within the meaning of Section 162(m) of the Code, the Plan will be administered by a Committee of two (2) or more "outside directors" within the meaning of Section 162(m) of the Code. For the avoidance of doubt, to the extent that the Administrator is providing for the grant, and/or the administration of, Awards that are not intended to qualify as "performance-based compensation" under Section 162(m) of the Code, the Plan may be administered by a Committee that consists of other than two (2) or more "outside directors" within the meaning of Section 162(m) of the Code.

(iii) <u>Rule 16b-3</u>. To the extent desirable to qualify transactions hereunder as exempt under Rule 16b-3, the transactions contemplated hereunder will be structured to satisfy the requirements for exemption under Rule 16b-3.

(iv) <u>Other Administration</u>. Other than as provided above, the Plan will be administered by (A) the Board or (B) a Committee, which committee will be constituted to satisfy Applicable Laws.

(b) <u>Powers of the Administrator</u>. Subject to the provisions of the Plan, and in the case of a Committee, subject to the specific duties delegated by the Board to such Committee, the Administrator will have the authority, in its discretion:

- (i) to determine the Fair Market Value;
- (ii) to select the Service Providers to whom Awards may be granted hereunder;
- (iii) to determine the number of Shares to be covered by each Award granted hereunder;

(iv) to approve forms of Award Agreements for use under the Plan and to expressly delegate to the Officers the authority to approve and modify such forms of Award Agreement and/or individual Award Agreements;

(v) to determine the terms and conditions, not inconsistent with the terms of the Plan, of any Award granted hereunder. Such terms and conditions include, but are not limited to, the exercise price, the time or times when Awards may be exercised (which may be based on performance criteria), any vesting acceleration or waiver of forfeiture restrictions, and any restriction or limitation regarding any Award or the Shares relating thereto, based in each case on such factors as the Administrator will determine;

(vi) to construe and interpret the terms of the Plan and Awards granted pursuant to the Plan;

(vii) to prescribe, amend and rescind rules and regulations relating to the Plan, including rules and regulations relating to sub-plans established for the purpose of satisfying applicable foreign laws or for qualifying for favorable tax treatment under applicable foreign laws. Pursuant to subsection 4(b)(iv), this authority may be expressly delegated to the Officers;

(viii) to modify or amend each Award (subject to Section(s) 4(e)(i) and 20(c) of the Plan), including but not limited to the discretionary authority to extend the posttermination exercisability period of Awards and to extend the maximum term of an Option (subject to Section 6(e));

(ix) to allow Participants to satisfy withholding tax obligations in such manner as prescribed in Section 16;

(x) to authorize any person to execute on behalf of the Company any instrument required to effect the grant of an Award previously granted by the Administrator;

(xi) to allow a Participant to defer the receipt of the payment of cash or the delivery of Shares that would otherwise be due to such Participant under an Award pursuant to such procedures as the Administrator may determine; and

(xii) to make all other determinations deemed necessary or advisable for administering the Plan.

(c) <u>Effect of Administrator's Decision</u>. The Administrator's decisions, determinations and interpretations will be final and binding on all Participants and any other holders of Awards.

(d) <u>No Liability</u>. Under no circumstances shall the Company, its Affiliates, the Administrator, or the Board incur liability for any indirect, incidental, consequential or special damages (including lost profits) of any form incurred by any person, whether or not foreseeable and regardless of the form of the act in which such a claim may be brought, with respect to the Plan or the Company's, its Affiliates', the Administrator's or the Board's roles in connection with the Plan.

(e) Limitations .

(i) <u>Prohibition Against Repricing</u>. Notwithstanding Section 4(b)(viii), the Administrator may not modify or amend an Option or Stock Appreciation Right to reduce the exercise price of such Option or Stock Appreciation Right after it has been granted (except for adjustments made pursuant to Section 15), and neither may the Administrator cancel any outstanding Option or Stock Appreciation Right and immediately replace it with any other Award with a lower exercise price, unless such action is approved by stockholders prior to such action being taken.

(ii) <u>Buyout Provisions</u>. The Administrator may at any time offer to buy out for a payment in cash an Option previously granted based on such terms and conditions as the Administrator will establish and communicate to the Participant at the time that such offer is made. Notwithstanding anything contained in this Section 4(e)(ii) to the contrary, the Administrator shall not be allowed to authorize the buyout of underwater Options or Stock Appreciation Rights without the prior consent of the Company's stockholders.

5. <u>Eligibility</u>. Nonstatutory Stock Options, Restricted Stock, Stock Units, Stock Appreciation Rights, Performance Units, and Performance Shares may be granted to Service Providers. Incentive Stock Options may be granted only to employees of the Company or any Parent or Subsidiary of the Company.

6. Stock Options .

(a) <u>Grant of Stock Options</u>. Subject to the terms and conditions of the Plan, an Option may be granted to Service Providers at any time and from time to time as will be determined by the Administrator, in its sole discretion. Each Option will be designated in the Award Agreement as either an Incentive Stock Option or a Nonstatutory Stock Option. However, notwithstanding such designation, to the extent that the aggregate Fair Market Value of the Shares with respect to which Incentive Stock Options are exercisable for the first time by the Participant during any calendar year (under all plans of the Company and any Parent or Subsidiary) exceeds one hundred thousand U.S. dollars (\$100,000), such Options will be treated as Nonstatutory Stock Options. For purposes of this Section 6(a), Incentive Stock Options will be taken into account in the order in which they were granted. The Fair Market Value of the Shares will be determined as of the time the Option with respect to such Shares is granted.

Notwithstanding anything herein to the contrary, the date of grant of an Option shall, for all purposes, be the date on which the Administrator makes the determination granting such Option or, in the event that the Administrator's meeting takes place during a period in which the trading window is closed, on such future date as the Administrator specifies at that time (e.g., two (2) days after the Company's next public earnings announcement). Notice of the determination shall be given to each individual to whom an Option is so granted promptly but in no event more than three (3) weeks after the date of such grant. Determination shall be defined as including at a minimum, the number of Shares subject to Options granted to each individual and the terms of such Options.

(b) <u>Number of Shares</u>. The Administrator will have complete discretion to determine the number of Shares subject to an Option granted to any Participant, provided that during any Fiscal Year, no Participant will be granted Options covering more than 1,000,000 Shares. Notwithstanding the limitation in the previous sentence, in

connection with his or her initial service as an Employee, an Employee may be granted Options covering up to an additional 500,000 Shares. The foregoing limitations will be adjusted proportionately in connection with any change in the Company's capitalization as described in Section 15.

(c) <u>Exercise Price and Other Terms</u>. The Administrator, subject to the provisions of the Plan, will have complete discretion to determine the terms and conditions of Options granted under the Plan, provided, however, that the exercise price will not be less than one hundred percent (100%) of the Fair Market Value of a Share on the date of grant. In addition, in the case of an Incentive Stock Option granted to an employee of the Company or any Parent or Subsidiary of the Company who, at the time the Incentive Stock Option is granted, owns stock representing more than ten percent (10%) of the voting power of all classes of stock of the Company or any Parent or Subsidiary, the per Share exercise price will be no less than one hundred ten percent (110%) of the Fair Market Value per Share on the date of grant. Notwithstanding the foregoing provisions of this Section 6(c), Options may be granted with a per Share exercise price of less than one hundred percent (100%) of the Fair Market Value per Share on the date of grant pursuant to a transaction described in, and in a manner consistent with, Section 424(a) of the Code and the Treasury Regulations thereunder.

(d) Option Agreement.

(i) <u>*Terms and Conditions*</u>. Each Option grant will be evidenced by an Award Agreement that will specify the exercise price, the term of the Option, the acceptable forms of consideration for exercise (which may include any form of consideration permitted by Section 6(d)(ii)), the conditions of exercise, and such other terms and conditions as the Administrator, in its sole discretion, will determine.

(ii) <u>Form of Consideration</u>. The Administrator will determine the acceptable form(s) of consideration for exercising an Option, including the method of payment, to the extent permitted by Applicable Laws. In the case of an Incentive Stock Option, the Administrator will determine the acceptable form of consideration at the time of grant. Such consideration to the extent permitted by Applicable Laws may include, but is not limited to:

(1) cash;

(2) check;

(3) other Shares which have a Fair Market Value on the date of surrender equal to the aggregate exercise price of the Shares as to which said Option will be exercised and provided that accepting such Shares, in the sole discretion of the Administrator, will not result in any adverse accounting consequences to the Company;

(4) by net exercise;

(5) consideration received by the Company under a cashless exercise program implemented by the Company in connection with the Plan;

(6) a reduction in the amount of any Company liability to the Participant, including any liability attributable to the Participant's participation in any Company-sponsored deferred compensation program or arrangement;

(7) such other consideration and method of payment for the issuance of Shares to the extent permitted by Applicable Laws; or

(8) any combination of the foregoing methods of payment.

(e) <u>Term of Option</u>. An Option granted under the Plan will expire upon the date determined by the Administrator, in its sole discretion, and set forth in the Award Agreement; provided, however, that the term will be no more than seven (7) years from the date of grant thereof. In the case of an Incentive Stock Option granted to a Participant who, at the time the Incentive Stock Option is granted, owns stock representing more than ten percent (10%) of the total combined voting power of all classes of stock of the Company or any Parent or Subsidiary, the term of the Incentive Stock Option will be five (5) years from the date of grant or such shorter term as may be provided in the Award Agreement.

(f) Exercise of Option .

(i) <u>Procedure for Exercise; Rights as a Stockholder</u>. Any Option granted hereunder will be exercisable according to the terms of the Plan and at such times and under such conditions as determined by the Administrator and set forth in the Award Agreement. An Option may not be exercised for a fraction of a Share.

An Option will be deemed exercised when the Company receives: (i) notice of exercise (in such form as the Administrator or its delegate specifies from time to time, including, but not limited to an electronic process) from the person entitled to exercise the Option, and (ii) full payment for the Shares with respect to which the Option is exercised (together with applicable tax withholdings). Full payment may consist of any consideration and method of payment authorized by the Administrator or its delegate and permitted by the Award Agreement and the Plan. Shares issued upon exercise of an Option will be issued in the name of the Participant. Until the Shares are issued (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company), no right to vote or receive dividends or any other rights as a stockholder will exist with respect to the Shares subject to an Option, notwithstanding the exercise of the Option. The Company will issue (or cause to be issued) such Shares promptly after the Option is exercised. No adjustment will be made for a dividend or other right for which the record date is prior to the date the Shares are issued, except as provided in Section 15 of the Plan.

Exercising an Option in any manner will decrease the number of Shares thereafter available, both for purposes of the Plan and for sale under the Option, by the number of Shares as to which the Option is exercised.

(ii) <u>Termination of Relationship as a Service Provider</u>. If a Participant ceases to be a Service Provider, other than upon the Participant's termination as the result of the Participant's death or Disability, the Participant may exercise his or her Option within such period of time as is specified in the Award Agreement to the extent that the Option is vested on the date of termination (but in no event later than the expiration of the term of such Option as set forth in the Award Agreement). In the absence of a specified time in the Award Agreement, the Option will remain exercisable for three (3) months following the Participant's termination. Unless otherwise provided by the Administrator, if on the date of termination the Participant is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option will revert to the Plan. If after termination the Participant does not exercise his or her Option within the time specified by the Administrator, the Option will terminate, and the Shares covered by such Option will revert to the Plan.

(iii) <u>Disability of Participant</u>. If a Participant ceases to be a Service Provider as a result of the Participant's Disability, the Participant may exercise his or her Option within such period of time as is specified in the Award Agreement to the extent the Option is vested on the date of termination (but in no event later than the expiration of the term of such Option as set forth in the Award Agreement). In the absence of a specified time in the Award Agreement, the Option will remain exercisable for twelve (12) months following the Participant's termination. Unless otherwise provided by the Administrator, if on the date of termination the Participant is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option will revert to the Plan. If after termination the Participant does not exercise his or her Option within the time specified herein, the Option will terminate, and the Shares covered by such Option will revert to the Plan.

(iv) <u>Death of Participant</u>. If a Participant dies while a Service Provider, the Option may be exercised following the Participant's death within such period of time as is specified in the Award Agreement to the extent that the Option is vested on the date of death (but in no event may the Option be exercised later than the expiration of the term of such Option as set forth in the Award Agreement), by the Participant's designated beneficiary, provided such beneficiary has been designated prior to Participant's death in a form acceptable to the Administrator. If no such beneficiary has been designated by the Participant, then such Option may be exercised by the personal representative of the Participant's estate or by the person(s) to whom the Option is transferred pursuant to the Participant's will or in accordance with the laws of descent and distribution. In the absence of a specified time in the Award Agreement, the Option will remain exercisable for twelve (12) months following Participant's death. Unless otherwise provided by the Administrator, if at the time of death Participant is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option will immediately revert to the Plan. If the Option is not so exercised within the time specified herein, the Option will terminate, and the Shares covered by such Option will revert to the Plan.

(v) <u>Other Termination</u>. A Participant's Award Agreement also may provide that if the exercise of the Option following the termination of Participant's status as a Service Provider (other than upon the Participant's death or Disability) would result in liability under Section 16(b), then the Option will terminate on the earlier of (A) the expiration of the term of the Option set forth in the Award Agreement, or (B) the tenth (10th) day after the last date on which such exercise would result in such liability under Section 16(b). Finally, a Participant's Award Agreement may also provide that if the exercise of the Option following the termination of the Participant's status as a Service Provider (other than upon the Participant's death or Disability) would be prohibited at any time solely because the issuance of Shares would violate the registration requirements under the Securities Act, then the Option will terminate on the earlier of (A) the expiration of the term of the Option, or (B) the expiration of a period of three (3) months after the termination of the Participant's status as a Service Provider during which the exercise of the Option would not be in violation of such registration requirements.

7. Stock Appreciation Rights .

(a) <u>Grant of Stock Appreciation Rights</u>. Subject to the terms and conditions of the Plan, a Stock Appreciation Right may be granted to Service Providers at any time and from time to time as will be determined by the Administrator, in its sole discretion.

Notwithstanding anything herein to the contrary, the date of grant of a Stock Appreciation Right shall, for all purposes, be the date on which the Administrator makes the determination granting such Stock Appreciation Right or, in the event that the Administrator's meeting takes place during a period in which the trading window is closed, on such future date as the Administrator specifies at that time (e.g., two (2) days after the Company's next public earnings announcement). Notice of the determination shall be given to each individual to whom a Stock Appreciation Right is so granted promptly but in no event more than three (3) weeks after the date of such grant. Determination shall be defined as including at a minimum, the number of Shares subject to Stock Appreciation Rights granted to each individual and the terms of such Stock Appreciation Rights.

(b) <u>Number of Shares</u>. The Administrator will have complete discretion to determine the number of Stock Appreciation Rights granted to any Participant, provided that during any Fiscal Year, no Participant will be granted Stock Appreciation Rights covering more than 1,000,000 Shares. Notwithstanding the limitation in the previous sentence, in connection with his or her initial service as an Employee, an Employee may be granted Stock Appreciation Rights covering up to an additional 1,000,000 Shares. The foregoing limitations will be adjusted proportionately in connection with any change in the Company's capitalization as described in Section 15.

(c) <u>Exercise Price and Other Terms</u>. The Administrator, subject to the provisions of the Plan, will have complete discretion to determine the terms and conditions of Stock Appreciation Rights granted under the Plan, provided, however, that the exercise price will not be less than one hundred percent (100%) of the Fair Market Value of a Share on the date of grant. Notwithstanding the foregoing provisions of this Section 7(c), Stock Appreciation Rights may be granted with a per Share exercise price of less than one hundred percent (100%) of the Fair Market Value per Share on the date of grant pursuant to a transaction described in, and in a manner consistent with, Section 424(a) of the Code and the Treasury Regulations thereunder.

(d) <u>Stock Appreciation Right Agreement</u>. Each Stock Appreciation Right grant will be evidenced by an Award Agreement that will specify the exercise price, the term of the Stock Appreciation Right, the acceptable forms of consideration for exercise (which may include any form of consideration permitted by Section 6(d)(ii)), the conditions of exercise, and such other terms and conditions as the Administrator, in its sole discretion, will determine.

(e) <u>Expiration of Stock Appreciation Rights</u>. A Stock Appreciation Right granted under the Plan will expire upon the date determined by the Administrator, in its sole discretion, and set forth in the Award Agreement; provided, however, that the term will be no more than seven (7) years from the date of grant thereof. Notwithstanding the foregoing, the rules of Section 6(f) relating to exercise also will apply to Stock Appreciation Rights.

(f) <u>Payment of Stock Appreciation Right Amount</u>. Upon exercise of a Stock Appreciation Right, a Participant will be entitled to receive payment from the Company in an amount determined by multiplying:

(i) The difference between the Fair Market Value of a Share on the date of exercise over the exercise price; times

(ii) The number of Shares with respect to which the Stock Appreciation Right is exercised.

At the discretion of the Administrator, the payment upon Stock Appreciation Right exercise may be in cash, in Shares of equivalent value, or in some combination thereof.

(g) <u>Modification or Assumption of Stock Appreciation Rights</u>. Subject to Section 4(b)(viii) and the provisions of the Plan, the Administrator may modify, extend or assume outstanding Stock Appreciation Rights or may accept the cancellation of outstanding Stock Appreciation Rights (whether granted by the Company or by another issuer) in return for the grant of new Stock Appreciation Rights for the same or a different number of Shares and at the same or a different exercise price. The foregoing notwithstanding, no modification of a Stock Appreciation Right shall, without the consent of the Participant, alter or impair his or her rights or obligations under such Stock Appreciation Right.

8. Restricted Stock.

(a) <u>Grant of Restricted Stock</u>. Subject to the terms and provisions of the Plan, the Administrator, at any time and from time to time, may grant Shares of Restricted Stock to Service Providers in such amounts as the Administrator, in its sole discretion, will determine.

(b) <u>Restricted Stock Agreement</u>. Each Award of Restricted Stock will be evidenced by an Award Agreement that will specify the Period of Restriction, the number of Shares granted, and such other terms and conditions as the Administrator, in its sole discretion, will determine. Unless the Administrator determines otherwise, the Company as escrow agent will hold Shares of Restricted Stock until the restrictions on such Shares have lapsed. Notwithstanding the foregoing sentence, for restricted stock intended to qualify as "performance-based compensation" within the meaning of Section 162(m) of the Code, during any Fiscal Year no Participant will receive more than an aggregate of 500,000 Shares of Restricted Stock. Notwithstanding the limitation in the previous sentence, in connection with his or her initial service as an Employee, an Employee may be granted Shares of Restricted Stock covering up to an additional 500,000 Shares. The foregoing limitations will be adjusted proportionately in connection with any change in the Company's capitalization as described in Section 15.

(c) <u>Transferability</u>. Except as provided in this Section 8, Shares of Restricted Stock may not be sold, transferred, pledged, assigned, or otherwise alienated or hypothecated until the end of the applicable Period of Restriction.

(d) <u>Other Restrictions</u>. The Administrator, in its sole discretion, may impose such other restrictions on Shares of Restricted Stock as it may deem advisable or appropriate.

(e) <u>Removal of Restrictions</u>. Except as otherwise provided in this Section 8, Shares of Restricted Stock covered by each Restricted Stock grant made under the Plan will be released from escrow as soon as practicable after the last day of the Period of Restriction or at such other time as the Administrator may determine. The Administrator, in its sole discretion, may reduce or waive any restrictions for such Award and may accelerate the time at which any restrictions will lapse or be removed.

(f) <u>Voting Rights</u>. During the Period of Restriction, Service Providers holding Shares of Restricted Stock granted hereunder may exercise full voting rights with respect to those Shares, unless the Administrator determines otherwise.

(g) <u>Dividends and Other Distributions</u>. During the Period of Restriction, Service Providers holding Shares of Restricted Stock will be entitled to receive all dividends and other distributions paid with respect to such Shares, unless the Administrator provides otherwise. If any such dividends or distributions are paid in Shares, the Shares will be subject to the same restrictions on transferability and forfeitability as the Shares of Restricted Stock with respect to which they were paid.

(h) <u>Return of Restricted Stock to Company</u>. On the date set forth in the Award Agreement, the Restricted Stock for which restrictions have not lapsed will revert to the Company and again will become available for grant under the Plan.

(i) <u>Section 162(m) Performance Restrictions</u>. For purposes of qualifying grants of Restricted Stock as "performance-based compensation" under Section 162(m) of the Code, the Administrator, in its discretion, may set restrictions based upon the achievement of Performance Goals. The Performance Goals will be set by the Administrator on or before the Determination Date. In granting Restricted Stock which is intended to qualify under Section 162(m) of the Code, the Administrator will follow any procedures determined by it from time to time to be necessary or appropriate to ensure qualification of the Award under Section 162(m) of the Code (e.g., in determining the Performance Goals).

9. Stock Units .

(a) <u>Grant</u>. Stock Units may be granted at any time and from time to time as determined by the Administrator. After the Administrator determines that it will grant Stock Units under the Plan, it will advise the Participant in an Award Agreement of the terms, conditions, and restrictions related to the grant, including the number of Stock Units. Notwithstanding anything to the contrary in this subsection (a), for Stock Units intended to qualify as "performance-based compensation" within the meaning of Section 162(m) of the Code, during any Fiscal Year of the Company, no Participant will receive more than an aggregate of 500,000 Stock Units. Notwithstanding the limitation in the previous sentence, in connection with his or her initial service as an Employee, an Employee may be granted Stock Units covering up to an additional 500,000 Shares. The foregoing limitations will be adjusted proportionately in connection with any change in the Company's capitalization as described in Section 15.

(b) <u>Vesting Criteria and Other Terms</u>. The Administrator will set vesting criteria in its discretion, which, depending on the extent to which the criteria are met, will determine the number of Stock Units that will be paid out to the Participant. The Administrator may set vesting criteria based upon the achievement of Company-wide, business unit, or individual goals (including, but not limited to, continued employment), applicable federal or state securities laws, or any other basis determined by the Administrator in its discretion.

(c) <u>Earning Stock Units</u>. Upon meeting the applicable vesting criteria, the Participant will be entitled to receive a payout as determined by the Administrator. Notwithstanding the foregoing, at any time after the grant of Stock Units, the Administrator, in its sole discretion, may reduce or waive any vesting criteria that must be met to receive a payout and may accelerate the time at which any restrictions will lapse or be removed.

(d) <u>Form and Timing of Payment</u>. Payment of earned Stock Units will be made as soon as practicable after the date(s) set forth in the Award Agreement or as otherwise provided in the applicable Award Agreement or as required by Applicable Laws. The Administrator, in its sole discretion, may pay earned Stock Units in cash, Shares, or a combination thereof. Shares represented by Stock Units that are fully paid in cash again will not reduce the number of Shares available for grant under the Plan.

(e) Cancellation. On the date set forth in the Award Agreement, all unearned Stock Units will be forfeited to the Company.

(f) <u>Section 162(m) Performance Restrictions</u>. For purposes of qualifying grants of Stock Units as "performance-based compensation" under Section 162(m) of the Code, the Administrator, in its discretion, may set restrictions based upon the achievement of Performance Goals. The Performance Goals will be set by the Administrator on or before the Determination Date. In granting Stock Units which are intended to qualify under Section 162(m) of the Code, the Administrator will follow any procedures determined by it from time to time to be necessary or appropriate to ensure qualification of the Award under Section 162(m) of the Code (e.g., in determining the Performance Goals).

10. Performance Units and Performance Shares .

(a) <u>Grant of Performance Units/Shares</u>. Performance Units and Performance Shares may be granted to Service Providers at any time and from time to time, as will be determined by the Administrator, in its sole discretion. The Administrator will have complete discretion in determining the number of Performance Units and Performance Shares granted to each Participant provided that during any Fiscal Year, for Performance Units or

Performance Shares intended to qualify as "performance-based compensation" within the meaning of Section 162(m) of the Code, (i) no Participant will receive Performance Units having an initial value greater than \$5,000,000, and (ii) no Participant will receive more than 500,000 Performance Shares. Notwithstanding the foregoing limitation, in connection with his or her initial service as an Employee, an Employee may be granted: (i) additional Performance Units having an initial value up to \$5,000,000 and (ii) up to an additional 500,000 Performance Shares. The foregoing limitations will be adjusted proportionately in connection with any change in the Company's capitalization as described in Section 15.

(b) <u>Value of Performance Units/Shares</u>. Each Performance Unit will have an initial value that is established by the Administrator on or before the date of grant. Each Performance Share will have an initial value equal to the Fair Market Value of a Share on the date of grant.

(c) <u>Performance Objectives and Other Terms</u>. The Administrator will set performance objectives or other vesting provisions (including, without limitation, continued status as a Service Provider) in its discretion which, depending on the extent to which they are met, will determine the number or value of Performance Units/Shares that will be paid out to the Service Providers. The time period during which the performance objectives or other vesting provisions must be met will be called the "Performance Period." Each Award of Performance Units/Shares will be evidenced by an Award Agreement that will specify the Performance Period, and such other terms and conditions as the Administrator, in its sole discretion, will determine. The Administrator may set performance objectives based upon the achievement of Company-wide, divisional, or individual goals, applicable federal or state securities laws, or any other basis determined by the Administrator in its discretion.

(d) <u>Earning of Performance Units/Shares</u>. After the applicable Performance Period has ended, the holder of Performance Units/Shares will be entitled to receive a payout of the number of Performance Units/Shares earned by the Participant over the Performance Period, to be determined as a function of the extent to which the corresponding performance objectives or other vesting provisions have been achieved. Notwithstanding the foregoing, the Administrator will have the right to reduce or eliminate (but not to increase) the amount payable at a given level of performance to take into account additional factors that the Administrator may deem relevant to the assessment of individual or corporate performance for the Performance Period. After the grant of a Performance Unit/Share, the Administrator, in its sole discretion, may reduce or waive any performance objectives or other vesting provisions for such Performance Unit/Share and may accelerate the time at which any restrictions will lapse or be removed.

(e) <u>Form and Timing of Payment of Performance Units/Shares</u>. Payment of earned Performance Units/Shares will be made as soon as practicable after the expiration of the applicable Performance Period, or as otherwise provided in the applicable Award Agreement or as required by Applicable Laws. The Administrator, in its sole discretion, may pay earned Performance Units/Shares in the form of cash, in Shares (which have an aggregate Fair Market Value equal to the value of the earned Performance Units/Shares at the close of the applicable Performance Period) or in a combination thereof.

(f) <u>Cancellation of Performance Units/Shares</u>. On the date set forth in the Award Agreement, all unearned or unvested Performance Units/Shares will be forfeited to the Company, and again will be available for grant under the Plan.

(g) <u>Section 162(m) Performance Restrictions</u>. For purposes of qualifying grants of Performance Units/Shares as "performance-based compensation" under Section 162(m) of the Code, the Administrator, in its discretion, may set restrictions based upon the achievement of Performance Goals. The Performance Goals will be set by the Administrator on or before the Determination Date. In granting Performance Units/Shares which are intended to qualify under Section 162(m) of the Code, the Administrator will follow any procedures determined by it from time to time to be necessary or appropriate to ensure qualification of the Award under Section 162(m) of the Code (e.g., in determining the Performance Goals).

11. Performance-Based Compensation Under Code Section 162(m).

(a) <u>General</u>. If the Administrator, in its discretion, decides to grant an Award intended to qualify as "performance-based compensation" under Section 162(m) of the Code, the provisions of this Section 11 will control over any contrary provision in the Plan; provided, however, that the Administrator may in its discretion grant Awards that are not intended to qualify as "performance-based compensation" under Section 162(m) of the Code to

such Participants that are based on Performance Goals or other specific criteria or goals but that do not satisfy the requirements of this Section 11.

(b) Performance Goals. The granting and/or vesting of Awards of Restricted Stock, Stock Units, Performance Shares and Performance Units and other incentives under the Plan may be made subject to the attainment of performance goals relating to one or more business criteria within the meaning of Section 162(m) of the Code and may provide for a targeted level or levels of achievement ("Performance Goals") including: attainment of research and development milestones, bookings, business divestitures and acquisitions, cash flow, cash position, contract awards or backlog, customer renewals, customer retention rates from an acquired company, business unit or division, earnings (which may include earnings before interest and taxes, earnings before taxes and net earnings), earnings per Share, expense reduction, gross margin, growth with respect to any of the foregoing measures, growth in bookings, growth in revenues, growth in stockholder value relative to the moving average of the S&P 500 Index or another index, internal rate of return, market share, net income, net profit, net sales, new product development, new product invention or innovation, number of customers, operating cash flow, operating expenses, operating income, operating margin, pre-tax profit, product defect measures, product release timelines, productivity, profit, return on assets, return on capital, return on stockholder equity, return on investment, return on sales, revenue, revenue growth, sales results, sales growth, stock price increase, time to market, total stockholder return, working capital. Any criteria used may be (A) measured in absolute terms, (B) measured in terms of growth, (C) compared to another company or companies, (D) measured against the market as a whole and/or according to applicable market indices, (E) measured against the performance of the Company as a whole or a segment of the Company and/or (F) measured on a pre-tax or post-tax basis (if applicable). Further, any Performance Goals may be used to measure the performance of the Company as a whole or a business unit or other segment of the Company, or one or more product lines or specific markets and may be measured relative to a peer group or index. The Performance Goals may differ from Participant to Participant and from Award to Award. Prior to the Determination Date, the Administrator will determine whether any significant element(s) will be included in or excluded from the calculation of any Performance Goal with respect to any Participant. In all other respects, Performance Goals will be calculated in accordance with the Company's financial statements, generally accepted accounting principles, or under a methodology established by the Administrator prior to or at the time the issuance of an Award and which is consistently applied with respect to a Performance Goal in the relevant Performance Period. The Administrator will appropriately adjust any evaluation of performance under a Performance Goal to exclude (i) any extraordinary non-recurring items as described in Accounting Principles Board Opinion No. 30 and/or in management's discussion and analysis of financial conditions and results of operations appearing in the Company's annual report to stockholders for the applicable year, or (ii) the effect of any changes in accounting principles affecting the Company's or a business units' reported results. In addition, the Administrator will adjust any performance criteria, Performance Goal or other feature of an Award that relates to or is wholly or partially based on the number of, or the value of, any stock of the Company, to reflect any stock dividend or split, repurchase, recapitalization, combination, or exchange of shares or other similar changes in such stock.

(c) <u>Procedures</u>. To the extent necessary to comply with the performance-based compensation provisions of Section 162(m) of the Code, with respect to any Award granted subject to Performance Goals and intended to qualify as "performance-based compensation" under Section 162(m) of the Code, within the first twenty-five percent (25%) of the Performance Period, but in no event more than ninety (90) days following the commencement of any Performance Period (or such other time as may be required or permitted by Section 162(m) of the Code), the Administrator will, in writing, (i) designate one or more Participants to whom an Award will be made, (ii) select the Performance Goals applicable to the Performance Period, (iii) establish the Performance Goals, and amounts of such Awards, as applicable, which may be earned for such Performance Period, and (iv) specify the relationship between Performance Goals and the amounts of such Awards, as applicable, to be earned by each Participant for such Performance Period.

(d) <u>Additional Limitations</u>. Notwithstanding any other provision of the Plan, any Award which is granted to a Participant and is intended to constitute qualified performance-based compensation under Section 162(m) of the Code will be subject to any additional limitations set forth in the Code (including any amendment to Section 162(m)) or any regulations and ruling issued thereunder that are requirements for qualification as qualified performance-

based compensation as described in Section 162(m) of the Code, and the Plan will be deemed amended to the extent necessary to conform to such requirements.

(e) <u>Determination of Amounts Earned</u>. Following the completion of each Performance Period, the Administrator will certify in writing whether the applicable Performance Goals have been achieved for such Performance Period. A Participant will be eligible to receive payment pursuant to an Award intended to qualify as "performancebased compensation" under Section 162(m) of the Code for a Performance Period only if the Performance Goals for such period are achieved. In determining the amounts earned by a Participant pursuant to an Award intended to qualified as "performance-based compensation" under Section 162(m) of the Code, the Administrator will have the right to (a) reduce or eliminate (but not to increase) the amount payable at a given level of performance to take into account additional factors that the Administrator may deem relevant to the assessment of individual or corporate performance for the Performance Period, (b) determine what actual Award, if any, will be paid in the event of a termination of employment as the result of a Participant's death or disability or upon a Change in Control or in the event of a termination of employment following a Change in Control prior to the end of the Performance Period, and (c) determine what actual Award, if any, will be paid in the event of a termination of employment other than as the result of a Participant's death or disability prior to a Change of Control and prior to the end of the Performance Period to the extent an actual Award would have otherwise been achieved had the Participant remained employed through the end of the Performance Period.

12. <u>Compliance With Code Section 409A</u>. Awards will be designed and operated in such a manner that they are either exempt from the application of, or comply with, the requirements of Code Section 409A such that the grant, payment, settlement or deferral will not be subject to the additional tax or interest applicable under Code Section 409A, except as otherwise determined in the sole discretion of the Administrator. The Plan and each Award Agreement under the Plan is intended to meet the requirements of Code Section 409A and will be construed and interpreted in accordance with such intent, except as otherwise determined in the sole discretion of the Administrator. To the extent that an Award or payment, or the settlement or deferral thereof, is subject to Code Section 409A the Award will be granted, paid, settled or deferred in a manner that will meet the requirements of Code Section 409A, such that the grant, payment, settlement or deferral will not be subject to the additional tax or interest applicable under Code Section 409A.

13. Leaves of Absence/Transfer Between Locations. Unless the Administrator provides otherwise and except as required by Applicable Laws, vesting of Awards granted hereunder will be suspended during any unpaid leave of absence. A Participant will not cease to be an Employee in the case of (i) any leave of absence approved by the Company or (ii) transfers between locations of the Company or between the Company, its Parent, or any Subsidiary. For purposes of Incentive Stock Options, no such leave may exceed three (3) months, unless reemployment upon expiration of such leave is guaranteed by statute or contract. If reemployment upon expiration of a leave of absence approved by the Company is not so guaranteed, then six (6) months following the first (1st) day of such leave, any Incentive Stock Option held by the Participant will cease to be treated as an Incentive Stock Option and will be treated for tax purposes as a Nonstatutory Stock Option.

14. Transferability of Awards .

(a) <u>Non-Transferability of Awards</u>. Unless determined otherwise by the Administrator, an Award may not be sold, pledged, assigned, hypothecated, transferred, or disposed of in any manner other than by will or by the laws of descent or distribution and may be exercised, during the lifetime of the Participant, only by the Participant.

(b) <u>Prohibition Against an Award Transfer Program</u>. Notwithstanding anything to the contrary in the Plan, in no event will the Administrator have the right to determine and implement the terms and conditions of any Award Transfer Program without stockholder approval.

15. Adjustments; Dissolution or Liquidation; Merger or Change in Control.

(a) <u>Adjustments</u>. In the event that any dividend or other distribution (whether in the form of cash, Shares, other securities, or other property), recapitalization, stock split, reverse stock split, reorganization, merger, consolidation, split-up, spin-off, combination, repurchase, or exchange of Shares or other securities of the Company, or other change in the corporate structure of the Company affecting the Shares occurs, the Administrator,

in order to prevent diminution or enlargement of the benefits or potential benefits intended to be made available under the Plan, will make such adjustments as it, in its sole discretion, deems appropriate in one or more of:

(i) the number and class of Shares that may be delivered under the Plan;

(ii) the number, class, and price of Shares covered by each outstanding Award; or

(iii) the numerical Share limits set forth in Sections 3, 6, 7, 8, 9 and 10 of the Plan.

(b) <u>Dissolution or Liquidation</u>. In the event of the proposed dissolution or liquidation of the Company, the Administrator will notify each Participant as soon as practicable prior to the effective date of such proposed transaction. To the extent it has not been previously exercised, an Award will terminate immediately prior to the consummation of such proposed action.

(c) <u>Change in Control</u>. In the event of a merger or Change in Control, each outstanding Award will be treated as the Administrator determines without a Participant's consent, including, without limitation, that:

(i) Awards will be assumed, or substantially equivalent Awards will be substituted, by the acquiring or succeeding corporation (or an affiliate thereof) with appropriate adjustments as to the number and kind of shares and prices;

(ii) upon written notice to a Participant, that the Participant's Awards will terminate upon or immediately prior to the consummation of such merger or Change in Control (for the avoidance of doubt, if Awards (or portion thereof) are not assumed or substituted by the successor corporation, a Participant's Awards will not be terminate before fully vesting as provided for in this Section 15(c));

(iii) outstanding Awards will vest and become exercisable, realizable, or payable, or restrictions applicable to an Award will lapse, in whole or in part prior to or upon consummation of such merger or Change in Control, and, to the extent the Administrator determines, terminate upon or immediately prior to the effectiveness of such merger of Change in Control;

(iv) (A) the termination of an Award in exchange for an amount of cash and/or property, if any, equal to the amount that would have been attained upon the exercise of such Award or realization of the Participant's rights as of the date of the occurrence of the transaction (and, for the avoidance of doubt, if as of the date of the occurrence of the transaction the Administrator determines in good faith that no amount would have been attained upon the exercise of such Award or realization of the Participant's rights, then such Award may be terminated by the Company without payment), or (B) the replacement of such Award with other rights or property selected by the Administrator in its sole discretion; or

(v) any combination of the foregoing.

In taking any of the actions permitted under this subsection 15(c), the Administrator will not be obligated to treat all Awards, all Awards held by a Participant, or all Awards of the same type, similarly.

In the event that the successor corporation does not assume or substitute for the Award (or portion thereof), the Participant will fully vest in and have the right to exercise all of his or her outstanding Options and Stock Appreciation Rights that are not assumed or substituted for, including Shares as to which such Awards would not otherwise be vested or exercisable, all restrictions on Restricted Stock, Stock Units, and Performance Shares/Units not assumed or substituted for will lapse, and, with respect to Awards with performance-based vesting not assumed or substituted for, all performance goals or other vesting criteria will be deemed achieved at one hundred percent (100%) of target levels and all other terms and conditions met. In addition, if an Option or Stock Appreciation Right is not assumed or substituted for in the event of a Change in Control, the Company will notify the Participant in writing or electronically that the Option or Stock Appreciation Right will be fully vested and exercisable for a period of time determined by the Administrator in its sole discretion, and the Option or Stock Appreciation Right will terminate upon the expiration of such period.

For the purposes of this subsection (c), an Award will be considered assumed if, following the Change in Control, the Award confers the right to purchase or receive, for each Share subject to the Award immediately prior to the Change in Control, the consideration (whether stock, cash, or other securities or property) or, in the case of a

Stock Appreciation Right upon the exercise of which the Administrator determines to pay cash or a Stock Unit, Performance Share or Performance Unit which the Administrator can determine to pay in cash, the fair market value of the consideration received in the merger or Change in Control by holders of Common Stock for each Share held on the effective date of the transaction (and if holders were offered a choice of consideration, the type of consideration chosen by the holders of a majority of the outstanding Shares); provided, however, that if such consideration received in the Change in Control is not solely common stock of the successor corporation or its Parent, the Administrator may, with the consent of the successor corporation, provide for the consideration to be received upon the exercise of an Option or Stock Appreciation Right or upon the payout of a Stock Unit, Performance Unit or Performance Share, for each Share subject to such Award (or in the case of an Award settled in cash, the number of implied shares determined by dividing the value of the Award by the per share consideration received by holders of Common Stock in the Change in Control), to be solely common stock of the successor corporation.

Notwithstanding anything in this Section 15(c) to the contrary, an Award that vests, is earned or paid-out upon the satisfaction of one or more Performance Goals will not be considered assumed if the Company or its successor modifies any of such Performance Goals without the Participant's consent; provided, however, a modification to such Performance Goals only to reflect the successor corporation's post-Change in Control corporate structure will not be deemed to invalidate an otherwise valid Award assumption.

Notwithstanding anything in this Section 15(c) to the contrary, if a payment under an Award Agreement is subject to Section 409A of the Code and if the change in control definition contained in the Award Agreement or other agreement related to the Award does not comply with the definition of "change in control" for purposes of a distribution under Section 409A of the Code, then any payment of an amount that is otherwise accelerated under this Section will be delayed until the earliest time that such payment would be permissible under Section 409A of the Code without triggering any penalties applicable under Section 409A of the Code.

16. Tax Withholding.

(a) <u>Withholding Requirements</u>. Prior to the delivery of any Shares or cash pursuant to an Award (or exercise thereof), the Company will have the power and the right to deduct or withhold, or require a Participant to remit to the Company, an amount sufficient to satisfy federal, state, local, foreign or other taxes (including the Participant's FICA obligation) required to be withheld with respect to such Award (or exercise thereof).

(b) <u>Withholding Arrangements</u>. The Administrator, in its sole discretion and pursuant to such procedures as it may specify from time to time, may permit a Participant to satisfy such tax withholding obligation, in whole or in part by (without limitation) (a) paying cash, (b) electing to have the Company withhold otherwise deliverable cash or Shares having a Fair Market Value equal to the amount required to be withheld, (c) delivering to the Company already-owned Shares having a Fair Market Value equal to the amount required to be withheld, provided the delivery of such Shares will not result in any adverse accounting consequences as the Administrator determines in its sole discretion, (d) selling a sufficient number of Shares otherwise deliverable to the Participant through such means as the Administrator may determine in its sole discretion (whether through a broker or otherwise) equal to the amount required to be withheld, or (e) retaining from salary or other amounts payable to the Participant cash having a sufficient value to satisfy the amount required to be withheld. The amount of the withholding requirement will be deemed to include any amount which the Administrator agrees may be withheld at the time the election is made, not to exceed the amount determined by using the maximum federal, state or local marginal income tax rates applicable to the Participant with respect to the Award on the date that the amount of tax to be withheld is to be determined. The Fair Market Value of the Shares to be withheld or delivered will be determined as of the date that the taxes are required to be withheld.

17. <u>No Effect on Employment or Service</u>. Neither the Plan nor any Award will confer upon a Participant any right with respect to continuing the Participant's relationship as a Service Provider with the Company, nor will they interfere in any way with the Participant's right or the Company's right to terminate such relationship at any time, with or without cause, to the extent permitted by Applicable Laws.

18. Date of Grant. The date of grant of an Award will be, for all purposes, the date on which the Administrator makes the determination granting such Award, or such other later date as is determined by the Administrator. Notice of the determination will be provided to each Participant within a reasonable time after the date of such grant.

19. <u>Term of Plan</u>. The Plan is effective as of its approval by the stockholders of the Company at the Company's 2010 annual meeting of stockholders as described in Section 23 of the Plan. It will continue in effect for a term of ten (10) years from the Plan's initial effectiveness, unless terminated earlier under Section 20 of the Plan.

20. Amendment and Termination of the Plan .

- (a) Amendment and Termination. The Administrator may at any time amend, alter, suspend or terminate the Plan.
- (b) Stockholder Approval. The Company will obtain stockholder approval of any Plan amendment to the extent necessary and desirable to comply with Applicable Laws.

(c) <u>Effect of Amendment or Termination</u>. No amendment, alteration, suspension or termination of the Plan will impair the rights of any Participant, unless mutually agreed otherwise between the Participant and the Administrator, which agreement must be in writing and signed by the Participant and the Company. Termination of the Plan will not affect the Administrator's ability to exercise the powers granted to it hereunder with respect to Awards granted under the Plan prior to the date of such termination.

21. Conditions upon Issuance of Shares .

(a) Legal Compliance. Shares will not be issued pursuant to the exercise of an Award unless the exercise of such Award and the issuance and delivery of such Shares will comply with Applicable Laws and will be further subject to the approval of counsel for the Company with respect to such compliance.

(b) <u>Investment Representations</u>. As a condition to the exercise of an Award, the Company may require the person exercising such Award to represent and warrant at the time of any such exercise that the Shares are being purchased only for investment and without any present intention to sell or distribute such Shares if, in the opinion of counsel for the Company, such a representation is required.

22. <u>Inability to Obtain Authority</u>. The inability of the Company to obtain authority from any regulatory body having jurisdiction, which authority is deemed by the Company's counsel to be necessary to the lawful issuance and sale of any Shares hereunder, will relieve the Company of any liability in respect of the failure to issue or sell such Shares as to which such requisite authority will not have been obtained.

23. <u>Stockholder Approval</u>. The Plan will be subject to approval by the stockholders of the Company at the Company's 2010 annual meeting of stockholders. Such stockholder approval will be obtained in the manner and to the degree required under Applicable Laws.

FOUNDSTONE, INC.

2000 STOCK PLAN

(Amended October 23, 2007)

1. <u>Purposes of the Plan</u>. The purposes of this 2000 Stock Plan are to attract and retain the best available personnel for positions of substantial responsibility, to provide additional incentive to Employees, Directors and Consultants and to promote the success of the Company's business. Options granted under the Plan may be Incentive Stock Options or Nonstatutory Stock Options, as determined by the Administrator at the time of grant. Stock Purchase Rights may also be granted under the Plan.

- 2. Definitions. As used herein, the following definitions shall apply:
 - (a) "Administrator" means the Board or any of its Committees as shall be administering the Plan in accordance with Section 4 hereof.

(b) "<u>Applicable Laws</u>" means the requirements relating to the administration of stock option plans under U.S. state corporate laws, U.S. federal and state securities laws, the Code, any stock exchange or quotation system on which the Common Stock is listed or quoted and the applicable laws of any other country or jurisdiction where Options or Stock Purchase Rights are granted under the Plan.

- (c) "Board" means the Board of Directors of the Company.
- (d) "Code" means the Internal Revenue Code of 1986, as amended.
- (e) "Committee" means a committee of Directors appointed by the Board in accordance with Section 4 hereof.
- (f) "Common Stock" means the Common Stock of the Company.
- (g) "Company" means Foundstone, Inc., a Delaware corporation.
- (h) "Consultant" means any person who is engaged by the Company or any Parent or Subsidiary to render consulting or advisory services to such entity.
- (i) "Director" means a member of the Board of Directors of the Company.
- (j) "Disability" means total and permanent disability as defined in Section 22(e)(3) of the Code.

(k) "Employee" means any person, including Officers and Directors, employed by the Company or any Parent or Subsidiary of the Company. A Service Provider shall not cease to be an Employee in the case of (i) any leave of absence approved by the Company or (ii) transfers between locations of the Company or between the Company, its Parent, any Subsidiary, or any successor. For purposes of Incentive Stock Options, no such leave may exceed ninety days, unless reemployment upon expiration of such leave is guaranteed by statute or contract. If reemployment upon expiration of a leave of absence approved by the Company is not so guaranteed, on the 181st day of such leave any Incentive Stock Option held by the Optionee shall cease to be treated as an Incentive Stock Option and shall be treated for tax purposes as a Nonstatutory Stock Option. Neither service as a Director nor payment of a director's fee by the Company shall be sufficient to constitute "employment" by the Company.

- (l) "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- (m) "Fair Market Value" means, as of any date, the value of Common Stock determined as follows:

(i) If the Common Stock is listed on any established stock exchange or a national market system, including without limitation the Nasdaq National Market or The Nasdaq SmallCap Market of The Nasdaq Stock Market, its Fair Market Value shall be the closing sales price for such stock (or the closing bid, if no sales were reported) as quoted on such exchange or system for the last market trading day prior to the time of determination, as reported in *The Wall Street Journal* or such other source as the Administrator deems reliable;

(ii) If the Common Stock is regularly quoted by a recognized securities dealer but selling prices are not reported, its Fair Market Value shall be the mean between the high bid and low asked prices for the Common Stock on the last market trading day prior to the day of determination; or

. . . .

(iii) In the absence of an established market for the Common Stock, the Fair Market Value thereof shall be determined in good faith by the

Administrator.

- (n) "Incentive Stock Option" means an Option intended to qualify as an incentive stock option within the meaning of Section 422 of the Code.
- (o) "Nonstatutory Stock Option" means an Option not intended to qualify as an Incentive Stock Option.

(p) "Officer" means a person who is an officer of the Company within the meaning of Section 16 of the Exchange Act and the rules and regulations promulgated thereunder.

(q) "Option" means a stock option granted pursuant to the Plan.

(r) "Option Agreement" means a written or electronic agreement between the Company and an Optionee evidencing the terms and conditions of an individual Option grant. The Option Agreement is subject to the terms and conditions of the Plan.

- (s) "Option Exchange Program" means a program whereby outstanding Options are exchanged for Options with a lower exercise price.
- (t) "Optioned Stock" means the Common Stock subject to an Option or a Stock Purchase Right.

- (u) "Optionee" means the holder of an outstanding Option or Stock Purchase Right granted under the Plan.
- (v) "Parent" means a "parent corporation," whether now or hereafter existing, as defined in Section 424(e) of the Code.
- (w) "Plan" means this 2000 Stock Plan, as amended.
- (x) "Restricted Stock" means shares of Common Stock acquired pursuant to a grant of a Stock Purchase Right under Section 11 below.
- (y) "Service Provider" means an Employee, Director or Consultant.
- (z) "Share" means a share of the Common Stock, as adjusted in accordance with Section 12 below.
- (aa) "Stock Purchase Right" means a right to purchase Common Stock pursuant to Section 11 below.
- (bb) "Subsidiary" means a "subsidiary corporation," whether now or hereafter existing, as defined in Section 424(f) of the Code.

3. <u>Stock Subject to the Plan</u>. Subject to the provisions of Section 12 of the Plan, the maximum aggregate number of Shares which may be subject to option and sold under the Plan is 1,250,000 Shares. The Shares may be authorized but unissued or reacquired Common Stock; provided, however, that Shares acquired by the Company with the proceeds from the sale of Shares pursuant to the exercise of Options may not be subject to option and sold under the Plan.

If an Option or Stock Purchase Right expires or becomes unexercisable without having been exercised in full or is surrendered pursuant to an Option Exchange Program, the unpurchased Shares which were subject thereto shall become available for future grant or sale under the Plan (unless the Plan has terminated). However, Shares that have actually been issued under the Plan, upon exercise of either an Option or Stock Purchase Right, shall not be returned to the Plan and shall not become available for future distribution under the Plan, including, without limitation, any and all Shares of Restricted Stock that are repurchased by the Company at their original purchase price.

4. Administration of the Plan.

(a) <u>Administrator</u>. The Plan shall be administered by the Board or a Committee appointed by the Board, which Committee shall be constituted to comply with Applicable Laws.

(b) <u>Powers of the Administrator</u>. Subject to the provisions of the Plan and, in the case of a Committee, the specific duties delegated by the Board to such Committee, and subject to the approval of any relevant authorities, the Administrator shall have the authority in its discretion:

- (i) to determine the Fair Market Value;
- (ii) to select the Service Providers to whom Options and Stock Purchase Rights may from time to time be granted hereunder;
- (iii) to determine the number of Shares to be covered by each such award granted hereunder;
- (iv) to approve forms of agreement for use under the Plan;

(v) to determine the terms and conditions, of any Option or Stock Purchase Right granted hereunder. Such terms and conditions include, but are not limited to, the exercise price, the time or times when Options or Stock Purchase Rights may be exercised (which may be based on performance criteria), any vesting acceleration or waiver of forfeiture restrictions, and any restriction or limitation regarding any Option or Stock Purchase Right or the Common Stock relating thereto, based in each case on such factors as the Administrator, in its sole discretion, shall determine;

(vi) to determine whether and under what circumstances an Option may be settled in cash under subsection 9(e) instead of Common Stock;

(vii) to reduce the exercise price of any Option to the then current Fair Market Value if the Fair Market Value of the Common Stock covered by such Option has declined since the date the Option was granted;

(viii) to initiate an Option Exchange Program;

(ix) to prescribe, amend and rescind rules and regulations relating to the Plan, including rules and regulations relating to sub-plans established for the purpose of qualifying for preferred tax treatment under foreign tax laws;

(x) to allow Optionees to satisfy withholding tax obligations by electing to have the Company withhold from the Shares to be issued upon exercise of an Option or Stock Purchase Right that number of Shares having a Fair Market Value equal to the amount required to be withheld. The Fair Market Value of the Shares to be withheld shall be determined on the date that the amount of tax to be withheld is to be determined. All elections by Optionees to have Shares withheld for this purpose shall be made in such form and under such conditions as the Administrator may deem necessary or advisable; and

- (xi) to construe and interpret the terms of the Plan and awards granted pursuant to the Plan.
- (c) Effect of Administrator's Decision. All decisions, determinations and interpretations of the Administrator shall be final and binding on all Optionees.
- 5. Eligibility.
 - (a) Nonstatutory Stock Options and Stock Purchase Rights may be granted to Service Providers. Incentive Stock Options may be granted only to Employees.

(b) Each Option shall be designated in the Option Agreement as either an Incentive Stock Option or a Nonstatutory Stock Option. However, notwithstanding such designation, to the extent that the aggregate Fair Market Value of the Shares with respect to which Incentive Stock Options are exercisable for the first time by the Optionee during any calendar year (under all plans of the Company and any Parent or Subsidiary) exceeds \$100,000, such Options shall be treated as Nonstatutory Stock Options. For purposes of this Section 5(b), Incentive Stock Options shall be taken into account in the order in which they were granted. The Fair Market Value of the Shares shall be determined as of the time the Option with respect to such Shares is granted.

(c) Neither the Plan nor any Option or Stock Purchase Right shall confer upon any Optionee any right with respect to continuing the Optionee's relationship as a Service Provider with the Company, nor shall it interfere in any way with his or her right or the Company's right to terminate such relationship at any time, with or without cause.

6. <u>Term of Plan</u>. The Plan shall become effective upon its adoption by the Board. It shall continue in effect for a term of ten (10) years unless sooner terminated under Section 14 of the Plan.

7. <u>Term of Option</u>. The term of each Option shall be stated in the Option Agreement; provided, however, that the term shall be no more than ten (10) years from the date of grant thereof. In the case of an Incentive Stock Option granted to an Optionee who, at the time the Option is granted, owns stock representing more than ten percent (10%) of the voting power of all classes of stock of the Company or any Parent or Subsidiary, the term of the Option shall be five (5) years from the date of grant or such shorter term as may be provided in the Option Agreement.

8. Option Exercise Price and Consideration.

(B)

(a) The per share exercise price for the Shares to be issued upon exercise of an Option shall be such price as is determined by the Administrator, but shall be subject to the following:

(i) In the case of an Incentive Stock Option

(A) granted to an Employee who, at the time of grant of such Option, owns stock representing more than ten percent (10%) of the voting power of all classes of stock of the Company or any Parent or Subsidiary, the exercise price shall be no less than 110% of the Fair Market Value per Share on the date of grant.

grant.

(B) granted to any other Employee, the per Share exercise price shall be no less than 100% of the Fair Market Value per Share on the date of

granted to any other Service Provider, the per Share exercise price shall be no less than 85% of the Fair Market Value per Share on the

(ii) In the case of a Nonstatutory Stock Option

(A) granted to a Service Provider who, at the time of grant of such Option, owns stock representing more than ten percent (10%) of the voting power of all classes of stock of the Company or any Parent or Subsidiary, the exercise price shall be no less than 110% of the Fair Market Value per Share on the date of grant.

date of grant.

(iii) Notwithstanding the foregoing, Options may be granted with a per Share exercise price other than as required above pursuant to a merger or

other corporate transaction.

(b) The consideration to be paid for the Shares to be issued upon exercise of an Option, including the method of payment, shall be determined by the Administrator (and, in the case of an Incentive Stock Option, shall be determined at the time of grant). Such consideration may consist of (1) cash, (2) check, (3) promissory note, (4) other Shares which (x) in the case of Shares acquired upon exercise of an Option, have been owned by the Optionee for more than six months on the date of surrender, and (y) have a Fair Market Value on the date of surrender equal to the aggregate exercise price of the Shares as to which such Option shall be exercised, (5) consideration received by the Company under a cashless exercise program implemented by the Company in connection with the Plan, or (6) any combination of the foregoing methods of payment. In making its determination as to the type of consideration to accept, the Administrator shall consider if acceptance of such consideration may be reasonably expected to benefit the Company.

9. Exercise of Option.

(a) <u>Procedure for Exercise; Rights as a Shareholder</u>. Any Option granted hereunder shall be exercisable according to the terms hereof at such times and under such conditions as determined by the Administrator and set forth in the Option Agreement. Except in the case of Options granted to Officers, Directors and Consultants, Options shall become exercisable at a rate of no less than 20% per year over five (5) years from the date the Options are granted. Unless the Administrator provides otherwise, vesting of Options granted hereunder to Officers and Directors shall be tolled during any unpaid leave of absence. An Option may not be exercised for a fraction of a Share.

An Option shall be deemed exercised when the Company receives: (i) written or electronic notice of exercise (in accordance with the Option Agreement) from the person entitled to exercise the Option, and (ii) full payment for the Shares with respect to which the Option is exercised. Full payment may consist of any consideration and method of payment authorized by the Administrator and permitted by the Option Agreement and the Plan. Shares issued upon exercise of an Option shall be issued in the name of the Optionee or, if requested by the Optionee, in the name of the Optionee and his or her spouse. Until the Shares are issued (as evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company), no right to vote or receive dividends or any other rights as a shareholder shall exist with respect to the Shares, notwithstanding the exercise of the Option. The Company shall issue (or cause to be issued) such Shares promptly after the Option is exercised. No adjustment will be made for a dividend or other right for which the record date is prior to the date the Shares are issued, except as provided in Section 12 of the Plan.

Exercise of an Option in any manner shall result in a decrease in the number of Shares thereafter available, both for purposes of the Plan and for sale under the Option, by the number of Shares as to which the Option is exercised.

(b) <u>Termination of Relationship as a Service Provider</u>. If an Optionee ceases to be a Service Provider, such Optionee may exercise his or her Option within such period of time as is specified in the Option Agreement (of at least thirty (30) days) to the extent that the Option is vested on the date of termination (but in no event later than the expiration of the term of the Option as set forth in the Option Agreement). In the absence of a specified time in the Option Agreement, the Option shall remain exercisable for three (3) months following the Optionee's termination. If, on the date of termination, the Optionee is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option shall revert to the Plan. If, after termination, the Optionee does not exercise his or her Option within the time specified by the Administrator, the Option shall terminate, and the Shares covered by such Option shall revert to the Plan.

(c) <u>Disability of Optionee</u>. If an Optionee ceases to be a Service Provider as a result of the Optionee's Disability, the Optionee may exercise his or her Option within such period of time as is specified in the Option Agreement (of at least six (6) months) to the extent the Option is vested on the date of termination (but in no event later than the expiration of the term of such Option as set forth in the Option Agreement). In the absence of a specified time in the Option Agreement, the Option shall remain exercisable for twelve (12) months following the Optionee's termination. If, on the date of termination, the Optionee is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option shall revert to the Plan. If, after termination, the Optionee does not exercise his or her Option within the time specified herein, the Option shall revert to the Plan.

(d) <u>Death of Optionee</u>. If an Optionee dies while a Service Provider, the Option may be exercised within such period of time as is specified in the Option Agreement (of at least six (6) months) to the extent that the Option is vested on the date of death (but in no event later than the expiration of the term of such Option as set forth in the Option Agreement) by the Optionee's estate or by a person who acquires the right to exercise the Option by bequest or inheritance. In the absence of a specified time in the Option Agreement, the Option shall remain exercisable for twelve (12) months following the Optionee's termination. If, at the time of death, the Optionee is not vested as to the entire Option, the Shares covered by the unvested portion of the Option shall immediately revert to the Plan. If the Option is not so exercised within the time specified herein, the Option shall terminate, and the Shares covered by such Option shall revert to the Plan. (e) <u>Buyout Provisions</u>. The Administrator may at any time offer to buy out for a payment in cash or Shares, an Option previously granted, based on such terms and conditions as the Administrator shall establish and communicate to the Optionee at the time that such offer is made.

10. <u>Non-Transferability of Options and Stock Purchase Rights</u>. The Options and Stock Purchase Rights may not be sold, pledged, assigned, hypothecated, transferred, or disposed of in any manner other than by will or by the laws of descent or distribution and may be exercised, during the lifetime of the Optionee, only by the Optionee.

11. Stock Purchase Rights.

(a) <u>Rights to Purchase</u>. Stock Purchase Rights may be issued either alone, in addition to, or in tandem with other awards granted under the Plan and/or cash awards made outside of the Plan. After the Administrator determines that it will offer Stock Purchase Rights under the Plan, it shall advise the offeree in writing or electronically of the terms, conditions and restrictions related to the offer, including the number of Shares that such person shall be entitled to purchase, the price to be paid, and the time within which such person must accept such offer. The terms of the offer shall comply in all respects with Section 260.140.42 of Title 10 of the California Code of Regulations. The offer shall be accepted by execution of a Restricted Stock purchase agreement in the form determined by the Administrator.

(b) <u>Repurchase Option</u>. Unless the Administrator determines otherwise, the Restricted Stock purchase agreement shall grant the Company a repurchase option exercisable upon the voluntary or involuntary termination of the purchaser's service with the Company for any reason (including death or disability). The purchase price for Shares repurchased pursuant to the Restricted Stock purchase agreement shall be the original price paid by the purchaser and may be paid by cancellation of any indebtedness of the purchaser to the Company. The repurchase option shall lapse at such rate as the Administrator may determine. Except with respect to Shares purchased by Officers, Directors and Consultants, the repurchase option shall in no case lapse at a rate of less than 20% per year over five (5) years from the date of purchase.

(c) <u>Other Provisions</u>. The Restricted Stock purchase agreement shall contain such other terms, provisions and conditions not inconsistent with the Plan as may be determined by the Administrator in its sole discretion.

(d) <u>Rights as a Shareholder</u>. Once the Stock Purchase Right is exercised, the purchaser shall have rights equivalent to those of a shareholder and shall be a shareholder when his or her purchase is entered upon the records of the duly authorized transfer agent of the Company. No adjustment shall be made for a dividend or other right for which the record date is prior to the date the Stock Purchase Right is exercised, except as provided in Section 12 of the Plan.

12. Adjustments Upon Changes in Capitalization, Merger or Asset Sale.

(a) <u>Changes in Capitalization</u>. Subject to any required action by the shareholders of the Company, the number of shares of Common Stock covered by each outstanding Option or Stock Purchase Right, and the number of shares of Common Stock which have been authorized for issuance under the Plan but as to which no Options or Stock Purchase Rights have yet been granted or which have been returned to the Plan upon cancellation or expiration of an Option or Stock Purchase Right, as well as the price per share of Common Stock covered by each such outstanding Option or Stock Purchase Right, shall be proportionately adjusted for any increase or decrease in the number of issued shares of Common Stock resulting from a stock split, reverse stock split, stock dividend, combination or reclassification of the Common Stock, or any other increase or decrease in the number of issued shares of Common Stock effected without receipt of consideration by the Company. The conversion of any convertible securities of the Company shall not be deemed to have been "effected without receipt of consideration." Such adjustment shall be made by the Board, whose determination in that respect shall be final, binding and conclusive. Except as expressly provided herein, no issuance by the Company of shares of stock of any class, or securities convertible into shares of stock of any class, shall affect, and no adjustment by reason thereof shall be made with respect to, the number or price of shares of Common Stock subject to an Option or Stock Purchase Right.

(b) <u>Dissolution or Liquidation</u>. In the event of the proposed dissolution or liquidation of the Company, the Administrator shall notify each Optionee as soon as practicable prior to the effective date of such proposed transaction. The Administrator in its discretion may provide for an Optionee to have the right to exercise his or her Option or Stock Purchase Right until fifteen (15) days prior to such transaction as to all of the Optioned Stock covered thereby, including Shares as to which the Option or Stock Purchase Right would not otherwise be exercisable. In addition, the Administrator may provide that any Company repurchase option applicable to any Shares purchased upon exercise of an Option or Stock Purchase Right shall lapse as to all such Shares, provided the proposed dissolution or liquidation takes place at the time and in the manner contemplated. To the extent it has not been previously exercised, an Option or Stock Purchase Right will terminate immediately prior to the consummation of such proposed action.

(c) Merger or Asset Sale. In the event of a merger of the Company with or into another corporation, or the sale of substantially all of the assets of the Company, each outstanding Option and Stock Purchase Right shall be assumed or an equivalent option or right substituted by the successor corporation or a Parent or Subsidiary of the successor corporation. In the event that the successor corporation refuses to assume or substitute for the Option or Stock Purchase Right, the Optionee shall fully vest in and have the right to exercise the Option or Stock Purchase Right as to all of the Optioned Stock, including Shares as to which it would not otherwise be vested or exercisable. If an Option or Stock Purchase Right becomes fully vested and exercisable in lieu of assumption or substitution in the event of a merger or sale of assets, the Administrator shall notify the Optionee in writing or electronically that the Option or Stock Purchase Right shall be fully exercisable for a period of fifteen (15) days from the date of such notice, and the Option or Stock Purchase Right shall terminate upon the expiration of such period. For the purposes of this paragraph, the Option or Stock Purchase Right shall be considered assumed if, following the merger or sale of assets, the option or right confers the right to purchase or receive, for each Share of Optioned Stock subject to the Option or Stock Purchase Right immediately prior to the merger or sale of assets, the consideration (whether stock, cash, or other securities or property) received in the merger or sale of assets by holders of a majority of the outstanding Shares); provided, however, that if such consideration received in the merger or sale of assets is not solely common stock of the successor corporation or its Parent, the Administrator may, with the consent of the successor corporation, provide for the consideration to be received upon the exercise of the Option or Stock Purchase Right, for each Share of Optioned Stock subject to the Option or Stock Purchase Right

13. <u>Time of Granting Options and Stock Purchase Rights</u>. The date of grant of an Option or Stock Purchase Right shall, for all purposes, be the date on which the Administrator makes the determination granting such Option or Stock Purchase Right, or such other date as is determined by the Administrator. Notice of the determination shall be given to each Service Provider to whom an Option or Stock Purchase Right is so granted within a reasonable time after the date of such grant.

- 14. Amendment and Termination of the Plan.
 - (a) <u>Amendment and Termination</u>. The Board may at any time amend, alter, suspend or terminate the Plan.

(b) <u>Shareholder Approval</u>. The Board shall obtain shareholder approval of any Plan amendment to the extent necessary and desirable to comply with Applicable Laws.

(c) <u>Effect of Amendment or Termination</u>. No amendment, alteration, suspension or termination of the Plan shall impair the rights of any Optionee, unless mutually agreed otherwise between the Optionee and the Administrator, which agreement must be in writing and signed by the Optionee and the Company. Termination of the Plan shall not affect the Administrator's ability to exercise the powers granted to it hereunder with respect to Options granted under the Plan prior to the date of such termination.

(a) Legal Compliance. Shares shall not be issued pursuant to the exercise of an Option unless the exercise of such Option and the issuance and delivery of such Shares shall comply with Applicable Laws and shall be further subject to the approval of counsel for the Company with respect to such compliance.

(b) <u>Investment Representations</u>. As a condition to the exercise of an Option, the Administrator may require the person exercising such Option to represent and warrant at the time of any such exercise that the Shares are being purchased only for investment and without any present intention to sell or distribute such Shares if, in the opinion of counsel for the Company, such a representation is required.

16. <u>Inability to Obtain Authority</u>. The inability of the Company to obtain authority from any regulatory body having jurisdiction, which authority is deemed by the Company's counsel to be necessary to the lawful issuance and sale of any Shares hereunder, shall relieve the Company of any liability in respect of the failure to issue or sell such Shares as to which such requisite authority shall not have been obtained.

17. <u>Reservation of Shares</u>. The Company, during the term of this Plan, shall at all times reserve and keep available such number of Shares as shall be sufficient to satisfy the requirements of the Plan.

18. <u>Shareholder Approval</u>. The Plan shall be subject to approval by the shareholders of the Company within twelve (12) months after the date the Plan is adopted. Such shareholder approval shall be obtained in the degree and manner required under Applicable Laws.

19. <u>Information to Optionees and Purchasers</u>. The Company shall provide to each Optionee and to each individual who acquires Shares pursuant to the Plan, not less frequently than annually during the period such Optionee or purchaser has one or more Options or Stock Purchase Rights outstanding, and, in the case of an individual who acquires Shares pursuant to the Plan, during the period such individual owns such Shares, copies of annual financial statements. The Company shall not be required to provide such statements to key employees whose duties in connection with the Company assure their access to equivalent information.

MX LOGIC, INC.

2002 EQUITY INCENTIVE PLAN

Adopted September 6, 2002

Approved By Stockholders September 6, 2002

Termination Date: September 6, 2012

1. PURPOSES.

(a) Background. This Plan was adopted on September 6, 2002, and approved by stockholders effective September 6, 2002.

(b) Eligible Stock Award Recipients. The persons eligible to receive Stock Awards are the Employees, Directors and Consultants of the Company and its Affiliates.

(c) Available Stock Awards. The purpose of the Plan is to provide a means by which eligible recipients of Stock Awards may be given an opportunity to benefit from increases in value of the Common Stock through the granting of the following Stock Awards: (i) Incentive Stock Options, (ii) Nonstatutory Stock Options, (iii) stock bonuses and (iv) rights to acquire restricted stock.

(d) General Purpose. The Company, by means of the Plan, seeks to retain the services of the group of persons eligible to receive Stock Awards, to secure and retain the services of new members of this group and to provide incentives for such persons to exert maximum efforts for the success of the Company and its Affiliates.

2. DEFINITIONS.

(a) "*Affiliate*" means any parent corporation or subsidiary corporation of the Company, whether now or hereafter existing, as those terms are defined in Sections 424(e) and (f), respectively, of the Code.

- (b) "Board" means the Board of Directors of the Company.
- (c) "Code" means the Internal Revenue Code of 1986, as amended.
- (d) "Committee" means a committee of one or more members of the Board appointed by the Board in accordance with subsection 3(c).
- (e) "Common Stock" means the common stock of the Company.
- (f) "Company" means MX Logic, Inc., a Delaware corporation.

(g) "Consultant" means any person, including an advisor, (i) engaged by the Company or an Affiliate to render consulting or advisory services and who is compensated for such services or (ii) who is a member of the Board of Directors of an Affiliate.

(h) "Continuous Service" means that the Participant's service with the Company or an Affiliate, whether as an Employee, Director or Consultant, is not interrupted or terminated. The Participant's Continuous Service shall not be deemed to have terminated merely because of a change in the capacity in which the Participant renders service to the Company or an Affiliate as an Employee, Consultant or Director or a change in the entity for which the Participant renders such service, provided that there is no interruption or termination of the Participant's Continuous Service. For example, a change in status from an Employee of the Company to a Consultant of an Affiliate or a Director will not constitute an interruption of Continuous Service. The Board or the chief executive officer of the Company, in that party's sole discretion, may determine whether Continuous Service shall be considered interrupted in the case of any leave of absence approved by that party, including sick leave, military leave or any other personal leave.

(i) "Director" means a member of the Board of Directors of the Company.

(j) "Disability" the inability of a person, in the opinion of a qualified physician acceptable to the Company, to perform the major duties of that person's position with the Company or an Affiliate of the Company because of the sickness or injury of the person.

(k) "Employee" means any person employed by the Company or an Affiliate. Mere service as a Director or payment of a director's fee by the Company or an Affiliate shall not be sufficient to constitute "employment" by the Company or an Affiliate.

(1) "Fair Market Value" means, as of any date, the value of the Common Stock determined as follows:

(i) If the Common Stock is listed on any established stock exchange, or traded on the Nasdaq National Market or the Nasdaq SmallCap Market, the Fair Market Value of a share of Common Stock shall be the closing sales price for such stock (or the closing bid, if no sales were reported) as quoted on such exchange or market (or the exchange or market with the greatest volume of trading in Common Stock) on (i) the day of determination, as reported in the Wall Street Journal or such other source as the Board deems reliable, or (ii) if the day of determination is not a trading day, then the trading day prior to the day of determination, as reported in the Wall Street Journal or such other source as the Board deems reliable.

(ii) In the absence of such markets for the Common Stock, the Fair Market Value shall be determined in good faith by the Board.

(m) "Incentive Stock Option" means an Option intended to qualify as an incentive stock option within the meaning of Section 422 of the Code and the regulations promulgated thereunder.

(n) "Non-Employee Director" means a Director who either (i) is not a current Employee or Officer of the Company or an Affiliate, does not receive compensation (directly or indirectly) from the Company or an Affiliate for services rendered as a consultant or in any capacity other than as a Director (except for an amount as to which disclosure would not be required under Item 404(a) of Regulation S-K promulgated pursuant to the Securities Act ("Regulation S-K")), does not possess an interest in any other transaction as to which disclosure would be required under Item 404(a) of Regulation S-K and is not engaged in a business relationship as to which disclosure would be required under Item 404(b) of Regulation S-K; or (ii) is otherwise considered a "non-employee director" for purposes of Rule 16b-3.

(o) "Nonstatutory Stock Option" means an Option not intended to qualify as an Incentive Stock Option.

- (p) "Officer" means any person designated by the Company as an officer.
- (q) "Option" means an Incentive Stock Option or a Nonstatutory Stock Option granted pursuant to the Plan.

(r) "Option Agreement" means a written agreement between the Company and an Optionholder evidencing the terms and conditions of an individual Option grant. Each Option Agreement shall be subject to the terms and conditions of the Plan.

(s) "Optionholder" means a person to whom an Option is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Option.

- (t) "Participant" means a person to whom a Stock Award is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Stock Award.
- (u) "Plan" means this MX Logic, Inc. 2002 Equity Incentive Plan.
- (v) "Securities Act" means the Securities Act of 1933, as amended.
- (w) "Stock Award" means any right granted under the Plan, including an Option, a stock bonus and a right to acquire restricted stock.

(x) "Stock Award Agreement" means a written agreement between the Company and a holder of a Stock Award evidencing the terms and conditions of an individual Stock Award grant. Each Stock Award Agreement shall be subject to the terms and conditions of the Plan.

(y) "Ten Percent Stockholder" means a person who owns (or is deemed to own pursuant to Section 424(d) of the Code) stock possessing more than ten percent (10%) of the total combined voting power of all classes of stock of the Company or of any of its Affiliates.

3. Administration.

(a) Administration by Board. The Board shall administer the Plan unless and until the Board delegates administration to a Committee, as provided in subsection 3(c).

(b) Powers of Board. The Board shall have the power, subject to, and within the limitations of, the express provisions of the Plan:

(i) To determine from time to time which of the persons eligible under the Plan shall be granted Stock Awards; when and how each Stock Award shall be granted; what type or combination of types of Stock Award shall be granted; the provisions of each Stock Award granted (which need not be identical), including the time or times when a person shall be permitted to receive Common Stock pursuant to a Stock Award; and the number of shares of Common Stock with respect to which a Stock Award shall be granted to each such person.

(ii) To construe and interpret the Plan and Stock Awards granted under it, and to establish, amend and revoke rules and regulations for its administration. The Board, in the exercise of this power, may correct any defect, omission or inconsistency in the Plan or in any Stock Award Agreement, in a manner and to the extent it shall deem necessary or expedient to make the Plan fully effective.

(iii) To amend the Plan or a Stock Award as provided in Section 12.

(iv) Generally, to exercise such powers and to perform such acts as the Board deems necessary or expedient to promote the best interests of the Company which are not in conflict with the provisions of the Plan.

(c) Delegation to Committee. The Board may delegate administration of the Plan to a Committee or Committee of one (1) or more members of the Board, and the term "Committee" shall apply to any person or persons to whom such authority has been delegated. If administration is delegated to a Committee, the Committee shall have, in connection with the administration of the Plan, the powers theretofore possessed by the Board, including the power to delegate to a subcommittee any of the administrative powers the Committee is authorized to exercise (and references in this Plan to the Board shall thereafter be to the Committee or subcommittee), subject, however, to such resolutions, not inconsistent with the provisions of the Plan, as may be adopted from time to time by the Board. The Board may abolish the Committee at any time and revest in the Board the administration of the Plan.

(d) Effect of Board's Decision. All determinations, interpretations and constructions made by the Board in good faith shall not be subject to review by any person and shall be final, binding and conclusive on all persons.

4. SHARES SUBJECT TO THE PLAN .

(a) Share Reserve. Subject to the provisions of Section 11 relating to adjustments upon changes in Common Stock, the Common Stock that may be issued pursuant to Stock Awards shall not exceed in the aggregate 100,000 shares of Common Stock.

(b) Reversion of Shares to the Share Reserve . If any Stock Award shall for any reason expire or otherwise terminate, in whole or in part, without having been exercised in full,

the shares of Common Stock not acquired under such Stock Award shall revert to and again become available for issuance under the Plan.

(c) Source of Shares. The shares of Common Stock subject to the Plan may be unissued shares or reacquired shares, bought on the market or otherwise.

5. ELIGIBILITY.

(a) Eligibility for Specific Stock Awards . Incentive Stock Options may be granted only to Employees. Stock Awards other than Incentive Stock Options may be granted to Employees, Directors and Consultants.

(b) Ten Percent Stockholders. A Ten Percent Stockholder shall not be granted an Incentive Stock Option unless the exercise price of such Option is at least one hundred ten percent (110%) of the Fair Market Value of the Common Stock at the date of grant and the Option is not exercisable after the expiration of five (5) years from the date of grant.

(c) Consultants.

(i) A Consultant shall not be eligible for the grant of a Stock Award if, at the time of grant, either the offer or the sale of the Company's securities to such Consultant is not exempt under Rule 701 of the Securities Act ("Rule 701") because of the nature of the services that the Consultant is providing to the Company, or because the Consultant is not a natural person, or as otherwise provided by Rule 701, unless the Company determines that such grant need not comply with the requirements of Rule 701 and will satisfy another exemption under the Securities Act as well as comply with the securities laws of all other relevant jurisdictions.

(ii) Rule 701 generally is available to consultants and advisors only if (i) they are natural persons; (ii) they provide bona fide services to the issuer, its parents, its majorityowned subsidiaries or majority-owned subsidiaries of the issuer's parent; and (iii) the services are not in connection with the offer or sale of securities in a capital-raising transaction, and do not directly or indirectly promote or maintain a market for the issuer's securities.

6. OPTION PROVISIONS.

Each Option shall be in such form and shall contain such terms and conditions as the Board shall deem appropriate. All Options shall be separately designated Incentive Stock Options or Nonstatutory Stock Options at the time of grant, and, if certificates are issued, a separate certificate or certificates will be issued for shares of Common Stock purchased on exercise of each type of Option. The provisions of separate Options need not be identical, but each Option shall include (through incorporation of provisions hereof by reference in the Option or otherwise) the substance of each of the following provisions:

(a) Term. Subject to the provisions of subsection 5(b) regarding Ten Percent Stockholders, no Option shall be exercisable after the expiration of ten (10) years from the date it was granted.



(b) Exercise Price of an Incentive Stock Option. Subject to the provisions of subsection 5(b) regarding Ten Percent Stockholders, the exercise price of each Incentive Stock Option shall be not less than one hundred percent (100%) of the Fair Market Value of the Common Stock subject to the Option on the date the Option is granted.

(c) Exercise Price of a Nonstatutory Stock Option. The exercise price of each Nonstatutory Stock Option shall be the price determined by the Board.

(d) Consideration. The purchase price of Common Stock acquired pursuant to an Option shall be paid, to the extent permitted by applicable statutes and regulations, either (i) in cash at the time the Option is exercised or (ii) at the discretion of the Board at the time of the grant of the Option (or subsequently in the case of a Nonstatutory Stock Option) (1) by delivery to the Company of other Common Stock, (2) according to a deferred payment or other similar arrangement with the Optionholder or (3) in any other form of legal consideration that may be acceptable to the Board. Unless otherwise specifically provided in the Option, the purchase price of Common Stock acquired pursuant to an Option that is paid by delivery to the Company of other Common Stock acquired, directly or indirectly from the Company, shall be paid only by shares of the Common Stock of the Company that have been held for more than six (6) months (or such longer or shorter period of time required to avoid a charge to earnings for financial accounting purposes). At any time that the Company is incorporated in Delaware, payment of the Common Stock's "par value," as defined in the Delaware General Corporation Law, shall not be made by deferred payment.

In the case of any deferred payment arrangement, interest shall be compounded at least annually and shall be charged at the market rate of interest necessary to avoid a charge to earnings for financial accounting purposes.

(e) Transferability of an Incentive Stock Option. An Incentive Stock Option shall not be transferable except by will or by the laws of descent and distribution and shall be exercisable during the lifetime of the Optionholder only by the Optionholder. Notwithstanding the foregoing, the Optionholder may, by delivering written notice to the Company, in a form satisfactory to the Company, designate a third party who, in the event of the death of the Optionholder, shall thereafter be entitled to exercise the Option.

(f) Transferability of a Nonstatutory Stock Option. A Nonstatutory Stock Option shall be transferable to the extent provided in the Option Agreement. If the Nonstatutory Stock Option does not provide for transferability, then the Nonstatutory Stock Option shall not be transferable except by will or by the laws of descent and distribution and shall be exercisable during the lifetime of the Optionholder only by the Optionholder. Notwithstanding the foregoing, the Optionholder may, by delivering written notice to the Company, in a form satisfactory to the Company, designate a third party who, in the event of the death of the Optionholder, shall thereafter be entitled to exercise the Option.

(g) Vesting Generally. The total number of shares of Common Stock subject to an Option may, but need not, vest and therefore become exercisable in periodic installments that may, but need not, be equal. The Option may be subject to such other terms and conditions on

the time or times when it may be exercised (which may be based on performance or other criteria) as the Board may deem appropriate. The vesting provisions of individual Options may vary. The provisions of this subsection 6(g) are subject to any Option provisions governing the minimum number of shares of Common Stock as to which an Option may be exercised.

(h) Termination of Continuous Service. In the event an Optionholder's Continuous Service terminates (other than upon the Optionholder's death or Disability), the Optionholder may exercise his or her Option (to the extent that the Optionholder was entitled to exercise such Option as of the date of termination) but only within such period of time ending on the earlier of (i) the date one (1) month following the termination of the Optionholder's Continuous Service (or such longer or shorter period specified in the Option Agreement), or (ii) the expiration of the term of the Option as set forth in the Option Agreement. If, after termination, the Optionholder does not exercise his or her Option within the time specified in the Option Agreement, the Option shall terminate.

(i) Extension of Termination Date. An Optionholder's Option Agreement may also provide that if the exercise of the Option following the termination of the Optionholder's Continuous Service (other than upon the Optionholder's death or Disability) would be prohibited at any time solely because the issuance of shares of Common Stock would violate the registration requirements under the Securities Act, then the Option shall terminate on the earlier of (i) the expiration of the Option set forth in subsection 6(a) or (ii) the expiration of a period of one (1) month after the termination of the Optionholder's Continuous Service during which the exercise of the Option would not be in violation of such registration requirements.

(j) Disability of Optionholder. In the event that an Optionholder's Continuous Service terminates as a result of the Optionholder's Disability, the Optionholder may exercise his or her Option (to the extent that the Optionholder was entitled to exercise such Option as of the date of termination), but only within such period of time ending on the earlier of (i) the date six (6) months following such termination (or such longer or shorter period specified in the Option Agreement), or (ii) the expiration of the term of the Option as set forth in the Option Agreement. If, after termination, the Optionholder does not exercise his or her Option within the time specified herein, the Option shall terminate.

(k) Death of Optionholder. In the event (i) an Optionholder's Continuous Service terminates as a result of the Optionholder's death or (ii) the Optionholder dies within the period (if any) specified in the Option Agreement after the termination of the Optionholder's Continuous Service for a reason other than death, then the Option may be exercised (to the extent the Optionholder was entitled to exercise such Option as of the date of death) by the Optionholder's estate, by a person who acquired the right to exercise the Option by bequest or inheritance or by a person designated to exercise the Option upon the Optionholder's death pursuant to subsection 6(e) or 6(f), but only within the period ending on the earlier of (1) the date six (6) months following the date of death (or such longer or shorter period specified in the Option Agreement), or (2) the expiration of the term of such Option as set forth in the Option Agreement. If, after death, the Option is not exercised within the time specified herein, the Option shall terminate.

(1) Right of Repurchase. Subject to the "Repurchase Limitation" in subsection 10(g), the Option may, but need not, include a provision whereby the Company may elect to repurchase all or any part of the vested shares of Common Stock acquired by the Optionholder pursuant to the exercise of the Option.

(m) Right of First Refusal. The Option may, but need not, include a provision whereby the Company may elect to exercise a right of first refusal following receipt of notice from the Optionholder of the intent to transfer all or any part of the shares of Common Stock received upon the exercise of the Option. Except as expressly provided in this subsection 6(n), such right of first refusal shall otherwise comply with any applicable provisions of the Bylaws of the Company.

7. PROVISIONS OF STOCK AWARDS OTHER THAN OPTIONS.

(a) Stock Bonus Awards. Each stock bonus agreement shall be in such form and shall contain such terms and conditions as the Board shall deem appropriate. The terms and conditions of stock bonus agreements may change from time to time, and the terms and conditions of separate stock bonus agreements need not be identical, but each stock bonus agreement shall include (through incorporation of provisions hereof by reference in the agreement or otherwise) the substance of each of the following provisions:

(i) Consideration. A stock bonus may be awarded in consideration for past services actually rendered to the Company or an Affiliate for its benefit.

(ii) Vesting. Subject to the "Repurchase Limitation" in subsection 10(g), shares of Common Stock awarded under the stock bonus agreement may, but need not, be subject to a share repurchase option in favor of the Company in accordance with a vesting schedule to be determined by the Board.

(iii) Transferability. Rights to acquire shares of Common Stock under the stock bonus agreement shall not be transferable except by will or by the laws of descent and distribution and shall be exercisable during the lifetime of the Participant only by the Participant

(b) Restricted Stock Awards. Each restricted stock purchase agreement shall be in such form and shall contain such terms and conditions as the Board shall deem appropriate. The terms and conditions of the restricted stock purchase agreements may change from time to time, and the terms and conditions of separate restricted stock purchase agreements need not be identical, but each restricted stock purchase agreement shall include (through incorporation of provisions hereof by reference in the agreement or otherwise) the substance of each of the following provisions:

(i) Purchase Price. The purchase price of restricted stock awards shall be at the price determined by the Board.

(ii) Consideration. The purchase price of Common Stock acquired pursuant to the restricted stock purchase agreement shall be paid either: (i) in cash at the time of 8

purchase; (ii) at the discretion of the Board, according to a deferred payment or other similar arrangement with the Participant; or (iii) in any other form of legal consideration that may be acceptable to the Board in its discretion; provided, however, that at any time that the Company is incorporated in Delaware, then payment of the Common Stock's "par value," as defined in the Delaware General Corporation Law, shall not be made by deferred payment.

(iii) Vesting. Subject to the "Repurchase Limitation" in subsection 10(g), shares of Common Stock acquired under the restricted stock purchase agreement may, but need not, be subject to a share repurchase option in favor of the Company in accordance with a vesting schedule to be determined by the Board.

(iv) Transferability. Rights to acquire shares of Common Stock under the restricted stock purchase agreement shall not be transferable except by will or by the laws of descent and distribution and shall be exercisable during the lifetime of the Participant only by the Participant.

8. COVENANTS OF THE COMPANY.

(a) Availability of Shares. During the terms of the Stock Awards, the Company shall keep available at all times the number of shares of Common Stock required to satisfy such Stock Awards.

(b) Securities Law Compliance. The Company shall seek to obtain from each regulatory commission or agency having jurisdiction over the Plan such authority as may be required to grant Stock Awards and to issue and sell shares of Common Stock upon exercise of the Stock Awards; provided, however, that this undertaking shall not require the Company to register under the Securities Act the Plan, any Stock Award or any Common Stock issued or issuable pursuant to any such Stock Award. If, after reasonable efforts, the Company is unable to obtain from any such regulatory commission or agency the authority which counsel for the Company deems necessary for the lawful issuance and sale of Common Stock under the Plan, the Company shall be relieved from any liability for failure to issue and sell Common Stock upon exercise of such Stock Awards unless and until such authority is obtained.

9. Use of Proceeds from Stock.

Proceeds from the sale of Common Stock pursuant to Stock Awards shall constitute general funds of the Company.

10. MISCELLANEOUS.

(a) Acceleration of Exercisability and Vesting. The Board shall have the power to accelerate the time at which a Stock Award may first be exercised or the time during which a Stock Award or any part thereof will vest in accordance with the Plan, notwithstanding the provisions in the Stock Award stating the time at which it may first be exercised or the time during which it will vest.

(b) Stockholder Rights. No Participant shall be deemed to be the holder of, or to have any of the rights of a holder with respect to, any shares of Common Stock subject to such Stock Award unless and until such Participant has satisfied all requirements for exercise of the Stock Award pursuant to its terms.

(c) No Employment or other Service Rights. Nothing in the Plan or any instrument executed or Stock Award granted pursuant thereto shall confer upon any Participant any right to continue to serve the Company or an Affiliate in the capacity in effect at the time the Stock Award was granted or shall affect the right of the Company or an Affiliate to terminate (i) the employment of an Employee with or without notice and with or without cause, (ii) the service of a Consultant pursuant to the terms of such Consultant's agreement with the Company or an Affiliate or (iii) the service of a Director pursuant to the Bylaws of the Company or an Affiliate, and any applicable provisions of the corporate law of the state in which the Company or the Affiliate is incorporated, as the case may be.

(d) Incentive Stock Option \$100,000 Limitation. To the extent that the aggregate Fair Market Value (determined at the time of grant) of Common Stock with respect to which Incentive Stock Options are exercisable for the first time by any Optionholder during any calendar year (under all plans of the Company and its Affiliates) exceeds one hundred thousand dollars (\$100,000), the Options or portions thereof which exceed such limit (according to the order in which they were granted) shall be treated as Nonstatutory Stock Options.

(e) Investment Assurances. The Company may require a Participant, as a condition of exercising or acquiring Common Stock under any Stock Award, (i) to give written assurances satisfactory to the Company as to the Participant's knowledge and experience in financial and business matters and/or to employ a purchaser representative reasonably satisfactory to the Company who is knowledgeable and experienced in financial and business matters and that he or she is capable of evaluating, alone or together with the purchaser representative, the merits and risks of exercising the Stock Award; and (ii) to give written assurances satisfactory to the Company stating that the Participant is acquiring Common Stock subject to the Stock Award for the Participant's own account and not with any present intention of selling or otherwise distributing the Common Stock. The foregoing requirements, and any assurances given pursuant to such requirements, shall be inoperative if (1) the issuance of the shares of Common Stock upon the exercise or acquisition of Common Stock under the Stock Award has been registered under a then currently effective registration statement under the Securities Act or (2) as to any particular requirement, a determination is made by counsel for the Company that such requirement need not be met in the circumstances under the then applicable securities laws. The Company may, upon advice of counsel to the Company, place legends on stock certificates issued under the Plan as such counsel deems necessary or appropriate in order to comply with applicable securities laws, including, but not limited to, legends restricting the transfer of the Common Stock.

(f) Withholding Obligations. The Company's obligation to deliver shares of Common Stock pursuant to any Stock Award shall be subject to satisfaction of all applicable income and employment tax withholding requirements. To the extent provided by the terms of a

Stock Award Agreement, the Participant (if such Participant is an Employee) may satisfy any federal, state or local tax withholding obligation relating to the exercise or acquisition of Common Stock under a Stock Award by tendering a cash payment (in addition to the Company's right to withhold from any compensation paid to such Participant by the Company).

(g) Repurchase Limitation. The terms of any repurchase option shall be specified in the Stock Award and may be either at Fair Market Value at the time of repurchase or at not less than the original purchase price.

(h) Share Escrow/Legends. Unvested shares may, in the Board's discretion, be held in escrow by the Company until the Participant's interest in such shares vests or may be issued directly to the Participant with restrictive legends on the certificates representing such unvested shares.

11. Adjustments upon Changes in Stock.

(a) Capitalization Adjustments . If any change is made in the Common Stock subject to the Plan, or subject to any Stock Award, without the receipt of consideration by the Company (through merger, consolidation, reorganization, recapitalization, reincorporation, stock dividend, dividend in property other than cash, stock split, liquidating dividend, combination of shares, exchange of shares, change in corporate structure or other transaction not involving the receipt of consideration by the Company), the Plan will be appropriately adjusted in the class(es) and maximum number of securities subject to the Plan pursuant to subsection 4(a), and the outstanding Stock Awards will be appropriately adjusted in the class(es) and number of securities and price per share of Common Stock subject to such outstanding Stock Awards. The Board shall make such adjustments, and its determination shall be final, binding and conclusive. (The conversion of any convertible securities of the Company shall not be treated as a transaction "without receipt of consideration" by the Company.)

(b) Dissolution or Liquidation . In the event of a dissolution or liquidation of the Company, then all outstanding Stock Awards shall terminate immediately prior to such event.

(c) Asset Sale, Merger, Consolidation or Reverse Merger. In the event of (i) a sale, lease or other disposition of all or substantially all of the assets of the Company, (ii) a merger or consolidation of the Company with or into any other corporation or entity or person, or any other corporate reorganization, in which the stockholders of the Company immediately prior to such consolidation, merger or reorganization, own less than 50% of the Company's outstanding voting power of the surviving entity (or its parent) following the consolidation, merger, or reorganization or (iii) any transaction (or series of related transactions involving a person or entity, or a group of affiliated persons or entities) in which in excess of fifty percent (50%) of the Company's outstanding voting power is transferred (individually, a "Corporate Transaction"), then any surviving corporation or acquiring corporation shall assume any Stock Awards outstanding under the Plan or shall substitute similar stock awards (including an award to acquire the same consideration paid to the stockholders in the Corporate Transaction for those outstanding under the Plan). In the event any surviving corporation or acquiring corporation refuses to assume such Stock Awards or to substitute similar stock awards for those outstanding

under the Plan, then all Stock Awards outstanding shall terminate upon the consummation of the Corporation Transaction.

12. Amendment of the Plan and Stock Awards.

(a) Amendment of Plan. The Board at any time, and from time to time, may amend the Plan. However, except as provided in Section 11 relating to adjustments upon changes in Common Stock, no amendment shall be effective unless approved by the stockholders of the Company to the extent stockholder approval is necessary to satisfy the requirements of Section 422 of the Code.

(b) Stockholder Approval. The Board may, in its sole discretion, submit any other amendment to the Plan for stockholder approval, including, but not limited to, amendments to the Plan intended to satisfy the requirements of Section 162(m) of the Code and the regulations thereunder regarding the exclusion of performance-based compensation from the limit on corporate deductibility of compensation paid to certain executive officers.

(c) Contemplated Amendments. It is expressly contemplated that the Board may amend the Plan in any respect the Board deems necessary or advisable to provide eligible Employees with the maximum benefits provided or to be provided under the provisions of the Code and the regulations promulgated thereunder relating to Incentive Stock Options and/or to bring the Plan and/or Incentive Stock Options granted under it into compliance therewith.

(d) No Impairment of Rights. Rights under any Stock Award granted before amendment of the Plan shall not be impaired by any amendment of the Plan unless (i) the Company requests the consent of the Participant and (ii) the Participant consents in writing.

(e) Amendment of Stock Awards. The Board at any time, and from time to time, may amend the terms of any one or more Stock Awards; provided, however, that the rights under any Stock Award shall not be impaired by any such amendment unless (i) the Company requests the consent of the Participant and (ii) the Participant consents in writing.

13. TERMINATION OR SUSPENSION OF THE PLAN.

(a) Plan Term. The Board may suspend or terminate the Plan at any time. Unless sooner terminated, the Plan shall terminate on the day before the tenth (10th) anniversary of the date the Plan is adopted by the Board or approved by the stockholders of the Company, whichever is earlier. No Stock Awards may be granted under the Plan while the Plan is suspended or after it is terminated.

(b) No Impairment of Rights. Suspension or termination of the Plan shall not impair rights and obligations under any Stock Award granted while the Plan is in effect except with the written consent of the Participant.

14. EFFECTIVE DATE OF PLAN.

The Plan shall become effective as determined by the Board, but no Stock Award shall be exercised (or, in the case of a stock bonus, shall be granted) unless and until the Plan has been approved by the stockholders of the Company. If such stockholder approval is not obtained within twelve (12) months after the date the Plan is adopted by the Board, then all Stock Awards previously granted under the Plan shall terminate and cease to be outstanding, and no further Stock Awards shall be granted and no shares shall be issued under the Plan.

15. CHOICE OF LAW .

The law of the State of Delaware shall govern all questions concerning the construction, validity and interpretation of this Plan, without regard to such state's conflict of laws rules.

Secure Computing Corporation 2000 STOCK PLAN (As Amended through May 10, 2007)

I. PURPOSE

Secure Computing Corporation (the "Company") 2000 Stock Plan is intended to encourage stock ownership by officers and other key employees of the Company and of its subsidiaries, to provide them with a proprietary interest or to increase their proprietary interest in the Company's success and/or to encourage them to remain in the employ of the Company or any of its subsidiaries. The 2000 Stock Plan was originally adopted by CipherTrust, Inc. which the Company acquired on August 31, 2006.

II. DEFINITIONS

Where the following words appear in the Plan, they shall have the respective meanings set forth below, unless their context clearly indicates a contrary meaning:

- A. Award An Option or Stock Award granted in accordance with the terms of the Plan.
- B. Award Agreement A Stock Award Agreement and/or Option Agreement.
- C. Board of Directors The Board of Directors of the Company.
- D. Code The Internal Revenue Code of 1986, as amended, including amendments hereafter adopted.
- E. Committee The Compensation Committee of the Board of Directors or any successor Committee appointed by the Board of Directors. In the absence of the appointment of the Committee, the Board of Directors of the Company shall exercise all of the powers of the Committee under the Plan.
- F. Company Secure Computing Corporation, a Delaware corporation, the successor corporation to CipherTrust, Inc., a Georgia corporation.
- G. Employee Employee shall mean any officer or other key employee (including an officer or other key employee who is also a director) employed on a full time basis by the Company or any present or future Parent or Subsidiary.
- H. ISO An option granted under the Plan which constitutes an incentive stock option within the meaning of Section 422 of the Code.
- I. Non-Qualified Stock Option or NQSO An option granted under the Plan which does not qualify as an ISO.
- J. Option An option granted under the Plan which may be either an ISO or a Non-Qualified Stock Option.

Stock Option Plan

- K. Option Agreement The document setting forth the terms and conditions of each Option.
- L. Optionee The holder of an Option.
- M. Parent Parent shall mean any present or future corporation as defined in Subsections 424(e) and (g) of the Code.
- N. Participant Any holder of one or more Options or Stock Awards or the Shares issuable or issued upon exercise of such Options or Stock Awards under the Plan.
- O. Plan Secure Computing Corporation 2000 Stock Plan, as the same may be amended from time to time in accordance with the terms hereof.
- P. Qualifying Performance Criteria Any one or more of the following performance criteria, either individually, alternatively or in any combination, applied to either the Company as a whole or to a business unit, a Parent or Subsidiary, or business segment, either individually, alternatively or in any combination, and measured either annually or cumulatively over a period of years, on an absolute basis or relative to a pre-established target, to previous years' results or to a designated comparison group, in each case as specified by the Committee in the Award: (i) cash flow; (ii) earnings (including gross margin, earnings before interest and taxes, earnings before taxes, and net earnings); (iii) earnings per share; (iv) growth in earnings or earnings per share; (v) stock price; (vi) return on equity or average stockholders' equity; (vii) total stockholder return; (viii) return on capital; (ix) return on assets or net assets; (x) return on investment; (xi) revenue; (xii) income or net income; (xiii) operating income or net operating income; (xiv) operating profit or net operating profit; (xv) operating margin; (xvi) return on operating revenue; (xvii) market share; (xviii) contract awards or backlog; (xix) overhead or other expense reductior; (xx) growth in stockholder value relative to the moving average of the S&P 500 Index or a peer group index; (xxi) credit rating; (xxii) strategic plan development and implementation (including individual performance objectives that relate to achievement of the Committee. The Committee may appropriately adjust any evaluation of performance under a Qualifying Performance Criteria to exclude any of the following events that occurs during a performance period: (A) asset write-downs; (B) litigation or claim judgments or settlements; (C) the effect of changes in tax law, accounting principles or other such laws or provisions affecting reported results; (D) accurals for reorganization and restructuring programs; and (E) any gains or losses classified as extraordinary or as di
- Q. Shares The shares of common stock of the Company, \$.001 par value, subject to adjustment and substitution as provided in Paragraph V of the Plan.
- R. Stock Appreciation Right A right to receive cash and/or Shares based on a change in the fair market value of a specific number of Shares granted under Paragraph IX.
- S. Stock Award A Stock Grant, a Stock Unit or a Stock Appreciation Right granted under Paragraphs VIII or IX below or other similar awards granted under the Plan (including phantom stock rights).
- T. Stock Award Agreement A written agreement, the form(s) of which shall be approved from time to time by the Committee, between the Company and a holder of a Stock Award evidencing the terms and conditions of an individual Stock Award grant. Each Stock Award Agreement shall be subject to the terms and conditions of the Plan.
- U. Stock Grant The award of a certain number of Shares granted under Paragraph VIII below.
- V. Stock Unit A bookkeeping entry representing an amount equivalent to the fair market value of one Share, payable in cash, property or Shares. Stock Units represent an unfunded and unsecured obligation of the Company, except as otherwise explicitly provided for by the Committee.
- W. Subsidiary Any present or future subsidiary of the Company as defined in Subsections 424(f) and (g) of the Code.

III. ADMINISTRATION.

- A. The Committee shall have full and complete authority in its sole discretion, but subject to the express provisions of the Plan: to grant Awards; to determine the exercise or purchase price, if any, of the Shares covered by each Award; to determine the Employees of the Company and of its Subsidiaries to whom, and the time or times at which, Awards shall be granted; to determine the number of Shares to be covered by each Award; to interpret the Plan; to prescribe, amend and rescind rules and regulations relating to the Plan; to determine the terms and provisions of each Award grant and Award Agreement (which terms need not be identical); to determine the vesting schedule of each Award (including the acceleration thereof); to cancel and amend Awards (with the consent of the holder of the Award where required); to impose such conditions on the grant of Awards as it determines to be appropriate, including the surrender of outstanding Awards issued under the Plan or any other stock plan, regardless of the price; and to make all other determinations and rules and such other action deemed necessary or advisable for the administration of the Plan. In addition, the Committee may extend the duration of any NQSO for a period not to exceed one year subject to the provisions of Paragraph VI B hereof without changing the Award price upon such terms as the Committee may deem advisable.
- B. Each determination, interpretation, rule or other action made or taken pursuant to the Plan by the Committee shall be final and conclusive for all purposes and binding upon all persons, including, but without limitation thereto, the Company, Subsidiaries, the Board of Directors, the Committee, Employees of the Company and its Subsidiaries and Participants and their respective successors in interest.
- C. The Committee shall consist of not less than two (2) directors. In the event any class of equity security of the Company is registered pursuant to Section 12 of the Securities Exchange Act of 1934 ("34 Act"), each member of the Committee shall be a member of the Board of Directors who is not eligible to participate under the Plan and who has not been granted or awarded equity

Stock Option Plan

securities of the Company for at least one year prior to the time the director becomes a member of the Committee or during such service on the Committee pursuant to the Plan or any other "plan" within the meaning of Rule 16b-3 promulgated under the 34 Act, except as otherwise permitted under Rule 16b-3 (or any successor rule or regulation).

- D. The Board of Directors may designate one (1) of the members of the Committee as its chairperson and the Committee shall hold its meetings at such times and places as it shall deem advisable. A majority of its members shall constitute a quorum. All determinations of the Committee shall be made by a majority of its members present at a meeting at which a quorum was present. Any decision or determination reduced to writing and signed by all the members of the Committee shall be effective as if it had been made by a vote at a meeting duly called and held. The Committee shall keep minutes of its meetings and shall make such rules and regulations for the conduct of its business as it shall deem advisable.
- E. No member of the Committee shall be liable for any action or determination made in good faith with respect to the administration of the Plan and the granting of Awards thereunder.

IV. ELIGIBILITY AND LIMITATIONS.

Awards may be granted only to Employees of the Company or of any Subsidiary or Parent. Persons who are not Employees of the Company or of a Subsidiary or Parent will not be eligible to receive an ISO. In determining the number of shares to be covered by each Award, subject to Paragraph V hereof, and persons to whom Awards shall be granted, the Committee shall take into account such factors as it shall deem relevant in connection with accomplishing the purpose of the Plan as set forth in Paragraph I hereof. Any person who has been granted an Award may be granted an additional Award or Awards if the Committee shall so determine. No ISO shall be granted to an individual who, at the time the ISO is granted, owns (within the meaning of subsection 422(b)(6) of the Code) stock possessing more than 10% of the total combined voting power of all classes of stock of the Company or of its Parent or any Subsidiary, unless, at the time the ISO is granted, the Award price is at least 110 percent (110%) of the fair market value of the Shares subject to the ISO, and the ISO by its terms is not exercisable after the expiration of five (5) years from the date the ISO is granted.

A. ISOs granted to an Optionee in excess of the limitations set forth in subsection 422(d) of the Code for any calendar year shall be deemed to be a Non-Qualified Stock Award.

B. Each Award must be granted prior to the 10th anniversary of the adoption of the Plan by the Board of Directors.

C. So long as the Company is a "publicly held corporation" within the meaning of Section 162(m) of the Code, no Employee may be granted one or more Options or Stock Awards within any fiscal year of the Company to purchase more than 750,000 Shares, subject to adjustment under Paragraph V.B. If an Award is cancelled without being exercised or if the option price of an Option is reduced, that cancelled or repriced Award shall continue to be counted against the limit on Awards that may be granted to any individual under this Paragraph IV.C.

V. AVAILABLE SHARES AND STOCK ADJUSTMENTS.

- A. The total number of Shares that may be issued pursuant to Awards granted under the Plan shall not exceed 7,000,000 Shares, subject to adjustment as set forth hereinafter. Shares subject to the Plan may be either authorized but unissued Shares or Shares that were once issued and subsequently reacquired by the Company. If an Award is terminated, expires, or otherwise becomes unexercisable without having been exercised in full or otherwise without the Shares covered by the Award having been issued in full, the unpurchased or unissued Shares that were subject to the Award shall revert to the Plan and shall again be available for future issuance under the Plan. The Company will reserve and keep available a sufficient number of authorized but unissued Shares and/or treasury Shares to be issued upon the exercise of the Options.
- B. In the event of a stock split, reverse stock split, stock dividend, or a reclassification of the Shares or other similar action by the Company, the total number of Shares which may be issued under the Plan, the number of Shares that may be granted to any individual under the Plan, including under Paragraph IV.C. of the Plan and the total number of Shares and/or the option price or purchase price, if any, of any outstanding Option or Stock Award, shall be proportionately adjusted by the Board of Directors. Any such adjustment in the number of Shares and/or option price of an ISO shall be made in such manner as to not constitute a modification as defined in Subsection 424(h) (3) of the Code and only to the extent permitted by Sections 422 and 424 of the Code.
- C. In the event of any merger or consolidation or other reorganization in which the Company shall be the surviving entity and its stockholders retain all of the Shares held immediately prior to such event and receive no securities or other property, there shall be no change in the securities or the number of Shares covered by outstanding Awards or the option, purchase or repurchase price of any outstanding Awards.
- D. In the event of any merger or consolidation or other reorganization 'in which the Company shall be the surviving entity and its stockholders have a right to receive securities for or other property in addition to, the outstanding Shares held, each holder of an outstanding Award shall be entitled to receive, upon the exercise, vesting or settlement of the Award, in lieu of the number of Shares as to which such holder of the Award would otherwise have been entitled to receive upon the exercise, vesting or settlement of the Award immediately prior to such merger or consolidation or other reorganization, the number and class of shares and other securities and other property to which such holder of the Award would have been entitled to receive (or retain) pursuant to the terms of the merger or consolidation or other reorganization, such holder of record of a number of Shares equal to the number of Shares to which such Award then covers. Comparable rights shall accrue to each holder of an Award in the event of successive mergers or consolidations.

Stock Option Plan

- E. In the event of any merger or consolidation or other reorganization, in which the Company is not the surviving corporation and the stockholders of the Company shall not receive any equity securities of the surviving entity (or its Parent) for their Shares, except as hereinafter set forth, all Awards (whether or not vested in whole or in part) which have not been exercised or settled prior to or upon such event, shall terminate upon such event unless and to the extent the Board of Directors shall have provided for the substitution of other Awards for, or for the assumption by the surviving corporation (or its Parent) of any unexercised, unvested or unsettled Awards then outstanding. Such action by the Board of Directors may be taken with respect to ISO's only to the extent permitted by the Code, including Sections 422 and 424. Except to the extent the Board of Directors shall have provided for the substitution of other Awards for, or for the assumption by another corporation of, any unexercised, unvested or unsettled Awards then outstanding or shall have provided for the substitution of other Awards for, or for the assumption by another corporation of, any unexercised, unvested or unsettled Awards then outstanding or shall have provided for the substitution of other Awards for, or for the assumption by this Subparagraph E, the Awards which have not vested shall not become exercisable or settleable upon such event and all outstanding Awards shall expire upon such event.
- F. In the event of any merger or consolidation or other reorganization in which the Company is not the surviving entity and in which its stockholders shall receive equity securities regardless of whether they receive other property) for their Shares, each holder of an outstanding Award shall be entitled to receive, upon the exercise, vesting or settlement of the Award, in lieu of the number of Shares as to which such holder of the Award would otherwise have been entitled to receive upon the exercise, vesting or settlement of the Award immediately prior to such merger or consolidation or other reorganization, the number and class of shares and other securities and other property to which such holder of the Award would have been entitled to receive pursuant to the terms of the merger or consolidation or other reorganization if, at the time of such merger or consolidation or other reorganization, such holder of the Award had been the holder of a number of Shares equal to the number of Shares to which such Award then covers. Comparable rights shall accrue to each holder of an Award in the event of successive mergers or consolidations or reorganizations.
- G. Upon the dissolution or liquidation of the Company, all Awards, whether or not vested in whole or in part, which have not been exercised prior to such event shall terminate upon such event.
- H. Any adjustments pursuant to this Paragraph V may provide for the elimination of any fractional interest which might otherwise become subject to an Award, with or without consideration, as determined by the Board of Directors.

VI. OPTION TERMS.

The Options will be granted under terms and conditions set forth in a written instrument as determined by the Committee from time to time. The Options will include (but not by way of limitation) the following:

Stock Option Plan

- A. Price and Payment. The purchase price of each Share covered by each Option as determined by the Committee. The purchase price of each Share covered by an ISO shall not be less than the fair market value of a Share at the time of the granting of the Option. The purchase price of each Share covered by NQSO may be less than or more than the fair market value of a Share at the time of the granting of the Option. The purchase price of the Shares to which an Option shall be exercised shall be paid in full at the time of the exercise in cash or by check, subject to collection. With respect to Options granted to Employees of the Company, the Committee may also provide that the purchase price may be paid in whole or in part by assigning to the Company a number of Shares having a fair market value, determined as of the date the Option is exercised, equal to the amount of the purchase price for the Shares being acquired upon the exercise of the Option which the Committee permits to be paid by the assigning of Shares to the Company. In such event, the Committee may, in its sole discretion, require certain representations and other conditions precedent to the acceptance of the Shares from the Optionee.
- B. Duration. The duration of the Options shall be as determined by the Committee, but in no event shall an Option granted hereunder be exercisable after the earliest of any of the following dates: (i) the expiration of ten (10) years from the date the Option is granted; (ii) one (1) year after the cessation of employment of the holder of the Option with the Company, any Subsidiary, or the Parent, except in the event of termination of such employment by reason of disability, death or retirement; (iii) two (2) years after the cessation of such employment in the event of termination of employment due to death, disability (within in the meaning of Subsection 422(c)(6) of the Code) or retirement. The Committee's determination as to whether such employment of an Optionee has ceased and the effective date thereof shall be final and conclusive on all persons affected thereby. Whether military or other government or eleemosynary service or other leave of absence will constitute termination of such employment shall be determined in each case by the Committee in its sole discretion.
- C. Non-transferability. ISO's granted under the Plan shall not be transferable otherwise than by will or the laws of descent and distribution or as otherwise permitted pursuant to the Code. ISOs may be exercised during the lifetime of the Optionee only by the Optionee personally or by the Optionee's legal representative.
- D. Exercise of Option. Options granted hereunder shall be exercisable in whole or in part as determined by the Committee.
- E. Conditions to Exercise of Options. Shares shall not be issued with respect to any Option granted under the Plan unless the issuance and delivery of such Shares shall comply with (or be exempt from) all relevant provisions of law, including, without limitation, the Securities Act of 1933, as amended, the rules and regulations promulgated thereunder, any applicable state securities law, and the requirements of any stock exchange or nation market system on which the Shares may then be listed. If the issuance or transfer of Shares to be issued or issued pursuant to any Option granted under the Plan may in the opinion of counsel to the Company conflict or be inconsistent with or not be permitted under any applicable law or regulation of any governmental agency having jurisdiction, including, without limitation, regulations promulgated pursuant to federal and state securities laws, the Company reserves the right to delay the issuance of the Shares upon the exercise of an Option and such delay shall be without liability to or other obligation of

the Company. The Company shall have no obligation hereunder to file registration statements or other reports or notices or obtain any license or permit or exemption under any federal or state law with respect to the grant of an Option or the issuance of Shares upon the exercise of an Option or the transfer of such Shares at any time thereafter. The Board of Directors or Committee may require that the holder of an Option, as a condition to each exercise of the Option in whole or in part to represent to the Company in writing that the Shares to be acquired upon the exercise of the Option are to be acquired by the holder of the Option for investment purposes only, for such person's own account, and not with a view to distribution and make such other representations as counsel to the Company may reasonably request to assure the availability of an exemption from or compliance with the registration, notice, reporting or permitting requirements of applicable federal or state securities laws. The Option may also set forth such other terms and conditions relating to the non registration or qualification of the Shares or the issuance of the Shares by the Company or the transfer of the Shares by the Optionee under the federal and state securities laws, as the Board of Directors or Committee may prescribe. Such representations and other terms and conditions shall continue in effect long as counsel to the Company may reasonably request.

- F. Disposition of Shares. In the event the disposition of Shares acquired upon the exercise of any Option is not covered by a then current registration statement under the Securities Act of 1933, as amended, and under applicable state securities laws, the Shares so purchased shall be restricted against transfer to the extent and for as long as required by such laws and regulations promulgated thereunder or until, and as long as, the Shares are covered by applicable registration statements filed by the Company in its sole discretion.
- G. Tax Withholdings. In any case where the Company, any Subsidiary or the Parent is obliged to account for (i) any tax (or similar liabilities) in any jurisdiction, and/or (ii) employee and/or employer social security contributions (or similar liabilities) in any jurisdiction, by virtue of the exercise, release or assignment of the Option acquisition and holding of Shares (together, the "Tax Liability"), the Company, Subsidiary or the Parent as the case may be, may recover the Tax Liability from the Optionee manner as the Board of Directors shall think fit and (without prejudice to the generality of the foregoing) Shares shall not be transferred to the Optionee unless the Optionee either (iii) made a payment to the Company, Subsidiary or the Parent, as the case may be, of an amount equal to the estimated Tax Liability or (iv) entered into arrangements the Company, Subsidiary or the Parent, as the case may be, to secure that such a payment is made.
- H. National Insurance Contributions Joint Election. Without prejudice to the generality of clause VI above (Conditions to Exercise of Options), the Optionees who are subject to the laws of the United Kingdom may not exercise, release or assign an Option granted under the Plan in any circumstances unless and until the Board of Directors is satisfied that the Optionee has entered into a binding election in the form prescribed by the Company pursuant to which the Optionee assumes liability for the whole of the employers' National Insurance contributions in respect of share option gains arising from the Option.
- I. Executive Officers. In the event any class of equity security of the Company is registered pursuant to Section 12 of the 34 Act, any election granted to an executive officer (as defined pursuant to rules promulgated under the 1934 Act) of the Parent shall only be made during the period set forth in Rule 16b-3 promulgated under the 1934 Act (or any successor rule or regulation), if any.

Stock Option Plan

VII. EXERCISE.

An Option granted hereunder shall be exercisable in whole or in part only by written notice delivered in person or by mail to the President of the Company or such other officer designated by the President, at its principal executive office, specifying the number of Shares to be purchased and accompanied by payment therefor and other consideration in accordance with the Option. The holder of an Option shall not be deemed to be a holder of any Shares subject to any Option and shall not be entitled to the rights of a holder of any Shares, including the right to vote the Shares and to receive dividends, unless and until such Shares have been issued.

VIII. STOCK GRANTS AND STOCK UNIT AWARDS.

Each Stock Award Agreement reflecting the issuance of a Stock Grant or Stock Unit shall be in such form and shall contain such terms and conditions as the Committee shall deem appropriate. The terms and conditions of such agreements may change from time to time, and the terms and conditions of separate agreements need not be identical, but each such agreement shall include (through incorporation of provisions hereof by reference in the agreement or otherwise) the substance of each of the following provisions:

- A. Consideration. A Stock Grant or Stock Unit may be awarded in consideration for such property or services as is permitted under applicable law, including for past services actually rendered to the Company or a Parent or Subsidiary for its benefit.
- B. Vesting. Shares of Common Stock awarded under an agreement reflecting a Stock Grant and a Stock Unit award may, but need not, be subject to a share repurchase option, forfeiture restriction or other conditions in favor of the Company in accordance with a vesting or lapse schedule to be determined by the Committee.
- C. Termination. In the event of a Participant's termination of employment for any reason, the Company may reacquire any or all of the Shares held by the Participant which have not vested or which are otherwise subject to forfeiture or other conditions as of the date of termination of employment under the terms of the Stock Award Agreement.
- D. Transferability. Except as determined by the Board of Directors and reflected in the Stock Award Agreement, no rights to acquire Shares under a Stock Grant or a Stock Unit shall be assignable or otherwise transferable by the Participant except by will or by the laws of descent and distribution.

IX. STOCK APPRECIATION RIGHTS.

A. In General. Stock Appreciation Rights may be granted either alone, in addition to, or in tandem with other Awards granted under the Plan. The Committee may grant Stock Appreciation Rights to eligible Participants subject to terms and conditions not inconsistent with the Plan and determined by the Committee. The specific terms and conditions applicable to the Participant shall be provided for in the Stock Award Agreement. Stock Appreciation Rights shall be exercisable, in whole or in part, at such times as the Committee shall specify in the Stock Award Agreement.

Stock Option Plan

- B. Exercise of Stock Appreciation Right. Upon the exercise of a Stock Appreciation Right, in whole or in part, the Participant shall be entitled to a payment in an amount equal to the excess of the fair market value, as determined by the Committee, on the date of exercise of a fixed number of Shares covered by the exercised portion of the Stock Appreciation Right, over the fair market value, as determined by the Committee, on the grant date of the Shares covered by the exercised portion of the Stock Appreciation Right (or if reflected in the Stock Award Agreement, such other amount calculated with respect to Shares subject to the award as the Committee may determine). The amount due to the Participant upon the exercise of a Stock Appreciation Right shall be paid in such form of consideration as determined by the Committee and may be in cash, Shares or a combination thereof, over the period or periods specified in the Stock Award Agreement. A Stock Award Agreement may place limits on the amount that may be paid over any specified period or periods upon the exercise of a Stock Appreciation Right, on an aggregate basis or as to any Participant. A Stock Appreciation Right shall be considered exercised when the Company receives written notice of exercise in accordance with the terms of the Stock Award Agreement from the person entitled to exercise the Stock Appreciation Right.
- C. Transferability. Except as determined by the Board of Directors and reflected in the Stock Award Agreement, no Stock Appreciation Rights shall be assignable or otherwise transferable by the Participant except by will or by the laws of descent and distribution.

X. SECTION 162(m) COMPLIANCE.

Any Stock Award (other than an Option or any other Stock Award having a purchase price equal to 100% of the fair market value of the Shares, as determined by the Committee, on the date such Award is made) that is intended as "qualified performance-based compensation" within the meaning of Section 162(m) of the Code must vest or become exercisable contingent on the achievement of one or more Qualifying Performance Criteria. Notwithstanding anything to the contrary herein, the Committee shall have the discretion to determine the time and manner of compliance with Section 162(m) of the Code as required under applicable regulations and to conform the procedures related to the Award to the requirements of Section 162(m) of the Code and may reduce the number of Shares granted or amount of cash or other property to which a Participant may otherwise have been entitled with respect to an Award designed to qualify as performance-based compensation under Section 162(m) of the Code.

XI. TERMINATION AND AMENDMENT.

The Board of Directors may at any time terminate the Plan, or make such amendments thereto or modifications thereof as it shall deem advisable, including amendments deemed necessary or desirable to conform any ISO to any change in the Code or regulations thereto provided, however, that the Board of Directors may not, without further approval by the stockholders of the Company, increase the maximum number of Shares for which Awards may be granted under the Plan or change the designation of the class of employees and other persons eligible to receive Awards. In addition, unless approved by the stockholders of the Company, no

Stock Option Plan

amendment shall be made that would result in a repricing of Options by (x) reducing the exercise price of outstanding Options or (y) canceling an outstanding Option held by a Participant and re-granting to the Participant a new Option with a lower exercise price, in either case other pursuant to Paragraph V.B. of the Plan. No termination, modification or amendment of the Plan shall, without the consent of the Participant to whom an Award shall theretofore have been granted adversely affect the rights of such Participant under such Award without the written consent of such Participant.

XII. MISCELLANEOUS.

- A. Applicable Law. The Plan shall be governed and construed in accordance with the laws of the State of Delaware.
- B. Employee/Employer Rights. The granting of Awards hereunder shall be entirely discretionary and nothing in the Plan shall be deemed to give any person any right of continued employment or give any person any right to receive Awards or additional Awards hereunder or interfere in any way with the right of the Company, its Parent or Subsidiary to terminate the Participant's employment for any reason or the right of the Participant to terminate his/her employment for any reason. Without prejudice to the generality of the foregoing, participation in this Plan is a matter entirely separate from any pension right or entitlement a Participant may have and from their terms and conditions of employment and in particular (but without limiting the generality of the foregoing) any individual who leaves the employment of the Company or its Subsidiaries, or the Parent or who otherwise ceases to be an employee shall not be entitled to any compensation for any loss of any right or benefit or prospective or benefit under the Plan which the Participant might otherwise have enjoyed whether such compensation is claimed by way of damages for wrongful dismissal or other breach of contract or by way of compensation for loss of office, employment or otherwise howsoever.
- C. ISO Grants. The Plan is intended to provide in part for the grant of ISO's pursuant to Section 422 of the Code, including amendments thereto hereafter adopted, and the provisions of the Plan as they relate to ISO's and the ISO's granted shall be construed to effectuate such purpose. If for any reason it is subsequently determined that an intended to qualify as an ISO does not so qualify, the Company, Parent and Subsidiary shall have no liability to the Participant and such Options shall be deemed to be Non-Qualified Stock Options.

XIII. EFFECTIVE DATE.

The Plan shall become effective on the date of its adoption by the Board of Directors subject to the approval of the Plan by the stockholders of the Company within twelve (12) months after the date of its adoption. The date of granting of an Award shall be the date on which the Committee makes the determination of granting such Award or such later date as designated by the Committee.

Stock Option Plan

2002 STOCK INCENTIVE PLAN OF SECURE COMPUTING CORPORATION

1. Purpose of this Plan

The purpose of this 2002 Stock Incentive Plan of Secure Computing Corporation is to enhance the long-term stockholder value of Secure Computing Corporation by offering opportunities to eligible individuals to participate in the growth in value of the equity of Secure Computing Corporation.

2. Definitions and Rules of Interpretation

2.1 Definitions. This Plan uses the following defined terms:

(a) "Administrator" means the Board, the Committee, or any officer or employee of the Company to whom the Board or the Committee delegates authority to administer this Plan.

(b) "Affiliate" means a "parent" or "subsidiary" (as each is defined in Section 424 of the Code) of the Company and any other entity that the Board or Committee designates as an "Affiliate" for purposes of this Plan.

(c) "*Applicable Law*" means any and all laws of whatever jurisdiction, within or without the United States, and the rules of any stock exchange or quotation system on which Shares are listed or quoted, applicable to the taking or refraining from taking of any action under this Plan, including the administration of this Plan and the issuance or transfer of Options, Stock Awards, Shares or Option Shares.

(d) "Award" means an Option or a Stock Award granted in accordance with the terms of the Plan.

(e) "Award Agreement" means a Stock Award Agreement and/or Option Agreement.

(f) "Award Shares" means Shares covered by an outstanding Award or purchased under an Award.

(g) "Board" means the board of directors of the Company.

(h) "Code" means the Internal Revenue Code of 1986, as amended.

(i) "Committee" means a committee composed of Company Directors appointed in accordance with the Company's charter documents and Section 4.

(j) "Company" means Secure Computing Corporation, a Delaware corporation.

(k) "Company Director" means a member of the Board.

(1) "Consultant" means an individual who, or an employee of any entity that, provides bona fide services to the Company or an Affiliate not in connection with the offer or sale of securities in a capital-raising transaction, but who is not an Employee.

(m) "Director" means a member of the board of directors of the Company or an Affiliate.

(n) "Divestiture" means any transaction or event that the Board specifies as a Divestiture under Section 11.5.

(o) "Effective Date" means the effective date of this Plan.

(p) "Employee" means a regular employee of the Company or an Affiliate, including an officer or Director, who is treated as an employee in the personnel records of the Company or an Affiliate, but not individuals who are classified by the Company or an Affiliate as: (i) leased from or otherwise employed by a third party, (ii) independent contractors, or (iii) intermittent or temporary workers. The Company's or an Affiliate's classification of an individual as an "Employee" (or as not an "Employee") for purposes of this Plan shall not be altered retroactively even if that classification is changed retroactively for another purpose as a result of an audit, litigation, or otherwise. A Participant shall not cease to be an Employee due to transfers between locations of the Company, or between the Company and an Affiliate, or to any successor to the Company or an Affiliate that assumes the Participant's Options or Stock Awards under Section 11. Neither service as a Director nor receipt of a director's fee shall be sufficient to make a Director an "Employee."

(q) "Event" means any transaction or event that the Board specifies as an Event under Section 11.4.

(r) "Exchange Act" means the Securities Exchange Act of 1934, as amended.

(s) *"Executive"* means, if the Company has any class of any equity security registered pursuant to Section 12 of the Exchange Act, an individual who is subject to Section 16 of the Exchange Act or who is a "covered employee" under Section 162(m) of the Code, in either case because of the individual's relationship with the Company or an Affiliate. If the Company does not have any class of any equity security registered under to Section 12 of the Exchange Act, "Executive" means any (i) Director, (ii) any officer elected or appointed by the Board, or (iii) any beneficial owner of more than 10% of any class of the Company's equity securities.

(t) *"Expiration Date"* means, with respect to an Option, the date stated in the Option Agreement as the expiration date of the Option or, if no such date is stated in the Option Agreement, then the last day of the maximum exercise period for the Option, disregarding the effect of an Optionee's Termination or any other event that would shorten that period.

(u) "Fair Market Value" means the value of Shares as determined under Section 20.2.

(v) "Fundamental Transaction" means any transaction or event described in Section 11.3.

(w) "Grant Date" means the date the Administrator approves the grant of an Option. However, if the Administrator specifies that an Option's Grant Date is a future date or the date on which a condition is satisfied, the Grant Date for such Option is that future date or the date that the condition is satisfied.

(x) *"Incentive Stock Option"* means an Option intended to qualify as an incentive stock option under Section 422 of the Code and designated as an Incentive Stock Option in the Option Agreement for that Option.

(y) "Incumbent Board" shall have the meaning set forth in Section 11.4.

- (z) "Nonstatutory Option" means any Option other than an Incentive Stock Option.
- (aa) "Officer" means an officer of the Company as defined in Rule 16a-1 adopted under the Exchange Act.
- (bb) "Option" means an option to purchase Shares of the Company granted under this Plan.
- (cc) "Option Agreement" means the document evidencing the grant and terms and conditions of an Option.

(dd) "Option Price" means the price payable under an Option to purchase the Shares covered thereby, not including any amount payable in respect of withholding or other taxes.

(ee) "Option Shares" means Shares covered by an outstanding Option or purchased under an Option.

(ff) *"Optionee"* means: (i) a person to whom an Option has been granted, including a holder of a Substitute Option, (ii) a person to whom an Option has been transferred in accordance with all applicable requirements of Sections 6.5, 7(h), and 18, and (iii) a person who holds Option Shares subject to any right of repurchase under Section 18.2.

(gg) "Outstanding Company Common Stock" shall have the meaning set forth in Section 11.4.

(hh) "Outstanding Company Voting Securities" shall have the meaning set forth in Section 11.4.

(ii) "Participant" means any holder of one or more Options or Stock Awards or the Shares issuable or issued upon exercise of such Options or Stock Awards under the Plan.

(jj) "Plan" means this 2002 Stock Incentive Plan of Secure Computing Corporation.

(kk) "Qualified Domestic Relations Order" means a judgment, order, or decree meeting the requirements of Section 414(p) of the Code except that references to the "plan" in that definition shall be to this Plan.

(ll) "*Qualifying Performance Criteria*" means any one or more of the following performance criteria, either individually, alternatively or in any combination, applied to either the Company as a whole or to a business unit, Affiliate or business segment, either individually, alternatively or in any combination, and measured either annually or cumulatively over a period of years, on an absolute basis or relative to a pre-established target, to previous years' results or to a designated comparison group, in each case as specified by the Committee in the Award: (i) cash flow; (ii) earnings (including gross margin, earnings before interest and taxes, earnings before taxes, and net earnings); (iii) earnings per share; (iv) growth in earnings or earnings per share; (v) stock price; (vi) return on equity or average stockholders' equity; (vii) total stockholder return; (viii) return on capital; (ix) return on assets or net assets; (x) return on investment; (xi) income or net income; (xiii) operating income or net operating income; (xiv) operating profit or net operating profit; (xv) operating margin; (xvi) return on operating revenue; (xvii) market share; (xviii) contract awards or backlog; (xix) overhead or other expense reduction; (xx) growth in stockholder value relative to the moving

average of the S&P 500 Index or a peer group index; (xxi) credit rating; (xxii) strategic plan development and implementation (including individual performance objectives that relate to achievement of the Company's or any Affiliate's or business unit's strategic plan); (xxiii) improvement in workforce diversity, and (xxiv) any other similar criteria as may be determined by the Administrator. The Committee may appropriately adjust any evaluation of performance under a Qualifying Performance Criteria to exclude any of the following events that occurs during a performance period: (A) asset write-downs; (B) litigation or claim judgments or settlements; (C) the effect of changes in tax law, accounting principles or other such laws or provisions affecting reported results; (D) accruals for reorganization and restructuring programs; and (E) any gains or losses classified as extraordinary or as discontinued operations in the Company's financial statements.

(mm) *"Reverse Vesting"* means that an Option is or was fully exercisable but that, subject to a "reverse" vesting schedule, the Company has a right to repurchase the Option Shares as specified in Section 18.2(a), with the Company's right of repurchase expiring in accordance with a "forward" vesting schedule that would otherwise have applied to the Option under which the Option Shares were purchased or other vesting schedule described in the Option Agreement.

(nn) "Rule 16b-3" means Rule 16b-3 adopted under Section 16(b) of the Exchange Act including any successor provisions.

(00) "Securities Act" means the Securities Act of 1933, as amended.

(pp) "Share" means a share of the common stock of the Company or other securities substituted for the common stock under Section 11.

(qq) "Stock Appreciation Right" means a right to receive cash and/or Shares based on a change in the Fair Market Value of a specific number of Shares granted under Section 10.

(rr) "Stock Award" means a Stock Grant, a Stock Unit or a Stock Appreciation Right granted under Sections 9 or 10 below or other similar awards granted under the Plan (including phantom stock rights).

(ss) "Stock Award Agreement" means a written agreement, the form(s) of which shall be approved from time to time by the Administrator, between the Company and a holder of a Stock Award evidencing the terms and conditions of an individual Stock Award grant. Each Stock Award Agreement shall be subject to the terms and conditions of the Plan.

(tt) "Stock Grant" means the award of a certain number of Shares granted under Section 9 below.

(uu) "Stock Unit" means a bookkeeping entry representing an amount equivalent to the Fair Market Value of one Share, payable in cash, property or Shares. Stock Units represent an unfunded and unsecured obligation of the Company, except as otherwise explicitly provided for by the Administrator.

(vv) "Substitute Award" means an Award granted in substitution for, or upon the conversion of, an option or stock award granted by another entity to purchase equity securities in the granting entity.

(ww) "*Termination*" means that the Participant has ceased to be, with or without any cause or reason, an Employee, Director, or Consultant. However, unless so determined by the Administrator, "Termination" shall not include a change in status from an Employee, Consultant, or Director to another such status. An event that causes an Affiliate to cease being an Affiliate shall be treated as the "Termination" of that Affiliate's Employees, Directors, and Consultants.

2.2 **Rules of Interpretation.** Any reference to a "Section," without more, is to a Section of this Plan. Captions and titles are used for convenience in this Plan and shall not, by themselves, determine the meaning of this Plan. Except when otherwise indicated by the context, the singular includes the plural and vice versa. Any reference to a statute is also a reference to the applicable rules and regulations adopted under that statute. Any reference to a statute, rule or regulation, or to a section of a statute, rule or regulation, is a reference to that statute, rule, regulation, or section as amended from time to time, both before and after the Effective Date and including any successor provisions.

3. Shares Subject to this Plan; Term of this Plan

3.1 Number of Award Shares. Subject to adjustment under Section 11, the maximum number of Shares that may be issued under this Plan is 5,000,000. If an Award is terminated, expires, or otherwise becomes unexercisable without having been exercised in full or otherwise without the Shares covered by the Award having been issued in full, the unpurchased or unissued Shares that were subject to the Award shall revert to this Plan and shall again be available for future issuance under this Plan. Any Shares of Common Stock which are retained by the Company upon exercise of an Award issued under the Plan in order to satisfy the exercise or purchase price for such Award or any withholding taxes due with respect to such exercise, purchase or issuance shall continue to be available for future issuance under the Plan. Shares actually issued under this Plan shall not be available for regrant even if repurchased by or forfeited to the Company.

3.2 Source of Shares. Shares may be authorized but unissued Shares.

3.3 Term of this Plan

(a) This Plan shall be effective on the date it is approved by the Board. If the Company's stockholders do not approve this Plan within 12 months after the Board approves this Plan, then no Incentive Stock Option may be granted under this Plan.

(b) This Plan has no set termination date. However, it may be terminated as provided in Section 16. Moreover, no Incentive Stock Option may be granted after the time described in Section 7(b).

4. Administration

4.1 General.

(a) The Board shall have ultimate responsibility for administering this Plan. To the extent permitted by Applicable Law, the Board may delegate certain of its responsibilities to a Committee, which shall consist of at least two members of the Board. In addition, to the extent permitted by Applicable Law, the Board or the Committee may further delegate its responsibilities to any Employee of the Company or any Affiliate. Where this Plan specifies that an action is to be taken or a determination made by the Board, only the Board may take that action or make that determination. Where this Plan references the Administrator, the action may be taken or determination made by the Board, the Committee, or other Administrator. However, only the Board or the Committee may approve grants of Options to Executives, and an Administrator other than the Board or the Committee may grant Options only within guidelines established by the Board or the Committee. Moreover, all actions and determinations by any Administrator are subject to the provisions of this Plan.

(b) So long as the Company has registered and outstanding a class of equity securities under Section 12 of the Exchange Act and to the extent necessary or helpful to comply with Applicable Law with respect to officers subject to Section 16 of the Exchange Act and/or others, the Committee shall consist of Company Directors who are "Non-Employee Directors" as defined in Rule 16b-3 and who are "outside directors" as defined in Section 162(m) of the Code.

4.2 Authority of Administrator. Subject to the other provisions of this Plan, the Administrator shall have the authority:

(a) to grant Awards, including Substitute Awards;

(b) to determine the Fair Market Value of Shares;

(c) to determine the Option Price of Options;

(d) to select the Participants to whom Awards may be granted hereunder;

(e) to determine the times Options and Stock Awards are granted;

(f) to determine the number of Shares subject to each Option or Stock Award;

(g) to determine the types of payment that may be used to purchase Shares subject to Awards;

(h) to determine the types of payment that may be used to satisfy withholding tax obligations;

(i) to determine the other terms of each Option or Stock Award, including but not limited to: the time or times at which Options or Stock Awards may vest, be exercised or settled, or become nonforfeitable (including any acceleration related to such terms), whether and under what conditions an Option or Stock Award is assignable, and whether an Option is a Nonstatutory Option or an Incentive Stock Option and any other conditions that are to apply to the Award;

(j) to modify or amend any Option or Stock Award;

(k) to authorize any person to sign any Award Agreement or other document related to this Plan on behalf of the Company;

(1) to determine the form of any Award Agreement or other document related to this Plan, and whether that document, including signatures, may be in electronic

form;

(m) to interpret this Plan and any Award Agreement or document related to this Plan;

(n) to correct any defect, remedy any omission, or reconcile any inconsistency in this Plan, any Option Agreement or Stock Award Agreement or any other document related to this Plan;

(o) to adopt, amend, and revoke rules and regulations under this Plan, including rules and regulations relating to sub-plans and Plan addenda;

(p) to adopt, amend, and revoke rules and procedures relating to the operation and administration of this Plan to accommodate non-U.S. Participants and the requirements of Applicable Law such as: (i) rules and procedures regarding the conversion of local currency, withholding procedures and the handling of stock certificates to comply with local practice and requirements, and (ii) sub-plans and Plan addenda for non-U.S. Participants;

(q) to determine whether a transaction or event should be treated as an Event, a Divestiture or neither;

(r) to determine the effect of a Fundamental Transaction and, if the Board determines that a transaction or event should be treated as an Event or a Divestiture, then the effect of that Event or Divestiture; and

(s) to make all other determinations the Administrator deems necessary or advisable for the administration of this Plan.

4.3 **Scope of Discretion**. Subject to the last sentence of this Section 4.3, on all matters for which this Plan confers the authority, right, or power on the Board, the Committee, or other Administrator to make decisions, that body may make those decisions in its sole and absolute discretion. Moreover, but again subject to the last sentence of this Section 4.3, in making those decisions the Board, the Committee, or other Administrator need not treat all persons eligible to receive Awards, all Participants, all Awards or all Shares subject to Awards the same way. However, the discretion of the Board, the Committee, or other Administrator is subject to the specific provisions and specific limitations of this Plan, as well as all rights conferred on specific Participants by Award Agreements and other agreements.

5. Persons Eligible to Receive Awards

5.1 Eligible Individuals. Options and Stock Awards may be granted to, and only to, Employees, Directors and Consultants, including to prospective Employees, Directors and Consultants conditioned on the beginning of their service for the Company or an Affiliate, provided that Incentive Stock Options many only be granted to Employees, as provided in Section 7(g).

5.2 Section 162(m) Limitation. So long as the Company is a "publicly held corporation" within the meaning of Section 162(m) of the Code, no Employee or prospective Employee may be granted one or more Options or Stock Awards within any fiscal year of the Company to purchase more than 750,000 Shares, subject to adjustment under Section 11. If an Award is cancelled without being exercised or if the Option Price of an Option is reduced, that cancelled or repriced Award shall continue to be counted against the limit on Awards that may be granted to any individual under this Section 5.2.

6. Terms and Conditions of Options

The following rules apply to all Options:

6.1 Price. No Option may have an Option Price less than 50% of the Fair Market Value of the Shares on the Grant Date. No Option intended as "qualified incentive-based compensation" within the meaning of Section 162(m) of the Code may have an Option Price less than 100% of the Fair Market Value of the Shares on the Grant Date. In no event will the Option Price of any Option be less than the par value of the Shares issuable under the Option if that is required by Applicable Law. The Option Price of an Incentive Stock Option shall be subject to Section 7(f).

6.2 No Option Repricings. Other than in accordance with Section 11, Options may not be repriced, replaced, regranted through cancellation or modified without stockholder approval, if the effect of the repricing, replacement, regrant or modification would be to reduce the effective Option Price of the Options.

6.3 Term. No Option shall be exercisable after its Expiration Date. No Option may have an Expiration Date that is more than 10 years after its Grant Date. The term of an Incentive Stock Option shall be subject to Sections 7(a) and 7(e).

6.4 Vesting.

(a) Options shall be exercisable: (a) on the Grant Date, or (b) in accordance with a schedule related to the Grant Date, the date the Optionee's directorship, employment, or consultancy begins, or a different date specified in the Option Agreement. If so provided in the Option Agreement, an Option may be exercisable subject to the application of Reverse Vesting to the Option Shares. The vesting of Incentive Stock Options shall be subject to Section 7(c).

(b) The Administrator shall have the discretion to determine whether and to what extent the vesting of Options shall be tolled during any unpaid leave of absence; provided, however, that in the absence of such determination, vesting of Options shall be tolled during any leave that is not a leave required to be provided to the Optionee under Applicable Law. In the event of military leave, vesting shall toll during any unpaid portion of such leave, provided that, upon an Optionee's returning from military leave (under conditions that would entitle him or her to protection upon such return under the Uniform Services Employment and Reemployment Rights Act), he or she shall be given vesting credit with respect to Options to the same extent as would have applied had the Optionee continued to provide services to the Company throughout the leave on the same terms as he or she was providing services immediately prior to such leave.

6.5 Form of Payment

(a) The Administrator shall determine the acceptable form and method of payment for exercising an Option.

(b) Acceptable forms of payment for all Option Shares are cash, check or wire transfer, denominated in U.S. dollars except as specified by the Administrator for non-U.S. Employees or non-U.S. sub-plans.

(c) In addition, the Administrator may permit payment of the Option Price to be made by any of the following methods:

(i) other Shares, or the designation of other Shares, which (A) if required to avoid the Company's incurring adverse accounting charges, in the case of Shares acquired upon exercise of an option (whether or not under this Plan) are "mature" shares for purposes of avoiding variable accounting treatment under generally accepted accounting principals (generally, mature shares are those that have been owned by the Optionee for more than six months on the date of surrender), and (B) have a Fair Market Value on the date of surrender equal to the Option Price of the Shares as to which the Option is being exercised;

(ii) provided that a public market exists for the Shares, through a "same day sale" commitment from the Optionee and a broker-dealer that is a member of the National Association of Securities Dealers (an "*NASD Dealer*") under which the Optionee irrevocably elects to exercise the Option and the NASD Dealer irrevocably commits to forward an amount equal to the Option Price plus any applicable withholding taxes, directly to the Company, upon receipt of the Option Shares (a "*Cashless Exercise*");

(iii) cancellation of any debt owed by the Company or by any Affiliate to the Optionee, including, without limitation, waiver of compensation due or accrued for services previously rendered to the Company; and

(iv) any combination of the methods of payment permitted by any paragraph of this Section 6.5.

(d) The Administrator may also permit any other form or method of payment for Option Shares permitted by Applicable Law.

6.6 Nonassignability of Nonstatutory Options. Except as set forth in any Option Agreement or as determined by the Administrator, no Nonstatutory Option shall be assignable or otherwise transferable by the Optionee except by will or by the laws of descent and distribution; provided however, Nonstatutory Options may be transferred and exercised in accordance with a Qualified Domestic Relations Order and Nonstatutory Options may be exercised by a guardian or conservator appointed to act for the Optionee. Notwithstanding the foregoing, Nonstatutory Options may be transferred by instrument to an inter vivos or testamentary trust in which the Nonstatutory Options are to be passed to beneficiaries upon the death of the trustor (settlor) or by gift or pursuant to domestic relations orders to "Immediate Family Members" (as defined below) of the Optionee. "Immediate Family." means any child, stepchild, grandchild, parent, stepparent, grandparent, spouse, former spouse, sibling, niece, nephew, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law, or sister-in-law (including adoptive relationships), a trust in which these persons have more than fifty percent of the beneficial interest, a foundation in which these persons (or the Optionee) control the management of assets, and any other entity in which these persons (or the Optionee) own more than fifty percent of the voting interests. Incentive Stock Options may only be assigned subject to Section 7(h).

7. Incentive Stock Options

The following rules apply only to Incentive Stock Options and only to the extent these rules are more restrictive than the rules that would otherwise apply under this Plan. With the consent of the Optionee, or where this Plan provides that an action may be taken notwithstanding any other provision of this Plan, the Administrator may deviate from the requirements of this Section, notwithstanding that any Incentive Stock Option modified by the Administrator will thereafter be treated as a Nonstatutory Option.

(a) The Expiration Date of an Incentive Stock Option shall not be later than 10 years from its Grant Date, with the result that no Incentive Stock Option may be exercised after the expiration of 10 years from its Grant Date.

(b) No Incentive Stock Option may be granted more than 10 years from the date this Plan was approved by the Board.

(c) Options intended to be Incentive Stock Options that are granted to any single Optionee under all incentive stock option plans of the Company and its Affiliates, including Incentive Stock Options granted under this Plan, may not become exercisable at a rate of more than \$100,000 in Fair Market Value of stock (measured on the grant dates of the options) during any calendar year. For this purpose, an Option becomes exercisable with respect to a given share of stock the first time its holder may purchase that share, notwithstanding any right of the Company to repurchase that share. Unless the Administrator specifies otherwise in the related agreement governing the Option, this limitation shall be applied by, to the extent necessary to satisfy this \$100,000 rule, treating certain stock options that were intended to be Incentive Stock Options as Nonstatutory Options. The stock options or portions of stock options to be reclassified as Nonstatutory Options are those with the highest option prices, whether granted under this Plan or any

other equity compensation plan of the Company or any Affiliate that permits that treatment. This Section 7(c) shall not cause an Incentive Stock Option to become exercisable before its original vesting or exercisability date or cause an Incentive Stock Option that has already vested or become exercisable to cease to be vested or exercisable.

(d) In order for an Incentive Stock Option to be exercised for any form of payment other than those described in Section 6.5(b), that right must be stated in the Option Agreement relating to that Incentive Stock Option.

(e) Any Incentive Stock Option granted to a Ten Percent Stockholder, must have an Expiration Date that is not later than five years from its Grant Date, with the result that no such Option may be exercised after the expiration of five years from the Grant Date. A "*Ten Percent Stockholder*" is any person who, directly or by attribution under Section 424(d) of the Code, owns stock possessing more than 10% of the total combined voting power of all classes of stock of the Company or of any Affiliate on the Grant Date.

(f) The Option Price of an Incentive Stock Option shall never be less than the Fair Market Value of the Shares at the Grant Date. The Option Price for the Shares covered by an Incentive Stock Option granted to a Ten Percent Stockholder shall never be less than 110% of the Fair Market Value of the Shares at the Grant Date.

(g) Incentive Stock Options may be granted only to Employees. If an Optionee changes status from an Employee to a Consultant, that Optionee's Incentive Stock Options become Nonstatutory Options if not exercised within the time period described in Section 7(i).

(h) No rights under an Incentive Stock Option may be transferred by the Optionee, other than by will or the laws of descent and distribution. During the life of the Optionee, an Incentive Stock Option may be exercised only by the Optionee.

(i) An Incentive Stock Option shall be treated as a Nonstatutory Option if it remains exercisable after, and is not exercised within, the three-month period beginning with the Optionee's Termination for any reason other than the Optionee's death or disability (as defined in Section 22(c) of the Code). In the case of Termination due to death, an Incentive Stock Option shall continue to be treated as an Incentive Stock Option if it remains exercisable after, but is not exercised within, the one year period provided it is exercised before the Expiration Date. In the case of Termination due to disability, an Incentive Stock Option shall be treated as a Nonstatutory Option if it remains exercisable after, but is not exercised within, one year after the Optionee's Termination.

(j) An Incentive Stock Option may be modified by the Board.

8. Exercise of Options

8.1 In General. An Option shall be exercisable in accordance with this Plan, the Option Agreement under which it is granted, and as prescribed by the Administrator.

8.2 **Time of Exercise.** An Option shall be considered exercised when the Company receives: (a) written notice of exercise from the person entitled to exercise the Option, (b) full payment, or provision for payment, in a form and method approved by the Administrator, for the Shares for which the Option is being exercised, and (c) with respect to Nonstatutory Options, payment, or provision for payment, in a form approved by the Administrator, of all applicable withholding taxes due upon exercise. An Option may not be exercised for a fraction of a Share.

8.3 **Issuance of Option Shares.** The Company shall issue Option Shares in the name of the person properly exercising the Option. If the Optione is that person and so requests, the Option Shares shall be issued in the name of the Optionee and the Optionee's spouse. The Company shall endeavor to issue Option Shares promptly after an Option is exercised. However, until Option Shares are actually issued, as evidenced by the appropriate entry on the stock books of the Company or its transfer agent, no right to vote or receive dividends or other distributions, and no other rights as a stockholder, shall exist with respect to the Option Shares, even though the Optionee has completed all the steps necessary to exercise the Option. No adjustment shall be made for any dividend, distribution, or other right for which the record date precedes the date the Option Shares are issued, except as provided in Section 11.

8.4 Termination

(a) *In General*. Except as provided in an Option Agreement or in writing by the Administrator, and as otherwise provided in Sections 8.4(b), (c), (d), (e), (f), (g) and (h), after an Optionee's Termination the Optionee's Options shall be exercisable to the extent (but only to the extent) they are vested on the date of that Termination and only during the three months (or such other period of time as is determined by the Administrator) after the Termination, but in no event after the Expiration Date. To the extent the Optionee does not exercise an Option within the time specified for exercise, the Option shall automatically terminate.

(b) *Leaves of Absence*. Unless otherwise provided in the Option Agreement, no Option may be exercised more than three months (or such other period of time as is determined by the Administrator) after the beginning of a leave of absence, other than a personal, medical or military leave approved by the Administrator with employment guaranteed upon return.

(c) **Death or Disability**. Unless otherwise provided by the Administrator or in the Option Agreement, if an Optionee's Termination is due to death or disability (as determined by the Administrator with respect to Nonstatutory Options and as defined by Section 22(e) of the Code with respect to Incentive Stock Options), all Options of that Optionee to the extent they are vested at the date of that Termination may be exercised for one year (or such other period of time as is determined by the Administrator) after that Termination, but in no event after the Expiration Date. In the case of Termination due to death, an Option may be exercised as provided in Section 19. In the case of Termination due to disability, if a guardian or conservator has been appointed to act for the Optionee and been granted this authority as part of that appointment, that guardian or conservator may exercise the Option on behalf of the Optionee. Death or disability occurring after an Optionee's Termination shall not cause the Termination to be treated as having occurred due to death or disability. To the extent an Option is not so exercised within the time specified for its exercise, the Option shall automatically terminate.

(d) Divestiture. If an Optionee's Termination is due to a Divestiture, the Board may take any one or more of the actions described in Section 11.3 or 11.5.

(e) *Retirement*. Unless otherwise provided in the Option Agreement or by the Administrator in writing, if an Optionee's Termination is due to the Optionee's retirement in accordance with the Company's or an Affiliate's retirement policy, all Options of that Optionee to the extent they are vested at the Optionee's date of retirement may be exercised for three months (or such other period of time as is determined by the Administrator) after the Optionee's date of retirement, but in no event after the Expiration Date. To the extent the Optionee does not exercise an Option within the time specified for exercise, the Option shall automatically terminate.

(f) *Severance Programs*. Unless otherwise provided in the Option Agreement or by the Administrator in writing, if an Optionee's Termination results from participation in a voluntary severance incentive program of the Company or an Affiliate

approved by the Board, all Options of that Employee to the extent they are vested at the time of that Termination shall be exercisable for three months (or such other period of time as is determined by the Administrator) after the Optionee's Termination, but in no event after the Expiration Date. If the Optionee does not exercise an Option within the time specified for exercise, the Option shall automatically terminate.

(g) *Termination for Cause*. If an Optionee's Termination is due to Cause (as defined below), all of the Optionee's Options shall automatically terminate and cease to be exercisable at the time of such termination and all Options exercised after the first event constituting Cause may be rescinded by the Administrator. "Cause" means breach of any provision of a Key Employment Agreement, Secure Computing Corporation Employment Agreement or any other agreement between the Company or any of its Affiliates and an Optionee, employment-related dishonesty, fraud, misconduct or disclosure or misuse of confidential information, or other employment-related conduct that is likely to cause significant injury to the Company, an Affiliate, or any of their respective employees, officers or directors (including, without limitation, commission of a felony or similar offense), in each case as determined by the Administrator. "Cause" shall not require that a civil judgment or criminal conviction have been entered against or guilty plea shall have been made by the Optionee regarding any of the matters referred to in the previous sentence. Accordingly, the Administrator shall be entitled to determine "Cause" based on the Administrator's good faith belief. If the Optionee is criminally charged with a felony or similar offense, that shall be a sufficient, but not a necessary, basis for such a belief.

9. Stock Grants and Stock Unit Awards. Each Stock Award Agreement reflecting the issuance of a Stock Grant or Stock Unit shall be in such form and shall contain such terms and conditions as the Administrator shall deem appropriate. The terms and conditions of such agreements may change from time to time, and the terms and conditions of separate agreements need not be identical, but each such agreement shall include (through incorporation of provisions hereof by reference in the agreement or otherwise) the substance of each of the following provisions:

9.1 Consideration. A Stock Grant or Stock Unit may be awarded in consideration for such property or services as is permitted under Applicable Law, including for past services actually rendered to the Company or an Affiliate for its benefit.

9.2 Vesting. Shares of Common Stock awarded under an agreement reflecting a Stock Grant and a Stock Unit award may, but need not, be subject to a share repurchase option, forfeiture restriction or other conditions in favor of the Company in accordance with a vesting or lapse schedule to be determined by the Administrator.

9.3 Termination. In the event of a Participant's Termination, the Company may reacquire any or all of the Shares held by the Participant which have not vested or which are otherwise subject to forfeiture or other conditions as of the date of Termination under the terms of the Stock Award Agreement.

9.4 Transferability. Except as determined by the Board and reflected in the Stock Award Agreement, no rights to acquire Shares under a Stock Grant or a Stock Unit shall be assignable or otherwise transferable by the Participant except by will or by the laws of descent and distribution.

10. Stock Appreciation Rights

10.1 In General. Stock Appreciation Rights may be granted either alone, in addition to, or in tandem with other Awards granted under the Plan. The Administrator may grant Stock Appreciation Rights to eligible Participants subject to terms and conditions not inconsistent with this Plan and determined by the Administrator. The specific terms and conditions applicable to the Participant shall be provided for in the Stock Award Agreement. Stock Appreciation Rights shall be exercisable, in whole or in part, at such times as the Administrator shall specify in the Stock Award Agreement.



10.2 Exercise of Stock Appreciation Right. Upon the exercise of a Stock Appreciation Right, in whole or in part, the Participant shall be entitled to a payment in an amount equal to the excess of the Fair Market Value on the date of exercise of a fixed number of Shares covered by the exercised portion of the Stock Appreciation Right, over the Fair Market Value on the grant date of the Shares covered by the exercised portion of the Stock Appreciation Right, such other amount calculated with respect to Shares subject to the award as the Administrator may determine). The amount due to the Participant upon the exercise of a Stock Appreciation Right shall be paid in such form of consideration as determined by the Administrator and may be in cash, Shares or a combination thereof, over the period or periods specified in the Stock Award Agreement. A Stock Award Agreement may place limits on the amount that may be paid over any specified period or periods upon the exercise of a Stock Appreciation Right, on an aggregate basis or as to any Participant. A Stock Appreciation Right shall be considered exercised when the Company receives written notice of exercise in accordance with the terms of the Stock Award Agreement from the person entitled to exercise the Stock Appreciation Right.

10.3 Transferability. Except as determined by the Board and reflected in the Stock Award Agreement, no Stock Appreciation Rights shall be assignable or otherwise transferable by the Participant except by will or by the laws of descent and distribution.

11. Certain Transactions and Events

11.1 In General. Except as provided in this Section 11, no change in the capital structure of the Company, merger, sale, or other disposition of assets or a subsidiary, change of control, issuance by the Company of shares of any class of securities convertible into shares of any class, conversion of securities, or other transaction or event shall require or be the occasion for any adjustments of the type described in this Section 11.

11.2 **Changes in Capital Structure**. In the event of any stock split, reverse stock split, recapitalization, combination or reclassification of stock, stock dividend, spin-off, or similar change to the capital structure of the Company (not including a Fundamental Transaction or an Event), the Board shall make whatever adjustments it concludes are appropriate to: (a) the number and type of Awards that may be granted under this Plan, (b) the number and type of Awards that may be granted to any individual under this Plan, including under Section 5.2 of this Plan, (c) the Option Price or Stock Award price, if any, and number and class of securities issuable under each outstanding Award, and (d) the repurchase price of any securities substituted for Award Shares that are subject to repurchase rights. The specific adjustments shall be determined by the Board in its sole and absolute discretion. Unless the Board specifies otherwise, any securities issuable as a result of any such adjustment shall be rounded to the next lower whole security.

11.3 Fundamental Transactions. If the Company merges with another entity in a transaction in which the Company is not the surviving entity or if, as a result of any other transaction or event, other securities are substituted for the Shares or Shares may no longer be issued (each a "Fundamental Transaction"), then, notwithstanding any other provision of this Plan, the Board shall do one or more of the following contingent on the closing or completion of the Fundamental Transaction: (a) arrange for the substitution in exchange for Awards of awards on equity securities other than Shares (including, if appropriate, equity securities of an entity other than the Company) (an "assumption" of Options) on such terms and conditions as the Board determines are appropriate, (b) accelerate the vesting and/or exercisability and termination of any restrictions of outstanding Awards, in whole or in part, so that

Awards can be exercised before or otherwise in connection with the closing or completion of a Fundamental Transaction or event but then terminate, (c) cancel or arrange for the cancellation of Awards in exchange for cash payments to Participants, and (d) either arrange for any repurchase rights of the Company with respect to Award Shares to apply to the securities issued in substitution for Shares or terminate repurchase rights on Award Shares. The Board need not adopt the same rules for each Award or each Participant.

11.4 Events. The majority of the "Incumbent Board" (as defined below) may also, but need not, specify that other transactions or events constitute an "*Event*," as set forth below. In connection with an Event, notwithstanding any other provision of this Plan, the Incumbent Board may take any one or more of the actions described in Section 11.3. In addition, the Incumbent Board may extend the date for the exercise of Options (but not beyond their original Expiration Date). The Incumbent Board need not adopt the same rules for each Award or each Participant. Examples of transactions or events that the Incumbent Board may treat as an Event are:

(1) The acquisition by any individual, entity or group (within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act) of beneficial ownership (within the meaning of Exchange Act Rule 13d-3) of 20% (except for acquisitions by any individual, entity or group that, prior to the Effective Date, owns 20% or more of any class of capital stock of the Company) or more of either (i) the then outstanding shares of common stock of the Company (the "*Outstanding Company Common Stock*") or (ii) the combined voting power of the then outstanding voting securities of the Company entitled to vote generally in the election of the Board (the "*Outstanding Company Voting Securities*"); provided, however, that the following acquisitions shall not constitute an Event:

(A) any acquisition of voting securities of the Company directly from the Company,

(B) any acquisition of voting securities of the Company by the Company or any of its wholly owned "Subsidiaries" (as defined in Section 424 of the Code),

(C) any acquisition of voting securities of the Company by any employee benefit plan (or related trust) sponsored or maintained by the Company or any of its Subsidiaries, or

(D) any acquisition by any corporation with respect to which, immediately following such acquisition, more than 60% of respectively, the then outstanding shares of common stock of such corporation and the combined voting power of the then outstanding voting securities of such corporation entitled to vote generally in the election of directors is then beneficially owned, directly or indirectly, by all or substantially all of the individuals and entities who were the beneficial owners, respectively, of the Outstanding Company Common Stock and Outstanding Company Voting Securities immediately prior to such acquisition in substantially the same proportions as was their ownership, immediately prior to such acquisition, of the Outstanding Company Common Stock and Outstanding Company Voting Securities, as the case may be;

(2) Individuals who, as of the Effective Date, constitute the Board (the "*Incumbent Board*") cease for any reason to constitute at least a majority of the Board; provided, however, that any individual becoming a Company Director subsequent to the Effective Date whose election, or nomination for election by the Company's stockholders, was approved by a vote of at least a majority of the Company Directors then comprising the Incumbent Board shall be considered a member of the Incumbent Board, but excluding, for this purpose, any such individual whose initial assumption of office occurs as a result of an actual or threatened election contest which was (or, if threatened, would have been) subject to Exchange Act Rule 14a-11;

(3) Approval by the stockholders of the Company of a reorganization, merger, consolidation or statutory exchange of Outstanding Company Voting Securities, unless immediately following such reorganization, merger, consolidation or exchange, all or substantially all of the individuals and entities who were the beneficial owners, respectively, of the Outstanding Company Common Stock and Outstanding Company Voting Securities immediately prior to such reorganization, merger, consolidation or exchange beneficially own, directly or indirectly, more than 60% of, respectively, the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors, as the case may be, of the corporation resulting from such reorganization, merger, consolidation or exchange in substantially the same proportions as was their ownership, immediately prior to such reorganization, merger, consolidation or exchange, of the Outstanding Company Voting Securities, as the case may be; or

(4) Approval by the stockholders of the Company of (i) a complete liquidation or dissolution of the Company or (ii) the sale or other disposition of all or substantially all of the assets of the Company, other than to a corporation with respect to which, immediately following such sale or other disposition, more than 60% of, respectively, the then outstanding shares of common stock of such corporation and the combined voting power of the then outstanding voting securities of such corporation entitled to vote generally in the election of directors is then beneficially owned, directly or indirectly, by all or substantially all of the individuals and entities who were the beneficial owners, respectively, of the Outstanding Company Common Stock and Outstanding Company Voting Securities immediately prior to such sale or other disposition in substantially the same proportion as was their ownership, immediately prior to such sale or other disposition, of the Outstanding Company Common Stock and Outstanding Company Voting Securities, as the case may be.

Notwithstanding the above, an Event shall not be deemed to occur with respect to a Participant if the acquisition of the 20% or greater interest referred to in paragraph (1) is by a group, acting in concert, that includes that Participant or if at least 40% of the then outstanding common stock or combined voting power of the then outstanding voting securities (or voting equity interests) of the surviving corporation or of any corporation (or other entity) acquiring all or substantially all of the assets of the Company shall be beneficially owned, directly or indirectly, immediately after a reorganization, merger, consolidation, statutory share exchange or sale or other disposition of assets referred to in paragraphs (3) or (4) by a group, acting in concert, that includes that Participant.

11.5 **Divestiture**. If the Company or an Affiliate sells or otherwise transfers equity securities of an Affiliate to a person or entity other than the Company or an Affiliate, or leases, exchanges or transfers all or any portion of its assets to such a person or entity, then the Board, in its sole and absolute discretion, may specify that such transaction or event constitutes a "*Divestiture*". In connection with a Divestiture, notwithstanding any other provision of this Plan, the Board may take one or more of the actions described in Section 11.3 with respect to Awards or Award Shares held by, for example, Employees, Directors or Consultants for whom that transaction or event results in a Termination. The Board need not adopt the same rules for each Award or each Participant.

11.6 **Dissolution**. If the Company adopts a plan of dissolution, the Board may, in its sole and absolute discretion, cause Awards to be fully vested and exercisable (but not after their Expiration Date) before the dissolution is completed but contingent on its completion and may cause the Company's repurchase rights on Award Shares to lapse upon completion of the dissolution. To the extent not exercised or settled before the earlier of the completion of the dissolution or their Expiration Date, Awards shall terminate just before the dissolution is completed. The Board need not adopt the same rules for each Award or each Participant.

11.7 **Cut-Back to Preserve Benefits**. If the Administrator determines that the net after-tax amount to be realized by any Participant, taking into account any accelerated vesting, termination of repurchase rights, or cash payments to that Participant in connection with any transaction or event addressed in this Section 11, would be greater if one or more of those steps were not taken with respect to that Participant's Awards or Award Shares, then and to that extent one or more of those steps shall not be taken.

11.8 **Substitute Awards.** The Board may cause the Company to grant Substitute Awards in connection with the acquisition by the Company or an Affiliate of equity securities of any entity (including by merger) or all or a portion of the assets of any entity. Any such substitution shall be effective when the acquisition closes. Unless and to the extent specified otherwise by the Board, Substitute Awards shall have the same terms and conditions as the options they replace, except that (subject to Section 11) Substitute Awards shall be Awards to purchase Shares rather than equity securities of the granting entity and shall have an exercise or purchase price that, as determined by the Board in its sole and absolute discretion, properly reflects the substitution.

12. Withholding and Tax Reporting

12.1 Tax Withholding Option

(a) *General*. Whenever Award Shares are granted, vest, transferred, purchased, issued or become free of restrictions, the Company may require the Participant to remit to the Company an amount sufficient to satisfy any applicable tax withholding requirement, whether the related tax is imposed on the Participant or the Company. The Company shall have no obligation to deliver Award Shares or release Award Shares from an escrow or permit a transfer of Award Shares until the Participant has satisfied those tax withholding obligations. Whenever payment in satisfaction of Awards is made in cash, the payment will be reduced by an amount sufficient to satisfy all tax withholding requirements.

(b) *Method of Payment*. The Participant shall pay any required withholding using the forms of consideration described in Section 6.5(b), except that, in the discretion of the Administrator, the Company may also permit the Participant to use any of the forms of payment described in Section 6.5(c). The Administrator may also permit Award Shares to be withheld to pay required withholding. If the Administrator permits Award Shares to be withheld, the Fair Market Value of the Award Shares withheld shall not exceed the amount determined by the applicable minimum statutory withholding rates and shall be determined as of the date that the amount of tax to be withheld or tendered for this purpose is to be determined.

12.2 **Reporting of Dispositions.** Any holder of Option Shares acquired under an Incentive Stock Option shall promptly notify the Administrator in writing of the sale or other disposition of any of those Option Shares if the disposition occurs during: (a) the longer of two years after the Grant Date of the Incentive Stock Option and one year after the date the Incentive Stock Option was exercised, or (b) such other period as the Administrator has established.

13. Consulting or Employment Relationship

Nothing in this Plan or in any Award Agreement, and no Award or the fact that Award Shares remain subject to repurchase rights, shall: (a) interfere with or limit the right of the Company or any Affiliate to terminate the employment or consultancy of any Participant at any time, whether with or without cause or reason, and with or without the payment of severance or any other compensation or payment, or (b) interfere with the application of any provision in any of the Company's or any Affiliate's charter documents or Applicable Law relating to the election, appointment, term of office, or removal of a Director.

14. Section 162(m) Compliance

Any Stock Award (other than an Option or any other Stock Award having a purchase price equal to 100% of the Fair Market Value on the date such Award is made) that is intended as "qualified performance-based compensation" within the meaning of Section 162(m) of the Code must vest or become exercisable contingent on the achievement of one or more Qualifying Performance Criteria. Notwithstanding anything to the contrary herein, the Committee shall have the discretion to determine the time and manner of compliance with Section 162(m) of the Code as required under applicable regulations and to conform the procedures related to the Award to the requirements of Section 162(m) of the Code and may reduce the number of Shares granted or amount of cash or other property to which a Participant may otherwise have been entitled with respect to an Award designed to qualify as performance-based compensation under Section 162(m) of the Code.

15. Compliance with Law

The grant of Awards and the issuance and subsequent transfer of Award Shares shall be subject to compliance with all Applicable Laws, including all applicable securities laws. Awards may not be exercised or settled, and Award Shares may not be transferred, in violation of Applicable Law. Thus, for example, Options may not be exercised unless: (a) a registration statement under the Securities Act is then in effect with respect to the related Option Shares, or (b) in the opinion of legal counsel to the Company, those Option Shares may be issued in accordance with an applicable exemption from the registration requirements of the Securities Act and any other applicable securities laws. The Company is under no requirement to register any Award Shares, and the failure or inability of the Company to obtain from any regulatory body the authority considered by the Company's legal counsel to be necessary or useful for the lawful issuance of any Award Shares or their subsequent transfer shall relieve the Company of any liability for failing to issue those Award Shares or permitting their transfer. As a condition to the exercise or settlement of any Award or the transfer of any Award Shares, the Company may require the Participant to satisfy any requirements or qualifications that may be necessary or appropriate to comply with or evidence compliance with any Applicable Law.

16. Amendment or Termination of this Plan or Outstanding Awards

16.1 Amendment and Termination. The Board may at any time amend, suspend, or terminate this Plan.

16.2 **Stockholder Approval**. The Company shall obtain the approval of the Company's stockholders for any amendment to this Plan if stockholder approval is necessary or desirable to comply with any Applicable Laws or with the requirements applicable to the grant of Options intended to be Incentive Stock Options. The Board may also, but need not, require that the Company's stockholders approve any other amendments to this Plan. In addition, unless approved by the stockholders of the Company, no amendment shall be made that would result in a repricing of Options by (x) reducing the exercise price of outstanding Options or (y) canceling an outstanding Option held by a Participant and regranting to the Participant a new Option with a lower exercise price, in either case other pursuant to Section 11 of the Plan.

16.3 Effect. No amendment, suspension, or termination of this Plan, and no modification of any Award even in the absence of an amendment, suspension, or termination of this Plan, shall impair any existing contractual rights of any Participant unless the affected Participant consents to the amendment, suspension, termination, or modification. However, no such consent shall be required

if the Administrator determines in its sole and absolute discretion that the amendment, suspension, termination, or modification: (a) is required or advisable in order for the Company, the Plan, or the Award to satisfy Applicable Law, to meet the requirements of any accounting standard or to avoid any adverse accounting treatment, or (b) in connection with any transaction or event described in Section 11, is in the best interests of the Company or its stockholders. The Administrator may, but need not, take the tax consequences to affected Participants into consideration in acting under the preceding sentence. Termination of this Plan shall not affect the Administrator's ability to exercise the powers granted to it under this Plan with respect to Awards granted before the termination or Award Shares issued under such Awards even if those Award Shares are issued after the termination.

17. Reserved Rights

17.1 Nonexclusivity of this Plan. This Plan shall not limit the power of the Company or any Affiliate to adopt other incentive arrangements including, for example, the grant or issuance of stock options, stock, or other equity-based rights under other plans or independently of any plan.

17.2 Unfunded Plan. This Plan shall be unfunded. Although bookkeeping accounts may be established with respect to Participants, any such accounts will be used merely as a convenience. The Company shall not be required to segregate any assets on account of this Plan, the grant of Awards, or the issuance of Award Shares. The Company and the Administrator shall not be deemed to be a trustee of stock or cash to be awarded under this Plan. Any obligations of the Company to any Participant shall be based solely upon contracts entered into under this Plan, such as Award Agreements. No such obligation shall be deemed to be secured by any pledge or other encumbrance on any assets of the Company. Neither the Company nor the Administrator shall be required to give any security or bond for the performance of any such obligation.

18. Special Arrangements Regarding Award Shares

18.1 Escrows and Pledges. To enforce any restrictions on Award Shares including restrictions related to Reverse Vesting, the Administrator may require their holder to deposit the certificates representing Award Shares, with stock powers or other transfer instruments approved by the Administrator endorsed in blank, with the Company or an agent of the Company to hold in escrow until the restrictions have lapsed or terminated. The Administrator may also cause a legend or legends referencing the restrictions to be placed on the certificates.

18.2 Repurchase Rights

(a) *Reverse Vesting*. If an Option is subject to Reverse Vesting, the Company shall have the right, during the seven months after the Optionee's Termination, to repurchase any or all of the Option Shares that were unvested as of the date of that Termination, for a price equal to the lower of: (i) the Option Price for such Shares, minus the amount of any cash dividends paid or payable with respect to the Option Shares for which the record date precedes the repurchase, and (ii) the Fair Market Value of those Option Shares as of the date of the Termination. The repurchase price shall be paid in (i) cash, (ii) if the Option Shares were purchased in whole or in part for a promissory note, cancellation of indebtedness under that note, (iii) cancellation of any indebtedness owed by the Optionee to the Company or any Affiliate, or (iv) a combination of those means. The Company may assign this right of repurchase.

(b) *Procedure.* The Company or its assignee may choose to give the Optionee a written notice of exercise of its repurchase rights under this Section 18.2. However, the Company's failure to give such a notice shall not affect its rights to repurchase Option Shares. The Company must, however, tender the repurchase price during the period specified in this Section 18.2 for exercising its repurchase rights in order to exercise such rights.

19. Beneficiaries

A Participant may file a written designation of one or more beneficiaries who are to receive the Participant's rights under the Participant's Awards after the Participant's death. A Participant may change such a designation at any time by written notice. If a Participant designates a beneficiary may exercise the Participant's Awards after the Participant's death. If a Participant dies when the Participant has no living beneficiary designated under this Plan, the Company shall allow the executor or administrator of the Participant's estate to exercise the Award or, if there is none, the person entitled to exercise the Award under the Participant's will or the laws of descent and distribution. In any case, no Option may be exercised after its Expiration Date.

20. Miscellaneous

20.1 Governing Law. This Plan and all determinations made and actions taken under this Plan shall be governed by the substantive laws, but not the choice of law rules, of the State of Delaware.

20.2 Determination of Value. Fair Market Value shall be determined as follows:

(a) *Listed Stock*. If the Shares are traded on any established stock exchange or quoted on a national market system, Fair Market Value shall be the closing sales price for the Shares as quoted on that stock exchange or system for the date the value is to be determined (the "*Value Date*") as reported in *The Wall Street Journal* or a similar publication. If no sales are reported as having occurred on the Value Date, Fair Market Value shall be that closing sales price for the last preceding trading day on which sales of Shares are reported as having occurred as having occurred during the five trading days before the Value Date, Fair Market Value shall be the closing bid for Shares on the Value Date. If Shares are listed on multiple exchanges or systems, Fair Market Value shall be based on sales or bids on the primary exchange or system on which Shares are traded or quoted.

(b) *Stock Quoted by Securities Dealer*. If Shares are regularly quoted by a recognized securities dealer but selling prices are not reported on any established stock exchange or quoted on a national market system, Fair Market Value shall be the mean between the high bid and low asked prices on the Value Date. If no prices are quoted for the Value Date, Fair Market Value shall be the mean between the high bid and low asked prices on the last preceding trading day on which any bid and asked prices were quoted.

(c) *No Established Market*. If Shares are not traded on any established stock exchange or quoted on a national market system and are not quoted by a recognized securities dealer, the Administrator (following guidelines established by the Board or Committee) will determine Fair Market Value in good faith. The Administrator will consider the following factors, and any others it considers significant, in determining Fair Market Value: (i) the price at which other securities of the Company have been issued to purchasers other than Employees, Directors, or Consultants, (ii) the Company's net worth, prospective earning power, dividend-paying capacity, and non-operating assets, if any, and (iii) any other relevant factors, including the economic outlook for the Company and the Company's industry, the Company's position in that industry, the Company's goodwill and other intellectual property, and the values of securities of other businesses in the same industry.

20.3 Reservation of Shares. During the term of this Plan, the Company will at all times reserve and keep available such number of Shares as are still issuable under this Plan.

20.4 Electronic Communications. Any Award Agreement, notice of exercise of an Award, or other document required or permitted by this Plan may be delivered in writing or, to the extent determined by the Administrator, electronically. Signatures may also be electronic if permitted by the Administrator.

20.5 Notices. Unless the Administrator specifies otherwise, any notice to the Company under any Award Agreement or with respect to any Awards or Award Shares shall be in writing (or, if so authorized by Section 20.4, communicated electronically), shall be addressed to the Secretary of the Company, and shall only be effective when received by the Secretary of the Company.

CYBERGUARD CORPORATION

THIRD AMENDED AND RESTATED EMPLOYEE STOCK OPTION PLAN

Dated September 4, 1998, as amended through December 4, 2003.

1. Purpose

The purpose of the CyberGuard Corporation Employee Stock Option Plan (the "Plan") is to promote the long-term growth and performance of CyberGuard Corporation (the "Corporation") and its affiliates and to attract and retain outstanding individuals by awarding stock options, stock appreciation rights, performance awards, restricted stock and/or other stock-based awards.

2. Definitions

The following definitions are applicable to the Plan:

"Award" means the grant of Options, SARs, Performance Awards, Restricted Stock or other stock-based award under the Plan.

"Board" means the Board of Directors of the Corporation.

"Board Committee" means the Compensation Committee of the Board, or in lieu thereof, the Board may appoint any committee of at least two directors to administer the Plan.

"Change of Control" means any of the events set forth below; *provided, however*, that the Board Committee, in its sole discretion, may specify a more restrictive definition of Change of Control in any Award agreement and, in such event, the definition of Change of Control set forth in the Award agreement shall apply to the Award granted under such Award agreement:

(a) The acquisition in one or more transactions, other than from the Corporation, by any individual, entity or group (within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act) of beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of either (i) 20% of the Outstanding Corporation Common Stock or (ii) 20% of the Corporation Voting Securities; *provided, however*, that the following shall not constitute a Change of Control: any acquisition by (1) the Corporation or any of its Subsidiaries, any employee benefit plan (or related trust) sponsored or maintained by the Corporation or any of its Subsidiaries, or (2) any corporation with respect to which, following such acquisition, more than 80% of, respectively, the then outstanding shares of common stock of such corporation and the combined voting power of the then outstanding voting securities of such corporation entitled to vote generally in the election of directors is then beneficially owned, directly or indirectly, by all or substantially all of the individuals and entities who were the beneficial owners, respectively, of the Outstanding Corporation Common Stock and Corporation Voting Securities immediately prior to such acquisition in substantially the same proportion as their ownership, immediately prior to such acquisition, of the Outstanding Corporation Common Stock and Corporation Voting Securities, as the case may be; or

(b) Individuals who constitute the Incumbent Board cease for any reason to constitute in excess of three-fourths (3/4) of the Board; *provided, however*, that any individual becoming a director subsequent to September 4,1998 whose election or nomination for election by the Corporation was approved by a vote of at least a majority of the directors then comprising the Incumbent Board shall be considered as though such individual were a member of the Incumbent Board, but excluding, for this purpose, any such individual whose initial assumption of office is in connection with an actual or threatened election contest relating to the election of members of the Board; or

(c) Approval by the stockholders of the Corporation of a reorganization, merger or consolidation, unless, following such reorganization, merger or consolidation, all or substantially all of the individuals and entities who were the respective beneficial owners of the Outstanding Corporation Common Stock and Corporation Voting Securities immediately prior to such reorganization, merger or consolidation, following such reorganization, merger or consolidation beneficially own, directly or indirectly, more than 80% of, respectively, the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors, as the case may be, of the corporation resulting from such reorganization, merger or consolidation in substantially the same proportion as their ownership of the Outstanding Corporation Common Stock and Corporation Voting Securities immediately prior to such reorganization, merger of consolidation, as the case may be; or

(d) Approval by the stockholders of the Corporation of (i) a complete liquidation or dissolution of the Corporation or (ii) a sale or other disposition of 60% or more by value of the assets of the Corporation other than to a corporation with respect to which, following such sale or disposition, more than 80% of, respectively, the then outstanding shares of common stock and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors is then owned beneficially, directly or indirectly, by all or substantially all of the individuals and entities who were the beneficial owners, respectively, of the Outstanding Corporation Common Stock and Corporation Voting Securities, as the case may be, immediately prior to such sale or disposition.

"Code" means the Internal Revenue Code of 1986, as amended.

"Commission" means the Securities and Exchange Commission.

"Common Stock" means the common stock of the Corporation, \$.01 par value per share.

"Corporation Voting Securities" means the combined voting power of all outstanding voting securities of the Corporation entitled to vote generally in the election of directors for the Board.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Fair Market Value" means, on any date, the closing sale price of one share of Common Stock, as reported in the NASDAQ National Market System or any national securities exchange on which the Common Stock is then listed, as published in the Wall Street Journal or another newspaper of general circulation, as of such date or, if there were no sales reported as of such date, as of the last date preceding such date as of which a sale was reported. In the event that the Common Stock is not listed on the NASDAQ National Market System or a national securities exchange, Fair Market Value shall be determined in good faith by the Board Committee in its sole discretion.

"Grant Date" means the date on which an Option under Section 5.1 hereof or a SAR under Section 6.1 hereof is granted.

"Incumbent Board" means the Board as constituted as of September 4, 1998.

"Non-Statutory Stock Option" means an option to purchase shares of Common Stock which is not an Incentive Stock Option.

"Option" means any option to purchase shares of Common Stock granted under Sections 5.1 and 10.1 hereof.

"Option Price" means the purchase price of each share of Common Stock under an Option.

"Outstanding Corporation Common Stock" means, at any time, the issued and outstanding shares of Corporation Common Stock.

"Participant" means any employee or consultant of the Corporation and its affiliates, or members of the Board, designated by the Board Committee to receive an Award under the Plan.

"Performance Award" means an Award of shares of Common Stock granted under Section 7 hereof.

"Performance Period" means the period of time established by the Board Committee for achievement of certain objectives under Section 7.1 hereof.

"Restriction Period" means the period of time specified in a Performance Award agreement or a Restricted Stock agreement, as the case may be, between the Participant and the Corporation during which the following conditions remain in effect: (i) certain restrictions on the sale or other disposition of shares of Common Stock awarded under the Plan, and (ii) subject to the terms of the applicable agreement, a requirement of continued employment of the Participant in order to prevent forfeiture of the Award.

"Retirement" and "Termination upon Retirement" shall be deemed to have occurred at anytime when all of the following three events (a, b and c) have occurred:

(a) an employee: (i) is terminated without cause, or (ii) resigns his or her position with the Corporation, or (iii) resigns for reasons of disability; and

(b) such termination occurs on or after his or her sixtieth (60th) birthday; and

(c) either (i) employee has at least ten (10) years of service with the Corporation, or (ii) employee has been continuously in the employ of the Corporation since July 1, 1996.

This provision shall apply only to employees, not consultants or others (that is, consultants or others under no circumstances may be deemed to have "retired" under the Plan.)

"Stock Appreciation Rights" or "SARs" means the right to receive a cash payment from the Corporation equal to the excess of the Fair Market Value of a stated number of shares of Common Stock at the exercise date over a fixed price for such shares.

"Subsidiary" means any corporation (other than the Corporation) in an unbroken chain of corporations beginning with the Corporation if each of the corporations (other than the last corporation in the chain) owns stock possessing 50% or more of the total combined voting power of all classes of stock in one of the other corporations in the chain.

3. Shares Subject to Plan

3.1 Shares Reserved under the Plan. Subject to adjustment as provided in Section 3.2, no more than ten million (10,000,000) shares of Common Stock shall be cumulatively available for the grant of Stock Options under the Plan. Any Common Stock issued by the Corporation through the assumption or substitution of outstanding grants from an acquired corporation or entity shall not reduce the shares available for grants under the Plan. Shares of Common Stock to be issued pursuant to the Plan may be authorized and unissued shares, treasury shares, or any combination thereof. Subject to Section 6.2 hereof, if any shares of Common Stock subject to an Award hereunder are forfeited or any such Award otherwise terminates without the issuance of such shares of Common Stock to a Participant, or if any shares of Common Stock are surrendered by a Participant in full or partial payment of the Option Price of an Option, such shares, to the extent of any such forfeiture, termination or surrender, shall again be available for grant under the Plan; *provided however*, that shares of Common Stock surrendered by a Participant who is subject to Section 16 of the Exchange Act, or any successor thereto, in payment of the Option Price, shall be available for grant under the Plan only to Participants not subject to such Section.

3.2 Adjustments. Subject to Section 12 hereof, the aggregate number of shares of Common Stock which may be awarded under the Plan and outstanding Awards shall be adjusted by the Board Committee to reflect a change in the capitalization of the Corporation, including but not limited to, a stock dividend or split, recapitalization, reorganization, merger, consolidation, combination, exchange of shares, spin-off, spin-out or other distribution of assets to shareholders.

4. Administration of Plan

4.1 Administration by the Board Committee. The Plan shall be administered by the Board Committee; provided however, the Board Committee may delegate some or all of its authority and responsibility under the Plan with respect to Awards to Participants who are not subject to Section 16 of the Exchange Act to the Chief Executive Officer of the Corporation. In addition to the Board Committee, the Board also has discretion to issue awards under the Plan and references herein to the Board Committee taking action with respect to the Plan should be considered as also referring to the Board, where appropriate. The Board Committee shall have authority to interpret the Plan, to establish, amend, and rescind any rules and regulations relating to the Plan, to prescribe the form of any agreement or instrument executed in connection herewith, and to make all other determinations necessary or advisable for the administration of the Plan. All such interpretations, rules, regulations and determinations shall be conclusive and binding on all persons and for all purposes. In addition, the Board Committee shall have authority, without amending the Plan, to grant Awards hereunder to Participants who are foreign nationals or employed outside the United States or both, on terms and conditions different from those specified herein as may, in the sole judgment and discretion of the Board Committee, be necessary or desirable to further the purpose of the Plan.

4.2 Designation of Participants. Participants shall be selected, from time to time, by the Board Committee, from those employees and consultants of the Corporation and its affiliates, and from those members of the Board, who, in the opinion of the Board Committee, have the capacity to contribute materially to the continued growth and successful performance of the Corporation.

5. Stock Options

5.1 *Grants.* Options may be granted, from time to time, to such employees and consultants of the Corporation and its affiliates, and to such members of the Board, as may be selected by the Board Committee. The Option Price shall be determined by the Board Committee effective on the Grant Date; *provided however*, that such price shall not be less than fifty (50%) of the Fair Market Value of a share of Common Stock on the Grant Date. The number of shares of Common Stock subject to each option granted to each Participant, the terms of each option, and any other terms and conditions of an Option granted hereunder shall be determined by the Board Committee, in its sole discretion, effective on the Grant Date; *provided however*, that no option shall be exercisable any later than ten (10) years from the Grant Date. Each option shall be evidenced by an Option agreement between the Participant and the Corporation which shall specify the Option Price, the term of the Option, the number of shares of Common Stock to which the Option pertains, the conditions upon which the Option becomes exercisable and such other terms and conditions as the Board Committee shall determine. All Options issued under this Plan shall become immediately exercisable upon occurrence of a Change of Control.

5.2 Payment of Option Price. No shares of Common Stock shall be issued upon exercise of an Option until full payment of the Option Price therefor by the Participant. Upon exercise, the Option Price may be paid in cash, in shares of Common Stock having a Fair Market Value equal to the Option Price, or in any combination thereof.

5.3 *Rights as Shareholders*. Participants shall not have any of the rights of a shareholder with respect to any shares subject to an Option until such shares have been issued upon the proper exercise of such Option.

5.4 Transferability of Options. Options granted under the Plan may not be sold, transferred, pledged, assigned, hypothecated or otherwise disposed of except by will or by the laws of descent and distribution. All

Options granted to a Participant under the Plan shall be exercisable during the lifetime of such Participant only by such Participant, his agent, guardian or attorney-infact.

5.5 Termination of Employment. If a Participant ceases to be an employee or a consultant of either the Corporation or of any of its affiliates, the Options granted hereunder shall be exercisable in accordance with the Option agreement between the Participant and the Corporation.

6. Stock Appreciation Rights

6.1 *Grants.* Stock Appreciation Rights may be granted, from time to time, to such employees and consultants of the Corporation and its affiliates, and to such members of the Board, as may be selected by the Board Committee. SARs may be granted at the discretion of the Board Committee either (i) in connection with an Option or (ii) independent of an Option. The price from which appreciation shall be computed shall be established by the Board Committee at the Grant Date; *provided however*, that such price shall not be less than fifty percent (50%) of the Fair Market Value of the number of shares of Common Stock subject of the grant on the Grant Date. In the event the SAR is granted in connection with an Option, the fixed price from which appreciation shall be computed shall be the Option Price. Each grant of a SAR shall be evidenced by a Stock Appreciation Rights agreement between the Participant and the Corporation which shall specify the type of SAR granted, the number of SARs, the conditions upon which the SARs vest and such other terms and conditions as the Board Committee shall determine.

6.2 *Exercise of SARs.* SARs may be exercised upon such terms and conditions as the Board Committee shall determine; *provided however*, that SARs granted in connection with Options may be exercised only to the extent the related Options are then exercisable. Notwithstanding Section 3.1 hereof, upon exercise of a SAR granted in connection with an Option as to all or some of the shares subject of such Award, the related Option shall be automatically canceled to the extent of the number of shares subject of the exercise, and such shares shall no longer be available for grant hereunder. Conversely, if the related Option is exercised as to some or all of the shares subject of such Award, the related SAR shall automatically be canceled to the extent of the number of shares of the exercise, and such shares shall no longer be available for grant hereunder. Some of the exercise, and such shares shall no longer be available for grant hereunder. All SARs shall become immediately exercisable upon occurrence of a Change of Control.

6.3 Payment of Exercise. Upon exercise of a SAR, the holder shall be paid in cash the excess of the Fair Market Value of the number of shares subject of the exercise over the fixed price, which in the case of a SAR granted in connection with an Option shall be the Option Price for such, shares.

6.4 Rights of Shareholders. Participants shall not have any of the rights of a shareholder with respect to any Options granted in connection with a SAR until shares have been issued upon the proper exercise of an Option.

6.5 *Transferability of SARs.* SARs granted under the Plan may not be sold, transferred, pledged, assigned, hypothecated or otherwise disposed of except by will or by the laws of descent and distribution. All SARs granted to a Participant under the Plan shall be exercisable during the lifetime of such Participant only by such Participant, his agent, guardian, or attorney-in-fact.

6.6 Termination of Employment. If a Participant ceases to be an employee or a consultant of either the Corporation or of any of its affiliates, SARs granted hereunder shall be exercisable in accordance with the Stock Appreciation Rights agreement between the Participant and the Corporation.

7. Performance Awards

7.1 *Awards*. Awards of shares of Common Stock may be made, from time to time, to such employees and consultants of the Corporation and its affiliates, and to such members of the Board, as may be selected by the Board Committee. The release of such shares to the Participant at the lapse of restriction on the sale or transfer of shares subject to such Awards shall be contingent upon (i) achievement of such corporate, sector, division or other objectives during the Performance Period as shall be established by the Board Committee and (ii) the

expiration of the Restriction Period. Except as provided in the Performance Award agreement between the Participant and the Corporation, shares subject to such Awards under this Section 7.1 shall be released to the Participant only after the expiration of the relevant Restriction Period. Each Award under this Section 7.1 shall be evidenced by a Performance Award agreement between the Participant and the Corporation which shall specify the applicable performance objectives, the Performance Period, the Restriction Period, any forfeiture conditions and such other terms and conditions as the Board Committee shall determine. All restrictions applicable to Performance Awards shall immediately expire upon occurrence of a Change of Control.

7.2 Stock Certificates. Upon an Award of shares of Common Stock under Section 7.1 of the Plan, the Corporation shall issue a certificate registered in the name of the Participant bearing the following legend and any other legend required by any federal or state securities laws or by the Florida Business Corporation Act:

"The sale or other transfer of the shares of stock represented by this certificate is subject to certain restrictions set forth in the CyberGuard Corporation Employee Stock Option Plan, administrative rules adopted pursuant to such Plan and a Performance Award agreement between the registered owner and CyberGuard Corporation. A copy of the Plan, such rules and such agreement may be obtained from the Secretary of CyberGuard Corporation.

Unless otherwise provided in the Performance Award agreement between the Participant and the Corporation, such certificates shall be retained by the Corporation until the expiration of the Restriction Period. Upon the expiration of the Restriction Period, the Corporation shall (i) cause the removal of the legend from the certificates for such shares as to which a Participant is entitled in accordance with the Performance Award agreement between the Participant and the Corporation and (ii) release such shares to the custody of the Participant.

7.3 *Rights as Shareholders.* Subject to the provisions of the Performance Award agreement between the Participant and the Corporation, during the Performance Period, Participants may exercise full voting rights with respect to all shares awarded thereto under Section 7.1 hereof and shall be entitled to receive dividends and other distributions paid with respect to those shares. During the period between the completion of the Performance Period and the expiration of the Restriction Period, Participants may exercise full voting rights and shall be entitled to receive dividends and other distributions only as to the number of shares determined in accordance with the Performance Award agreement between the Participant and the Corporation.

7.4 Transferability of Shares. Certificates evidencing the shares of Common Stock awarded under the Plan shall not be sold, exchanged, assigned, transferred, pledged, hypothecated or otherwise disposed of until the expiration of the Restriction Period.

7.5 Termination of Employment. If a Participant ceases to be an employee or a consultant of either the Corporation or of one of its affiliates, the number of shares subject of the Award, if any, to which the Participant shall be entitled shall be determined in accordance with the Performance Award agreement between the Participant and the Corporation.

7.6 Transfer of Employment. If a Participant transfers employment from one business unit of the Corporation or any of its affiliates to another business unit during a Performance Period, such Participant shall be eligible to receive such number of shares of Common Stock as the Board Committee may determine based upon such factors as the Board Committee in its sole discretion may deem appropriate.

8. Restricted Stock Awards

8.1 Awards. Awards of shares of Common Stock subject to such restrictions as to vesting and otherwise as the Board Committee shall determine, may be made, from time to time, to employees and consultants of the Corporation and its affiliates, and to such members of the Board, as may be selected by the Board Committee. The Board Committee may in its sole discretion at the time of the Award or at any time thereafter provide for the early vesting of such Award prior to the expiration of the Restriction Period. Each Award under this Section 8.1

shall be evidenced by a Restricted Stock agreement between the Participant and the Corporation which shall specify the vesting schedule, any rights of acceleration, any forfeiture conditions, and such other terms and conditions as the Board Committee shall determine. All restrictions applicable to Restricted Stock Awards shall immediately expire upon occurrence of a Change of Control.

8.2 Stock Certificates. Upon an Award of shares of Common Stock under Section 8.1 of the Plan, the Corporation shall issue a certificate registered in the name of the Participant bearing the following legend and any other legend required by any federal or state securities laws or by the Florida Business Corporation Act.

"The sale or other transfer of the shares of stock represented by this certificate is subject to certain restrictions set forth in the CyberGuard Corporation Employee Stock Option Plan, administrative rules adopted pursuant to such Plan and a Restricted Stock agreement between the registered owner and CyberGuard Corporation. A copy of the Plan, such rules and such agreement may be obtained form the Secretary of CyberGuard Corporation."

Unless otherwise provided in the Restricted Stock agreement between the Participant and the Corporation, such certificates shall be retained by the Corporation until the expiration of the Restriction Period. Upon the expiration of the Restriction Period, the Corporation shall (i) cause the removal of the legend from the certificates for such shares as to which a Participant is entitled in accordance with the Restricted Stock agreement between the Participant and the Corporation and (ii) release such shares to the custody of the Participant.

8.3 *Rights as Shareholders.* During the Restriction Period, Participants may exercise full voting rights with respect to all shares awarded thereto under Section 8.1 hereof and shall be entitled to receive dividends and other distributions paid with respect to those shares.

8.4 Transferability of Shares. Certificates evidencing the shares of Common Stock awarded under the Plan shall not be sold, exchanged, assigned, transferred, pledged, hypothecated or otherwise disposed of until the expiration of the Restriction Period.

8.5 Termination of Employment. If a Participant ceases to be an employee or a consultant of either the Corporation or of any of its affiliates, the number of shares subject of the Award, if any, to which the Participant shall be entitled shall be determined in accordance with the Restricted Stock agreement between the Participant and the Corporation. All remaining shares as to which restrictions apply at the date of termination of employment shall be forfeited subject to such exceptions, if any, authorized by the Board Committee.

9. Other Stock-Based Awards

Awards of shares of Common Stock and other awards that are valued in whole or in part by reference to, or are otherwise based on, Common Stock, may be made, from time to time, to employees and consultants of the Corporation and its affiliates, and to such members of the Board, as may be selected by the Board Committee. Such Awards may be made alone or in addition to or in connection with any other Award hereunder. The Board Committee may in its sole discretion determine the terms and conditions of any such Award. Each such Award shall be evidenced by an agreement between the Participant and the Corporation which shall specify the number of shares of Common Stock subject of the Award, any consideration therefor, any vesting or performance requirements and such other terms and conditions as the Board Committee shall determine. All such Awards shall be fully vested and payable immediately upon the occurrence of a Change of Control.

10. (Omitted Intentionally)

11. Amendment or Termination of Plan

Until such time as a Change of Control shall have occurred, the Board or the Board Committee may amend, suspend or terminate the Plan or any part thereof from time to time, provided that no change may be made which would impair the rights of a Participant to whom shares of Common Stock have theretofore been awarded without the consent of said Participant.
12. Miscellaneous

12.1 Rights of Participants. Nothing in the Plan shall interfere with or limit in any way the right of the Corporation or any affiliate to terminate any Participant's employment or consultancy at any time, nor confer upon any Participant any right to continued employment or consulting status with the Corporation or any affiliate.

12.2 Tax Withholding. The Corporation shall have the authority to withhold, or to require a Participant to remit to the Corporation, prior to issuance or delivery of any shares or cash hereunder, an amount sufficient to satisfy federal, state and a local tax withholding requirements associated with any Award. In addition, the Corporation may, in its sole discretion, permit a Participant to satisfy any tax withholding requirements, in whole or in part, by (i) delivering to the Corporation shares of Common Stock held by such Participant having a Fair Market Value equal to the amount of the tax or (ii) directing the Corporation to retain shares of Common stock otherwise issuable to the Participant under the Plan.

12.3 Status of Awards. Awards hereunder shall not be deemed compensation for purposes of computing benefits under any retirement plan of the Corporation or affiliate and shall not affect any benefits under any other benefit plan now or hereafter in effect under which the availability or amount of benefits is related to the level of compensation.

12.4 Waiver of Restrictions. The Board Committee may, in its sole discretion, based on such factors as the Board Committee may deem appropriate, waive in whole or in part, any remaining restrictions or vesting requirements in connection with any Award hereunder.

12.5 Delegation to Management. The Board Committee may delegate to one or more officers of the Corporation or a committee of officers the right to grant Awards hereunder to employees and consultants who are not officers or directors of the Corporation and to cancel or suspend Awards to employees and consultants who are not officers or directors of the Corporation.

12.6 Adjustment of Awards. Subject to Section 11, the Board Committee shall be authorized to make adjustments in performance award criteria or in the terms and conditions of other Awards in recognition of unusual or nonrecurring events affecting the Corporation or its financial statements or changes in applicable laws, regulations or accounting principles; provided however, that no such adjustment shall impair the rights of any Participant without his or her consent. The Board Committee may also make Awards hereunder in replacement of, or as alternatives to, Awards previously granted to Participants, including without limitation, previously granted Options having higher Option Prices and grants or rights under any other plan of the Corporation or of any acquired entity. The Board Committee may correct any defect, supply any omission or reconcile any inconsistency in the Plan or any Award in the manner and to the extent it shall deem desirable to carry it into effect. In the event the Corporation shall assume outstanding employee benefit awards or the right or obligation to make future such awards in connection with the acquisition of another corporation or business entity, the Board Committee may, in its discretion, make such adjustments in the terms of Awards under the Plan as it shall deem appropriate.

12.7 Consideration for Awards. Except as otherwise required in any applicable agreement or by the terms of the Plan, Participants under the Plan shall not be required to make any payment or provide consideration for an Award other than the rendering of services.

12.8 Effective Date and Term of Plan. The Plan shall be effective as of September 4, 1998. Unless terminated under the provisions of Section 11 hereof, the Plan shall continue in effect until terminated by the Board.

A-8

SAFEBOOT OPTION PLAN 2006

(as amended May 13, 2008)

Article 1 — Definitions

1.1 As used in this Plan, except where inconsistent with the subject or context, the words and expressions below shall have the following meanings:

Acceptance Form	The letter with which an Employee accepted all Former Options granted to him, substantially in the form as attached to this Plan as Annex I, whereby each Former Option accepted by an Employee will be deemed to be an acceptance of options relating to twelve (12) depositary receipts of shares in the capital of SafeBoot Holding B.V.
Business Day	Means each day that is not a Saturday, Sunday or other day on which the Company is closed for business or banking institutions located in San Francisco, California, Amsterdam, the Netherlands or London, United Kingdom are authorized or obligated by applicable law or executive order to close.
Closing	The closing of the transaction relating to the purchase by the Company (or a Subsidiary) of all shares in the capital of Safeboot Holding B.V., being November 19, 2007.
Company	McAfee, Inc., a Delaware corporation.
Company Stock Price	Means the weighted average of the closing sale price for a Share as quoted on the New York Stock Exchange, for the ten (10) trading day period ending on the trading day prior to the date of Closing.
Conversion Ratio	Means the quotient obtained by <i>dividing</i> (i) the amount of \$8.78 by (ii) the Company Stock Price.
Currency Conversion Rate	Means the exchange rate for the conversion of Euros into United States dollars as reported in <i>The Wall Street Journal</i> on the date, three (3) Business Days prior to the date of Closing.
Date of Grant	The date mentioned in the Offer Letter.
Designated Broker	Means the third party designated by the Company from time to time to perform certain employee brokerage services, which may include the provision of an electronic employee brokerage system through which (i) Participants may access information from or pertaining to their individual account, and (ii) Participants may

	exercise Options and provide instructions to place securities-related orders.
Employee	An individual with whom the Company or a Subsidiary has concluded a written employment agreement. An Employee shall not cease to be an Employee in case of:
	(i) a transfer between locations of the Company or between the Company and any of its Subsidiaries or any successors; or
	(ii) a situation approved by the Stichting.
Exercise Form	The electronic form by which Options may be exercised through the Designated Broker.
Exercise Price	The per Share price payable upon exercise of an Option equal to the quotient obtained by <i>dividing</i> (i) the exercise price of one Former Option which related to twelve (12) depositary receipts (converted to U.S. dollars by <i>multiplying</i> such price by the Currency Conversion Rate) <i>by</i> (ii) the Conversion Ratio, rounded up to the nearest whole cent.
Former Option	The right to acquire depositary receipts of shares (certificaten van aandelen) in the capital of SafeBoot Holding B.V. in accordance with the provisions of this Plan.
Liquidity Event	Any of the following:
	 the sale or transfer of more than fifty percent (50%) of the aggregate consolidated assets of the Company and its Subsidiaries (computed on the basis of book value, determined in accordance with generally accepted accounting principles consistently applied, or fair market value, as determined by an independent auditor) in any transaction or series of related transactions (other than sales of inventory in the ordinary course of business and any sale or transfer of a Subsidiary of the Company to the Company or another Subsidiary of the Company);
	(ii) any merger or consolidation to which the Company is a party, except for a merger in which the Company is the surviving corporation, the terms of the Shares are not changed, altered or otherwise affected and the Shares are not exchanged for cash, securities or other property and, after giving effect to such merger, the holders of the Company's outstanding shares possess the voting power (under ordinary circumstances) to elect a majority of the board of directors of the Company;
	2

	(iii) any sale, transfer or issue or series of sales, transfers and/or issues of the Shares or other equity securities by the Company or any holders thereof which results in any person or group of persons ("group" as used herein means persons controlling, controlled by or under common control with the other persons in such group), owning shares or other equity securities of the Company possessing the voting power (under ordinary circumstances) to elect a majority of the board of the Company; or
	(iv) the liquidation, dissolution or winding up of the Company.
Market Value	On any day shall mean the closing sale price for a Share as quoted on the New York Stock Exchange on such day.
Offer Letter	The letter of the Stichting to an Employee containing the offer of Former Options, substantially in the form as attached to this Plan as Annex IV, whereby each Former Option granted in the Offer Letter will be deemed to be a grant of twelve (12) depositary receipts of shares in the capital of SafeBoot Holding B.V.
Options	Former Options as converted effective on the Closing into options to purchase Shares, such that for each holder of Former Options, the aggregate number of Former Options held by such holder have been converted into such number of options exercisable (or will become exercisable in accordance with its terms) for that same number of whole Shares, equal to the product obtained by <i>multiplying</i> (i) the aggregate number of depositary receipts of shares in the capital of SafeBoot Holding B.V. issuable upon exercise of such Former Options immediately prior to the Closing <i>by</i> (ii) the Conversion Ratio, rounded up to the nearest whole Share and such holder holds such number of options to purchase that same number of Shares.
Option Period	The period commencing at the Date of Grant and lasting as specified in the Offer Letter.
Participant	An individual to whom as Employee the Stichting issued Former Options which the individual as Employee accepted by submission of the Acceptance Form.
Plan	This SafeBoot Option Plan 2006, as amended from time to time.
Shares	Each of the shares of the Company's common stock, par value \$0.01 per share.
Stichting	Stichting Administratiekantoor SafeBoot, a foundation organized under the laws of the Netherlands, having its registered office at Breukelen, the Netherlands, with address Stichting

		Administratiekantoor SateBoot, Fortis Intertrust (Netherlands) B.V., P.O. Box 990, 1000 AZ Amsterdam, the Netherlands, registered with the trade register under number 30199120.
Subsi	liary	Any person, whether or not existing on the date hereof, in which the Company, directly or indirectly, beneficially owns at least fifty (50%) of either the equity interest or voting power of or in such person.
Unves	sted Options	Options that are not Vested Options.
Vestee	l Options	Options that have vested according to the Vesting Schedule and are therefore exercisable.
Vestir	g Schedule	A schedule set out in an Offer Letter detailing when Options become exercisable.
1.2	The headings to these rules are for the sake of convenience only and should be ignored when construing the rules.	

1.3 Unless the context determines otherwise, terms defined in singular also include terms defined in plural, and vice versa.

Article 2 — Offer of (Former) Options

- 2.1 In 2006 and 2007, the Stichting has offered Former Options to certain Employees (i) as a means for retention and (ii) as a reward for lasting loyalty and commitment to the enterprises of SafeBoot Holding B.V. and of any Group Company or Subsidiary. The offering of Former Options was at the sole discretion of the Stichting.
- 2.2 The Offer Letter contained the terms and conditions of the offer of Former Options by the Stichting, including the Date of Grant, the exercise price for each Former Option (taking into account the amendment which results from the definition of Offer Letter), the Option Period, the number of Former Options, the Vesting Schedule and such other terms and conditions as may be contained therein, including but not limited to the applicability of the provisions contained in this Plan. The terms and conditions included in the Offer Letter relating to the Former Options are applicable *mutatis mutandis* to the Options.
- 2.3 Acceptance of the offer of Former Options contained in the Offer Letter has been made by returning to the Stichting the Acceptance Form, duly completed with the missing information and duly signed.
- 2.4 None of this Plan, the Offer Letter or any of the other documents related thereto will be considered part of an employment agreement between an Employee and the Company or a Subsidiary and the offer of Former Options does not qualify as an employment condition.
- 2.5 This Plan, the Offer Letter, any of the other documents related thereto nor the offer of Former Options nor the conversion of the Former Options into Options provides an

Employee with any rights to be offered additional Options or any other right against the Stichting other than those contained in the Offer Letter.

- 2.6 The offer of Former Options, the actual grant of Former Options and the conversion of the Former Options into Options will not be considered a guarantee for continuance of a Participant's employment relationship with the Company or with a Subsidiary and no guarantee or other warranty is made by the Stichting that any grant or exercise of (Former) Options shall be profitable. Furthermore, any profit realized by a Participant shall not form part of or be deemed to form part of the Participant's employment agreement. None of this Plan, the Offer Letter or any of the other documents related thereto will influence the Participant's pension rights or entitlements in any manner whatsoever.
- 2.7 The Options may not be sold (legally or beneficially), transferred, pledged or encumbered, other than by will and the laws of descent.

Article 3 — Option Period and Termination of Options

- 3.1 The Option Period commences at the Date of Grant and will last as specified in the Offer Letter. During the Option Period, Options may be exercised once they have vested in accordance with the Vesting Schedule.
- 3.2 Options that have not been exercised within the Option Period will lapse automatically at the end of the last day of the Option Period.
- 3.3 The Stichting is entitled to decide that Vested and Unvested Options shall terminate without any compensation for a Participant:
 - (a) upon violation by a Participant of any of the provisions of this Plan, or
 - (b) if, during the Option Period, a Participant has not carried out his duties under his employment agreement:
 - (i) for a period of one hundred (100) consecutive days, due to illness or for any reason other than the termination of the employment agreement; or
 - (ii) for two hundred (200) working days in aggregate during any period of three (3) consecutive years, due to illness or for any other reason (except for authorized holidays and termination of the employment agreement).
- 3.4 In case of termination of the employment agreement with a Participant on the basis of reasons other than the reasons listed in Article 3.5, all Unvested Options shall terminate without any compensation for losses suffered by the Participant on the date on which the employment agreement is terminated. All Vested Options can be exercised in accordance with this Plan.
- 3.5 In case of termination of the employment agreement with a Participant for urgent cause ("ontslag op staande voet") or material breach, or in case of a court rescission of the employment agreement of a Participant at the request of the Company or a Subsidiary for

reasons which would have constituted an urgent cause, a termination for cause or for material breach, if the employment agreement would have been immediately terminated by the Company or a Subsidiary, all Options granted whether Vested Options or Unvested Options shall terminate without any compensation for losses suffered by the Participant on the date on which the urgent cause or material breach occurs.

Article 4 — Vesting Schedule and Exercise of Options

- 4.1 Options granted under this Plan are subject to a Vesting Schedule as provided for in the Offer Letter.
- 4.2 A Participant may exercise Options, once they have vested, in accordance with this Plan.
- **4.3** Vested Options are exercised by means of submitting the Exercise Form, duly completed with the missing information, through the Designated Broker by the Participant. Vested Options are only validly exercised when the Exercise Form has been submitted through the Designated Broker within the Option Period, therefore ultimately at the end of the last day of the Option Period.

Article 5 — Delivery and Sale of Shares

- 5.1 Promptly following (i) the submission of the Exercise Form and (ii) the payment of the total Exercise Price payable to the Company via the Designated Broker, the number of Shares indicated in the Exercise Form will be delivered to the Participant through the Designated Broker. Such delivery will be evidenced by an increase of the number of Shares held by the Participant with the number of Shares indicated in the Exercise Form in the Participant's account with the Designated Broker. All or any part of the Exercise Price and any withholding taxes may be paid by delivering (on a form prescribed by the Company) an irrevocable direction to the Designated Broker to sell all or part of the Shares being purchased under the Plan and to deliver all or part of the sales proceeds to the Company.
- 5.2 Until the Shares granted to any Participant are registered on Form S-8 with the Securities and Exchange Commission, any Shares acquired upon the exercise of Vested Options may not be sold (legally or beneficially), transferred, pledged or encumbered, other than by will and the laws of descent.
- 5.3 The Stichting may, in its discretion, decide that Article 5.2 is not applicable.
- 5.4 All administrative and transfer costs connected with the abovementioned delivery and sale of Shares and the Plan shall be borne by the Stichting.

Article 6 — Adjustments

6.1 In the event of a consolidation of Shares or any other change in the Company's equity capitalization, the Stichting is authorized to make a similar adjustment to the Exercise Price per Share and the number of Options and/or any other terms and conditions applicable to the Options or this Plan. It is not possible to acquire parts of a Share.

Fractional Shares will be rounded down to whole numbers of Shares, without any compensation for the Participant.

6.2 In case of a Liquidity Event and at the sole discretion of the Stichting and to the extent permitted under the legal, financial and beneficial mechanisms of the Liquidity Event, either (i) the Options shall be terminated against the grant of options to acquire shares or depositary receipts for shares in the capital of a legal entity or entities according to a ratio to be determined by the Stichting, or (ii) the Options shall be terminated against payment by the Stichting to the Participant for each Option of an amount equal to the difference of (A) the Market Value of one (1) Share on the date of the Liquidity Event *minus* (B) the Exercise Price of each Option, or (iii) the Option Period shall be reduced by the Stichting to a date prior to the date of the Liquidity Event; or (iv) any other effect shall be attributed to the Options as reasonably appropriate.

Article 7 — Taxes and premiums

- 7.1 Any wage tax, income tax and social security premiums due in respect of the Options and the Shares will be for the account of the Participant. The Company or Subsidiary shall be entitled to withhold the amount of any tax and social security premiums related to or payable in respect of Options and Shares, which withholding may be made by the Company or the appropriate Subsidiary from any amount (for the purpose hereof 'amount' includes Options and Shares) payable to the Participant. To the extent that the Company or a Subsidiary shall have made any payment of tax or social security premiums in respect of Options or Shares, the Participant will immediately reimburse the Company or the Subsidiary for the full amount so paid. The Company or the appropriate Subsidiary will not be responsible or liable for any incorrect or incomplete withholding of tax and social security premiums.
- 7.2 The Participant is and remains at all times fully responsible for the due payment of any wage tax, income tax and social security premiums due in respect of Options and Shares.

Article 8 — Termination and amendments

- 8.1 This Plan may be terminated by a resolution of the Stichting to that effect. Termination of the Plan shall in no event prejudice the existing rights of a Participant pursuant to this Plan, unless the Participant in writing consents to the termination.
- **8.2** The Stichting is authorized to amend this Plan at any time and in any manner it deems fit. Amendments to this Plan will in no event prejudice the existing rights of a Participant, unless the Participant in writing consents to the amendments. Written notice of any amendment of the Plan shall be given to a Participant so that the Participant is aware of the changes made.

Article 9 — Notices

9.1 A notice under or in connection with this Plan shall be in writing, in the English language and may be delivered personally, sent by recognized courier service or sent by fax to the party due to receive the notice to the address specified in Article 9.2 or to another

address, person or fax number specified by that party by not less than seven (7) days' written notice to the other parties received before the notice was dispatched.

9.2 The address referred to in Article 9.1 is:

in the case of the Stichting: Address: Stichting Administratiekantoor SafeBoot, Fortis Intertrust (Netherlands) B.V., P.O. Box 990, 1000 AZ Amsterdam, the Netherlands Fax: +31.20.5214836 Attention of: Jaap Veerman

in the case of the Participant: the address as specified by the Participant in the Acceptance Form.

in the case of the Company: Address: McAfee, Inc., 5000 Headquarters Dr., Plano, Texas 75024 Attention: Stock Administration, with a copy to the Corporate Secretary

- 9.3 Unless there is evidence that it was received earlier a notice is deemed given:
 - (a) if delivered personally or by courier, when the person delivering the notice obtains the signature of a person at the address referred to in Article 9.2; and
 - (b) if sent by fax, when confirmation of its transmission has been recorded by the sender's fax machine.

Article 10 — Dispute-rules and applicable law

- 10.1 Options are granted pursuant to this Plan and are in all respects limited by and subject to the provisions contained in this Plan. All decisions of the Stichting with respect to any question or issue arising under the Plan, the Offer Letter or any other documents related thereto shall be conclusive and binding on all persons having an interest in the Options.
- **10.2** This Plan shall be governed by the laws of the Netherlands.
- 10.3 All disputes arising in connection with this Plan, the Options or any other documents related thereto shall be submitted in first instance to the jurisdiction of the competent court in Amsterdam.

WARNING

Granting options to directors and officers under this plan may violate NASD or stock exchange rules if the plan does not meet the broad based plan exemption from shareholder approval

NETWORKS ASSOCIATES, INC.

2000 NONSTATUTORY STOCK OPTION PLAN

- 1. Purposes of the Plan. The purposes of this Nonstatutory Stock Option Plan are:
- to attract and retain the best available personnel for positions of substantial responsibility,
- to provide additional incentive to Employees, Directors and Consultants, and
- to promote the success of the Company's business.
- Options granted under the Plan will be Nonstatutory Stock Options.
- 2. Definitions. As used herein, the following definitions shall apply:
- (a) "Administrator" means the Board or any of its Committees as shall be administering the Plan, in accordance with Section 4 of the Plan.

(b) "Applicable Laws" means the requirements relating to the administration of stock option plans under U.S. state corporate laws, U.S. federal and state securities laws, the Code, any stock exchange or quotation system on which the Common Stock is listed or quoted and the applicable laws of any foreign country or jurisdiction where Options are, or will be, granted under the Plan.

- (c) "Board" means the Board of Directors of the Company.
- (d) "Code" means the Internal Revenue Code of 1986, as amended.
- (e) "Committee" means a committee of Directors appointed by the Board in accordance with Section 4 of the Plan.
- (f) "Common Stock" means the Common Stock of the Company.
- (g) "Company" means Networks Associates, Inc., a Delaware corporation.
- (h) "Consultant" means any person, including an advisor, engaged by the Company or a Parent or Subsidiary to render services to such entity.
- (i) "Director" means a member of the Board.

(j) "Disability" means total and permanent disability as defined in Section 22(e)(3) of the Code.

(k) "Employee" means any person, including Officers, employed by the Company or any Parent or Subsidiary of the Company. A Service Provider shall not cease to be an Employee in the case of (i) any leave of absence approved by the Company or (ii) transfers between locations of the Company or between the Company, its Parent, any Subsidiary, or any successor. Neither service as a Director nor payment of a director's fee by the Company shall be sufficient to constitute "employment" by the Company.

(1) "Exchange Act" means the Securities Exchange Act of 1934, as amended.

(m) "Fair Market Value" means, as of any date, the value of Common Stock determined as follows:

(i) If the Common Stock is listed on any established stock exchange or a national market system, including without limitation the Nasdaq National Market or The Nasdaq SmallCap Market of The Nasdaq Stock Market, its Fair Market Value shall be the closing sales price for such stock (or the closing bid, if no sales were reported) as quoted on such exchange or system for the last market trading day prior to the time of determination, as reported in The Wall Street Journal or such other source as the Administrator deems reliable;

(ii) If the Common Stock is regularly quoted by a recognized securities dealer but selling prices are not reported, the Fair Market Value of a Share of Common Stock shall be the mean between the high bid and low asked prices for the Common Stock on the last market trading day prior to the day of determination, as reported in The Wall Street Journal or such other source as the Administrator deems reliable;

(iii) In the absence of an established market for the Common Stock, the Fair Market Value shall be determined in good faith by the Administrator.

(n) "Notice of Grant" means a written or electronic notice evidencing certain terms and conditions of an individual Option grant. The Notice of Grant is part of the Option Agreement.

(o) "Officer" means a person who is an officer of the Company within the meaning of Section 16 of the Exchange Act and the rules and regulations promulgated thereunder.

(p) "Option" means a nonstatutory stock option granted pursuant to the Plan, that is not intended to qualify as an incentive stock option within the meaning of Section 422 of the Code and the regulations promulgated thereunder.

(q) "Option Agreement" means an agreement between the Company and an Optionee evidencing the terms and conditions of an individual Option grant. The Option Agreement is subject to the terms and conditions of the Plan.

(r) "Option Exchange Program" means a program whereby outstanding options are surrendered in exchange for options with a lower exercise price.

(s) "Optioned Stock" means the Common Stock subject to an Option.

(t) "Optionee" means the holder of an outstanding Option granted under the Plan.

(u) "Parent" means a "parent corporation," whether now or hereafter existing, as defined in Section 424(e) of the Code.

(v) "Plan" means this 2000 Nonstatutory Stock Option Plan.

(w) "Service Provider" means an Employee including an Officer, Consultant or Director.

(x) "Share" means a share of the Common Stock, as adjusted in accordance with Section 12 of the Plan.

(y) "Subsidiary" means a "subsidiary corporation," whether now or hereafter existing, as defined in Section 424(f) of the Code.

3. Stock Subject to the Plan. Subject to the provisions of Section 12 of the Plan, the maximum aggregate number of Shares which may be optioned and sold under the Plan is Three Million (3,000,000) Shares. The Shares may be authorized, but unissued, or reacquired Common Stock.

If an Option expires or becomes unexercisable without having been exercised in full, or is surrendered pursuant to an Option Exchange Program, the unpurchased Shares which were subject thereto shall become available for future grant or sale under the Plan (unless the Plan has terminated).

4. Administration of the Plan.

(a) Administration. The Plan shall be administered by (i) the Board or (ii) a Committee, which committee shall be constituted to satisfy Applicable Laws.

(b) Powers of the Administrator. Subject to the provisions of the Plan, and in the case of a Committee, subject to the specific duties delegated by the Board to such Committee, the Administrator shall have the authority, in its discretion:

(i) to determine the Fair Market Value of the Common Stock;

- (ii) to select the Service Providers to whom Options may be granted hereunder;
- (iii) to determine whether and to what extent Options are granted hereunder;
- (iv) to determine the number of shares of Common Stock to be covered by each Option granted hereunder;
- (v) to approve forms of agreement for use under the Plan;

(vi) to determine the terms and conditions, not inconsistent with the terms of the Plan, of any award granted hereunder. Such terms and conditions include, but are not limited to, the exercise price, the time or times when Options may be exercised (which may be based on performance criteria), any vesting acceleration or waiver of forfeiture restrictions, and any restriction or limitation regarding any Option or the shares of Common Stock relating thereto, based in each case on such factors as the Administrator, in its sole discretion, shall determine;

(vii) to reduce the exercise price of any Option to the then current Fair Market Value if the Fair Market Value of the Common Stock covered by such Option shall have declined since the date the Option was granted;

(viii) to institute an Option Exchange Program;

(ix) to construe and interpret the terms of the Plan and awards granted pursuant to the Plan;

(x) to prescribe, amend and rescind rules and regulations relating to the Plan, including rules and regulations relating to sub-plans established for the purpose of qualifying for preferred tax treatment under foreign tax laws;

(xi) to modify or amend each Option (subject to Section 14(b) of the Plan), including the discretionary authority to extend the post-termination exercisability period of Options longer than is otherwise provided for in the Plan;

(xii) to authorize any person to execute on behalf of the Company any instrument required to effect the grant of an Option previously granted by the Administrator;

(xiii) to determine the terms and restrictions applicable to Options;

(xiv) to allow Optionees to satisfy withholding tax obligations by electing to have the Company withhold from the Shares to be issued upon exercise of an Option that number of Shares having a Fair Market Value equal to the amount required to be withheld. The Fair Market Value of the Shares to be withheld shall be determined on the date that the amount of tax to be withheld is to be determined. All elections by an Optionee to have Shares withheld for this purpose shall be made in such form and under such conditions as the Administrator may deem necessary or advisable; and

(xv) to make all other determinations deemed necessary or advisable for administering the Plan.

(c) Effect of Administrator's Decision. The Administrator's decisions, determinations and interpretations shall be final and binding on all Optionees and any other holders of Options.

5. Eligibility. Options may be granted to Service Providers; provided, however, that notwithstanding anything to the contrary contained in the Plan, Options may not be granted to Officers and Directors.

6. Limitation. Neither the Plan nor any Option shall confer upon an Optionee any right with respect to continuing the Optionee's relationship as a Service Provider with the Company, nor shall they interfere in any way with the Optionee's right or the Company's right to terminate such relationship at any time, with or without cause.

7. Term of Plan. The Plan shall become effective upon its adoption by the Board. It shall continue in effect for ten (10) years, unless sooner terminated under Section 14 of the Plan.

8. Term of Option. The term of each Option shall be stated in the Option Agreement.

9. Option Exercise Price and Consideration.

(a) Exercise Price. The per share exercise price for the Shares to be issued pursuant to exercise of an Option shall be determined by the Administrator.

(b) Waiting Period and Exercise Dates. At the time an Option is granted, the Administrator shall fix the period within which the Option may be exercised and shall determine any conditions which must be satisfied before the Option may be exercised.

(c) Form of Consideration. The Administrator shall determine the acceptable form of consideration for exercising an Option, including the method of payment. Such consideration may consist entirely of:

(i) cash;

(ii) check;

(iii) promissory note;

(iv) other Shares which (A) in the case of Shares acquired upon exercise of an option, have been owned by the Optionee for more than six months on the date of surrender, and (B) have a Fair Market Value on the date of surrender equal to the aggregate exercise price of the Shares as to which said Option shall be exercised;

(v) consideration received by the Company under a cashless exercise program implemented by the Company in connection with the Plan;

(vi) a reduction in the amount of any Company liability to the Optionee, including any liability attributable to the Optionee's participation in any Company-sponsored deferred compensation program or arrangement;

(vii) such other consideration and method of payment for the issuance of Shares to the extent permitted by Applicable Laws; or

(viii) any combination of the foregoing methods of payment.

10. Exercise of Option.

(a) Procedure for Exercise; Rights as a Shareholder. Any Option granted hereunder shall be exercisable according to the terms of the Plan and at such times and under such conditions as determined by the Administrator and set forth in the Option Agreement. An Option may not be exercised for a fraction of a Share.

An Option shall be deemed exercised when the Company receives:

(i) written or electronic notice of exercise (in accordance with the Option Agreement) from the person entitled to exercise the Option, and (ii) full payment for the Shares with respect to which the Option is exercised. Full payment may consist of any consideration and method of payment authorized by the Administrator and permitted by the Option Agreement and the Plan. Shares issued upon exercise of an Option shall be issued in the name of the Optionee or, if requested by the Optionee, in the name of the Optionee and his or her spouse. Until the Shares are issued (as evidenced by the appropriate entry on the books of the Company or of a duly authorized

transfer agent of the Company), no right to vote or receive dividends or any other rights as a shareholder shall exist with respect to the Optioned Stock, notwithstanding the exercise of the Option. The Company shall issue (or cause to be issued) such Shares promptly after the Option is exercised. No adjustment will be made for a dividend or other right for which the record date is prior to the date the Shares are issued, except as provided in Section 12 of the Plan.

Exercising an Option in any manner shall decrease the number of Shares thereafter available, both for purposes of the Plan and for sale under the Option, by the number of Shares as to which the Option is exercised.

(b) Termination of Relationship as a Service Provider. If an Optionee ceases to be a Service Provider, other than upon the Optionee's death or Disability, the Optionee may exercise his or her Option, but only within such period of time as is specified in the Option Agreement, and only to the extent that the Option is vested on the date of termination (but in no event later than the expiration of the term of such Option as set forth in the Option Agreement). In the absence of a specified time in the Option Agreement, the Option shall remain exercisable for three (3) months following the Optionee's termination. If, on the date of termination, the Optionee is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option shall revert to the Plan. If, after termination, the Optionee does not exercise his or her Option within the time specified by the Administrator, the Option shall terminate, and the Shares covered by such Option shall revert to the Plan.

(c) Disability of Optionee. If an Optionee ceases to be a Service Provider as a result of the Optionee's Disability, the Optionee may exercise his or her Option within such period of time as is specified in the Option Agreement, to the extent the Option is vested on the date of termination (but in no event later than the expiration of the term of such Option as set forth in the Option Agreement). In the absence of a specified time in the Option Agreement, the Option shall remain exercisable for twelve (12) months following the Optionee's termination. If, on the date of termination, the Optionee is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option shall revert to the Plan. If, after termination, the Optionee does not exercise his or her Option within the time specified herein, the Option shall terminate, and the Shares covered by such Option shall revert to the Plan.

(d) Death of Optionee. If an Optionee dies while a Service Provider, the Option may be exercised within such period of time as is specified in the Option Agreement (but in no event later than the expiration of the term of such Option as set forth in the Notice of Grant), by the Optionee's estate or by a person who acquires the right to exercise the Option by bequest or inheritance, but only to the extent that the Option is vested on the date of death. In the absence of a specified time in the Option Agreement, the Option shall remain exercisable for twelve (12) months following the Optionee's termination. If, at the time of death, the Optionee is not vested as to his or her entire Option, the Shares covered by the unvested portion of the Option shall immediately revert to the Plan. The Option may be exercised by the executor or administrator of the Optionee's estate or, if none, by the person(s) entitled to exercise the Option under the Optionee's will or the laws of descent or distribution. If the Option is not so exercised within the time specified herein, the Option shall terminate, and the Shares covered by such Option shall revert to the Plan.

(e) Buyout Provisions. The Administrator may at any time offer to buy out for a payment in cash or Shares, an Option previously granted based on such terms and conditions as the Administrator shall establish and communicate to the Optionee at the time that such offer is made.

11. Non-Transferability of Options. Unless determined otherwise by the Administrator, an Option may not be sold, pledged, assigned, hypothecated, transferred, or disposed of in any manner other than by will or by the laws of descent or distribution and may be exercised, during the lifetime of the Optionee,

only by the Optionee. If the Administrator makes an Option transferable, such Option shall contain such additional terms and conditions as the Administrator deems appropriate.

12. Adjustments Upon Changes in Capitalization, Dissolution, Merger or Asset Sale.

(a) Changes in Capitalization. Subject to any required action by the shareholders of the Company, the number of shares of Common Stock covered by each outstanding Option, and the number of shares of Common Stock which have been authorized for issuance under the Plan but as to which no Options have yet been granted or which have been returned to the Plan upon cancellation or expiration of an Option, as well as the price per share of Common Stock covered by each outstanding Option, shall be proportionately adjusted for any increase or decrease in the number of issued shares of Common Stock resulting from a stock split, reverse stock split, stock dividend, combination or reclassification of the Common Stock, or any other increase or decrease in the number of issued shares of Common Stock effected without receipt of consideration by the Company; provided, however, that conversion of any convertible securities of the Company shall not be deemed to have been "effected without receipt of consideration." Such adjustment shall be made by the Board, whose determination in that respect shall be final, binding and conclusive. Except as expressly provided herein, no issuance by the Company of shares of stock of any class, or securities convertible into shares of stock of any class, shall affect, and no adjustment by reason thereof shall be made with respect to, the number or price of shares of Common Stock subject to an Option.

(b) Dissolution or Liquidation. In the event of the proposed dissolution or liquidation of the Company, the Administrator shall notify each Optionee as soon as practicable prior to the effective date of such proposed transaction. The Administrator in its discretion may provide for an Optionee to have the right to exercise his or her Option until ten (10) days prior to such transaction as to all of the Optioned Stock covered thereby, including Shares as to which the Option would not otherwise be exercisable. In addition, the Administrator may provide that any Company repurchase option applicable to any Shares purchased upon exercise of an Option shall lapse as to all such Shares, provided the proposed dissolution or liquidation takes place at the time and in the manner contemplated. To the extent it has not been previously exercised, an Option will terminate immediately prior to the consummation of such proposed action.

(c) Merger or Asset Sale. In the event of a merger of the Company with or into another corporation, or the sale of substantially all of the assets of the Company, each outstanding Option shall be assumed or an equivalent option or right substituted by the successor corporation or a Parent or Subsidiary of the successor corporation. In the event that the successor corporation refuses to assume or substitute for the Option, the Optionee shall fully vest in and have the right to exercise the Option as to all of the Optioned Stock, including Shares as to which it would not otherwise be vested or exercisable. If an Option becomes fully vested and exercisable in lieu of assumption or substitution in the event of a merger or sale of assets, the Administrator shall notify the Optionee in writing or electronically that the Option shall be fully vested and exercisable for a period of fifteen (15) days from the date of such notice, and the Option shall terminate upon the expiration of such period. For the purposes of this paragraph, the Option shall be considered assumed if, following the merger or sale of assets, the option or right confers the right to purchase or receive, for each Share of Optioned Stock, immediately prior to the merger or sale of assets, the consideration (whether stock, cash, or other securities or property) received in the merger or sale of assets by holders of Common Stock for each Share held on the effective date of the transaction (and if holders were offered a choice of consideration, the type of consideration chosen by the holders of a majority of the outstanding Shares); provided, however, that if such consideration received in the merger or sale of assets is not solely common stock of the successor corporation or its Parent, the Administrator may, with the consent of the successor corporation, provide for the consideration to be received upon the exercise of the Option, for each Share of Optioned Stock in the merger or sale of assets.

13. Date of Grant. The date of grant of an Option shall be, for all purposes, the date on which the Administrator makes the determination granting such Option, or such other later date as is determined by the Administrator. Notice of the determination shall be provided to each Optionee within a reasonable time after the date of such grant.

14. Amendment and Termination of the Plan.

(a) Amendment and Termination. The Board may at any time amend, alter, suspend or terminate the Plan.

(b) Effect of Amendment or Termination. No amendment, alteration, suspension or termination of the Plan shall impair the rights of any Optionee, unless mutually agreed otherwise between the Optionee and the Administrator, which agreement must be in writing and signed by the Optionee and the Company. Termination of the Plan shall not affect the Administrator's ability to exercise the powers granted to it hereunder with respect to options granted under the Plan prior to the date of such termination.

15. Conditions Upon Issuance of Shares.

(a) Legal Compliance. Shares shall not be issued pursuant to the exercise of an Option unless the exercise of such Option and the issuance and delivery of such Shares shall comply with Applicable Laws and shall be further subject to the approval of counsel for the Company with respect to such compliance.

(b) Investment Representations. As a condition to the exercise of an Option the Company may require the person exercising such Option to represent and warrant at the time of any such exercise that the Shares are being purchased only for investment and without any present intention to sell or distribute such Shares if, in the opinion of counsel for the Company, such a representation is required.

16. Inability to Obtain Authority. The inability of the Company to obtain authority from any regulatory body having jurisdiction, which authority is deemed by the Company's counsel to be necessary to the lawful issuance and sale of any Shares hereunder, shall relieve the Company of any liability in respect of the failure to issue or sell such Shares as to which such requisite authority shall not have been obtained.

17. Reservation of Shares. The Company, during the term of this Plan, will at all times reserve and keep available such number of Shares as shall be sufficient to satisfy the requirements of the Plan.